Democratic Services

Riverside, Temple Street, Keynsham, Bristol BS31 1LA Telephone: (01225) 477000 *main switchboard* Direct Lines - Tel: 01225 394358 Fax: 01225 394439 Web-site - http://www.bathnes.gov.uk

Date: 11 February 2013 E-mail: Democratic_Services@bathnes.gov.uk

To: All Members of the Council

Chief Executive and other appropriate officers Press and Public

Dear Member

Council: Tuesday, 19th February, 2013

You are invited to attend a meeting of the **Council** to be held on **Tuesday**, **19th February**, **2013** at **6.30 pm** in the **Council Chamber** - **Guildhall**, **Bath**.

The agenda is set out overleaf.

Sandwiches and fruit and tea/coffee/cold drinks will be available for Councillors from 5pm in the Aix-en-Provence Room (next to the Banqueting Room) on Floor 1.

Yours sincerely



Jo Morrison Democratic Services Manager for Chief Executive

Please note the following arrangements for pre-group meetings:

Conservative	Brunswick Room, Ground Floor
Liberal Democrat	Kaposvar Room, Floor 1
Labour	Small Meeting Room, Floor 1
Independent	Performance and Improvement Team Office, Floor 1

If you need to access this agenda or any of the supporting reports in an alternative accessible format please contact Democratic Services or the relevant report author whose details are listed at the end of each report.

This Agenda and all accompanying reports are printed on recycled paper

NOTES:

- 1. **Inspection of Papers:** Any person wishing to inspect minutes, reports, or a list of the background papers relating to any item on this Agenda should contact Jo Morrison who is available by telephoning Bath 01225 394358.
- 2. **Details of decisions taken at this meeting** can be found in the minutes which will be circulated with the agenda for the next meeting. In the meantime, details can be obtained by contacting as above. Papers are available for inspection as follows:

Public Access points – Guildhall – Bath, Riverside – Keynsham, Hollies – Midsomer Norton, and Bath Central, Keynsham and Midsomer Norton Public Libraries.

For Councillors and officers, papers may be inspected via Political Group Research Assistants and Group Rooms/Members' Libraries.

- 3. **Spokespersons:** The Political Group Spokespersons for the Council are the Group Leaders, who are Councillors Paul Crossley (Liberal Democrat Group), Francine Haeberling (Conservative Group), John Bull (Labour Group) and Doug Deacon (Independent Group).
- 4. **Attendance Register:** Members should sign the Register, which will be circulated at the meeting.
- 5. Public Speaking at Meetings: The Council has a scheme to encourage the public to make their views known at meetings. They may make a statement relevant to what the meeting has power to do. They may also present a petition or a deputation on behalf of a group. They may also ask a question to which a written answer will be given. Advance notice is required not less than two full working days before the meeting. This means that for meetings held on Thursdays notice must be received in Democratic Services by 5.00pm the previous Monday. Further details of the scheme can be obtained by contacting Jo Morrison as above.
- 6. THE APPENDED SUPPORTING DOCUMENTS ARE IDENTIFIED BY AGENDA ITEM NUMBER.

7. Emergency Evacuation Procedure

When the continuous alarm sounds, you must evacuate the building by one of the designated exits and proceed to the named assembly point. The designated exits are sign-posted.

Arrangements are in place for the safe evacuation of disabled people.

8. **Presentation of reports:** Officers of the Council will not normally introduce their reports unless requested by the meeting to do so. Officers may need to advise the meeting of new information arising since the agenda was sent out.

Council - Tuesday, 19th February, 2013 at 6.30 pm in the Council Chamber - Guildhall, Bath

<u>A G E N D A</u>

1. EMERGENCY EVACUATION PROCEDURE

The Chairman will draw attention to the emergency evacuation procedure as set out under Note 7.

- 2. APOLOGIES FOR ABSENCE
- 3. DECLARATIONS OF INTEREST

At this point in the meeting declarations of interest are received from Members in any of the agenda items under consideration at the meeting. Members are asked to complete the green interest forms circulated to groups in their pre-meetings (which will be announced at the Council Meeting) to indicate:

- (a) The agenda item number in which they have an interest to declare.
- (b) The nature of their interest.
- (c) Whether their interest is a disclosable pecuniary interest <u>or</u> an other interest, (as defined in Part 2, A and B of the Code of Conduct and Rules for Registration of Interests)

Any Member who needs to clarify any matters relating to the declaration of interests is recommended to seek advice from the Council's Monitoring Officer before the meeting to expedite dealing with the item during the meeting.

4. MINUTES - 8TH NOVEMBER 2012 (Pages 7 - 16)

To be confirmed as a correct record and signed by the Chair(man)

5. ANNOUNCEMENTS FROM THE CHAIRMAN OF THE COUNCIL OR FROM THE CHIEF EXECUTIVE

These are matters of information for Members of the Council. No decisions will be required arising from the announcements.

6. TO ANNOUNCE ANY URGENT BUSINESS AGREED BY THE CHAIRMAN

If there is any urgent business arising since the formal agenda was published, the Chairman will announce this and give reasons why he has agreed to consider it at this meeting. In making his decision, the Chairman will, where practicable, have consulted with the Leaders of the Political Groups. Any documentation on urgent business will be circulated at the meeting, if not made available previously.

7. QUESTIONS, STATEMENTS, PETITIONS AND DEPUTATIONS FROM THE PUBLIC

The Democratic Services Manager will announce any submissions received under the arrangements set out in note 5 above. The Council will be invited to decide what action

it wishes to take, if any, on the matters raised in these submissions. As the questions received and the answers given will be circulated in written form there is no requirement for them to be read out at the meeting. The questions and answers will be published with the draft minutes.

 MEDIUM TERM SERVICE & RESOURCE PLANNING 2013/14 - 2015/16 & BUDGET & COUNCIL TAX 2013/14 (Pages 17 - 158)

This report presents the Cabinet's draft medium term financial plan, and revenue and capital budgets for the 2013/14 financial year together with a proposal for a Council Tax level for 2013/14.

9. TRANSPORT IMPROVEMENT CAPITAL PROGRAMME 2013/14 (Pages 159 - 164)

This report sets out the draft Integrated Transport Improvement Capital Programme for consultation. The programme of expenditure aims to develop the policies of the Joint Local Transport Plan in accordance with Government guidelines.

10. TREASURY MANAGEMENT STRATEGY STATEMENT & ANNUAL INVESTMENT STRATEGY 2013/14 (Pages 165 - 190)

In February 2012, the Council adopted the revised CIPFA Treasury Management in Public services Code of Practice 2011 Edition, which requires the Council to approve a Treasury Management Strategy before the start of each financial year and for this to be scrutinised by an individual / group of individuals or committee. In addition, the Department for Communities and Local Government (CLG) issued revised guidance on local authority investments in March 2010 that requires the Council to approve an investment strategy before the start of each financial year.

This report fulfils the Council's legal obligation under the *Local Government Act 2003* to have regard to both the CIPFA Code and the CLG Guidance. This report is being considered by the Corporate Audit Committee at the 5th February 2013 meeting, following which any recommendations will be reported back verbally as an update to this report.

11. NOMINATION OF COUNCIL CHAIR(MAN) DESIGNATE 2013/14 (Pages 191 - 194)

The purpose of this report is to invite the Council to name a Councillor as Chair(man) of the Council (Designate) for the next Council Year beginning in May 2013. The Council, at its Annual Meeting on 16th May 2013, will still be required formally to elect a Councillor to be the Chair(man) of the Council for the forthcoming Council Year.

12. QUESTIONS, STATEMENTS, PETITIONS AND DEPUTATIONS FROM COUNCILLORS

The Democratic Services Manager will announce any submissions received. The Council will be invited to decide what action it wishes to take, if any, on the matters raised in these submissions. As the questions received and the answers given will be circulated in written form there is no requirement for them to be read out at the meeting. The questions and answers will be published with the draft minutes. The Committee Administrator for this meeting is Jo Morrison who can be contacted on 01225 394358.

This page is intentionally left blank

BATH AND NORTH EAST SOMERSET COUNCIL

MINUTES OF COUNCIL MEETING

Thursday, 8th November, 2012

Present:- Councillors Simon Allen, Patrick Anketell-Jones, Rob Appleyard Chairman), Sharon Ball, Tim Ball, Colin Barrett, Gabriel Batt, Cherry Beath, David Bellotti, Sarah Bevan, Mathew Blankley, Lisa Brett, John Bull, Neil Butters, Bryan Chalker, Anthony Clarke, Nicholas Coombes, Paul Crossley, Gerry Curran, Sally Davis, Douglas Deacon, David Dixon, Peter Edwards, Michael Evans, Paul Fox, Andrew Furse, Charles Gerrish, Ian Gilchrist, Francine Haeberling, Alan Hale, Katie Hall, Liz Hardman, Nathan Hartley, Steve Hedges, Eleanor Jackson, Les Kew, Dave Laming, Malcolm Lees, Marie Longstaff, Barry Macrae, David Martin, Loraine Morgan-Brinkhurst MBE, Robin Moss, Paul Myers, Douglas Nicol, Bryan Organ, June Player, Vic Pritchard, Manda Rigby, Caroline Roberts, Nigel Roberts, Dine Romero, Will Sandry, Kate Simmons, Jeremy Sparks, Ben Stevens, Roger Symonds, David Veale, Martin Veal, Geoff Ward, Tim Warren, Chris Watt and Brian Webber

Apology for absence: Councillor Brian Simmons

54 EMERGENCY EVACUATION PROCEDURE

The Senior Democratic Services Officer read out the emergency evacuation procedure as set out on the Agenda

55 APOLOGIES FOR ABSENCE

An apology for absence was received from Councillor Brian Simmons

56 DECLARATIONS OF INTEREST

The Senior Democratic Services Officer stated that notice of a declaration of "other" interest had been given by Councillor Eleanor Jackson regarding Report 13 Public Health Contracts as she was a representative on Sirona Care and Health Community Interest Company. Councillors Vic Pritchard and Sally Davis also declared the same interests in this regard. Councillor Loraine Morgan-Brinkhurst MBE declared an "other" interest as Board Member of Sirona Care and Health Community Interest Company. Councillors Nigel Roberts declared an "other" interest in Report 13 as he was an employee of the Royal United Hospital. Councillor Caroline Roberts declared an "other" interest by association, ie his wife, in the same Report.

57 MINUTES - 13TH SEPTEMBER 2012

RESOLVED that, subject to the name of Simon Douglas in Minute No 45 being spelt Simon Douglas<u>s</u>, the minutes of the meeting of 13th September 2012 be approved as a correct record and signed by the Chairman

58 ANNOUNCEMENTS FROM THE CHAIRMAN OF THE COUNCIL OR FROM THE CHIEF EXECUTIVE

The Chairman made the following announcements:

- 1. He asked everyone to turn their phones to off/silent and reminded members of the public that some Councillors were accessing their meeting papers on their iPads.
- 2. He referred to the agenda timings and asked everyone to keep contributions relevant and not to repeat comments already made.
- 3. He referred to the recent death of Honorary Alderman and former Chair of the Council, Councillor Betty Perry and gave details of the Memorial Service to be held. At the Chairman's request, the Council stood in silence as a mark of respect.
- 4. He indicated that he proposed to waive Council Rule 37 (for the last time if recommendations in Report 15 were agreed by Council this evening) so as not to permit Councillors seconding motions or amendments being able to reserve their right to speak until later in the debate but to require all seconders, if they wished to speak, to do so when they seconded the motion or amendment. The Council indicated its agreement.
- 5. He stated that he would announce a comfort break between 8pm and 9pm if the meeting did not appear to be near its conclusion by then.
- 6. He referred to a programme of events to be held to fund Chairman's causes; a Charity Ball to be held in March which needed to be promoted; the British Empire Medal being awarded to Councillor Sally Davies in the recent Honours List (which was greeted with applause by Members); and an impending Awards Ceremony for Volunteers.

With the agreement of the Chairman, the Leader of the Council gave details of the refurbishment at Haycombe Crematorium including the design of the Cross which would be hung in the window and only removed if required.

59 TO ANNOUNCE ANY URGENT BUSINESS AGREED BY THE CHAIRMAN

There were no items of urgent business

60 QUESTIONS, STATEMENTS, PETITIONS AND DEPUTATIONS FROM THE PUBLIC

Statements to the meeting were made by the following people;

 Mark Owen, Managing Director of HorseWorld, Whitchurch, made a statement highlighting the work of the charity and their plans for a brand new visitor centre. Councillor Cherry Beath enquired as to the benefits to the community from Horseworld. Mr Owen replied that it was a major attraction in the area and therefore boosted the economy by bringing visitors from both Britain and abroad. The statement was referred to the Cabinet Member for Homes and Planning. (A copy of the statement has been retained in the Minute Book and added to the electronic record).

 Mr Peter Marsh, Campaign Co-ordinator in Kingsmead for Bath Labour Action Team, made a statement regarding the proposal to close the GP-led Health Centre in James Street West, Bath (aka Riverside Walk-in Centre). Councillor John Bull enquired as to what actual and projected monies would be saved from the move from Riverside to the RUH. Mr Marsh replied that the Clinical Commissioning Group had said that there would be a saving of £1.3M; however, the cost of the alternative provision at the RUH would give a best guess figure of up to £650k which was only 0.22% of the total. Plans had not been finalised and therefore figures could change. The statement was referred to the appropriate Cabinet Member. (A copy of the statement has been retained in the Minute Book and added to the electronic record).

A Question had been received from Mr Rae Harris. The response had been circulated to Councillors and at the meeting. (A copy of the Question and Answer is available in the Council's Minute book and on the website.)

61 THE LOCAL COUNCIL TAX SUPPORT SCHEME

The Council considered a report outlining proposals for a new Local Council Tax Scheme which will replace the existing system of Council Tax Benefit on 1st April 2013.

On a motion from Councillor David Bellotti, seconded by Councillor Paul Crossley, it was:

- To delegate authority to the Section 151 Officer, in consultation with the Cabinet Member for Community Resources, to implement recommendations of this report as soon as practicably possible, following the Royal Assent of the Local Government Finance Bill 2012, subject to no major changes to those already prescribed by the Department for Communities and Local Government;
- To approve the new Section 13a Local Government Finance Act 1992 Policy for Bath & North East Somerset Council as set out in Appendix 1 to the report, with the key changes to the previous Council Tax Benefit scheme set out in paragraph 5.6 of the Report but with an amendment that entitlement will now cease for claimants with capital/assets over £10k (and not over £6k as proposed);
- 3. To agree that any changes resulting from consideration of the final requirements are approved as set out in Resolution 1 above;
- 4. To agree the vulnerable requirements of the scheme as detailed in paragraph 5.4 of the report;

- 5. To agree not to apply for additional voluntary discretionary grant funding for the reasons described at paragraphs 5.8 5.13 of the report; and
- 6. To thank the Officers for their hard work and endeavours in producing this report.

(Notes: The amendments in Resolution 2 above, proposed by Councillor Charles Gerrish, seconded by Councillor Brian Webber, were accepted by the proposer and seconder of the motion.

There was a further amendment by Councillor John Bull, seconded by Councillor Robin Moss, expressing concern regarding insufficient detail and testing; concerns about the Parish Precept and the impact on affected residents; the lack of appropriate information in advance of the meeting; and calling for the decision to be deferred until a Working Group of Members and Officers could report back to Council before the end of January 2013 with options on this and the proposal contained in Agenda Item 9 "Council Tax Technical Changes for Discounts and Exemptions". The motion was lost, 8 Members voting in favour, 40 against and 10 abstaining.)

62 COUNCIL TAX TECHNICAL CHANGES FOR DISCOUNTS AND EXEMPTIONS

The Council considered a report describing options for adopting new discretionary powers for setting discounts and exemptions to Council Tax against a range of changes proposed by the Local Government Finance Bill/Act 2012, due to come into effect from April 2013.

It was moved by Councillor David Bellotti, seconded by Councillor Paul Crossley, to approve the Recommendations 2.1 - 2.3 set out in the report.

There were various amendments and a lot of debate regarding the motion and amendments. The meetings adjourned for 15 minutes for Officers to recommend some wording for one of the amendments. After substantial debate, Councillor Dave Dixon moved Rule 43 ie "that the Question be now put" which the Chairman put to the vote and was agreed. Accordingly, the motion was put to the vote and it was:

- 1. To delegate authority to the Section 151 Officer to make such technical changes and minor adjustment to these proposals as may be required as a result of current, unresolved Communities and Local Government consultations and any further Government or legislative activity;
- 2. To adopt the full range of technical changes to Council Tax as set out in Appendix 1 to the report;
- 3. That the Financial Implications be factored into the proposed budget for 2013/14 to be considered by Council in February 2013, and that the Cabinet be recommended to consider the allocation of additional sums collected from the technical changes, to supplement the Social Fund which is being delegated to Local Authorities, with particular reference to

vulnerable groups adversely affected by the Council Tax Support changes;

- 4. To create a scheme to assist landlords where tenants have left properties in a state not enabling the property to be immediately re-let and where other demonstrable cause of hardship occur; and
- 5. That a time limit of 3 months be applied to cases which are considered eligible for support in the above Scheme (Resolution 4 immediately above).

(Note: The Resolutions in bold above were accepted by the proposer and seconder of the motion as amendments, appropriately moved and seconded by Members.)

63 COUNCIL TAX BASE 2013-2014

The Council considered a report describing the calculation of Council Tax Base within an environment of changes proposed by the Local Government Finance Bill/Act 2012, due to come into effect from April 2013. Council is asked to approve calculation of the tax base for the area and the amounts for each Parish.

The Monitoring Officer and Divisional Director, Legal and Democratic Services, recommended additional wording to Recommendation 2.5 in the report.

Councillor David Bellotti moved the approval of the Recommendations in the report, together with the additional wording in 2.5, which was seconded by Councillor Paul Crossley, and it was:

- 1. To agree the calculation of the Council's tax base for the year 2013/14 as set out in the report;
- 2. To note an anticipated surplus of £200,000 on the collection fund at the end of 2012/13;
- To agree the amount calculated by Bath & North East Somerset Council as its tax base for the whole area for the year 2013/14 shall be 59,137.26 after adjustment for discretionary changes to discounts ad exemptions, and the impact of the proposed Local Council Tax Support scheme;
- 4. To agree that for the parts of the area, for the year 2013/14, the tax base is calculated as shown in Appendix 1, both protected from the impacts of the Council Tax Support scheme and unprotected from its impact, while allowing for a 98.25% collection rate; and
- 5. To agree that delegated authority be given to the Section 151 Officer to make such technical changes and minor adjustment to the tax base as may be required including as a result of current, unresolved Department of Community and Local Government consultations and any further Government or legislative activity.

64 TREASURY MANAGEMENT MONITORING REPORT TO 30TH SEPTEMBER 2012)

The Council considered a report giving details of performance against the Council's Treasury Management Strategy and Annual Investment Plan 2012/13 for the first six months of 2012/13.

On a motion from Councillor David Bellotti, seconded by Councillor Paul Crossley, it was:

RESOLVED

- 1. To note the Treasury Management report to 30th September 2012, prepared in accordance with the CIPFA Treasury Code of Practice; and
- 2. To note the Treasury Management Indicators to 30th September 2012.

65 REVIEW OF THE COUNCIL'S STATEMENT OF PRINCIPLES UNDER THE GAMBLING ACT 2005

The Council considered a report inviting adoption of the draft Statement of Principles ("the Statement") under the Gambling Act 2005 ("the Act") following conclusion of a consultation exercise.

On a motion from Councillor Dave Dixon, seconded by Councillor Sarah Bevan, it was

RESOLVED to adopt the Statement of Principles at Annex B to the report.

66 STATUTORY TRANSFER AND EXTENSION OF PUBLIC HEALTH CONTRACTS

The Council considered a report which proposed a way of ensuring continuity of services as a range of Public Health functions transfer from the NHS to the Local Authority on 1 April 2013.

On a motion from Councillor Simon Allen, seconded by Councillor Katie Hall, it was:

- To agree that approval in principle is given to proceed with an extension of relevant current contracts for up to 12 months from 31st March 2013 and an endorsement of remaining contracts which have expiry dates beyond 31st March 2013 on terms and conditions to be agreed, subject to appropriate officer scrutiny and in the light of confirmation of the relevant ring-fenced national funding allocation; and
- 2. To authorise the Chief Executive, in consultation with the Monitoring Officer and Chief Financial Officer and following an appropriate officer analysis of risks, to approve specific contractual commitments (including contract extensions as described) from 1st April 2013 to ensure continuity of services.

(Note: Councillor Vic Pritchard moved an amendment, seconded by Councillor Anthony Clarke, to replace Recommendation 2.2 with the following: That the Chief Executive, in consultation with the Monitoring Officer, the Chief Financial Officer and the Chair of the Wellbeing PD&S Panel, and following an appropriate analysis of risks, be authorised to approve specific contractual commitments (including contract extensions as described) from 1 April 2013 to ensure continuity of services. That amendment was lost, 24 voting in favour, 29 against with 3 abstentions.)

67 DESIGNATED PUBLIC PLACES ORDER TO CONTROL STREET DRINKING IN MIDSOMER NORTON

The Council considered a report describing arrangements proposed to restrict the consumption of alcohol in designated public places in Midsomer Norton to ameliorate the incidence of alcohol related nuisance and antisocial behaviour.

The Recommendations in the Report were moved by Councillor Dave Dixon and seconded by Councillor Michael Evans.

Members debated the motion. It was considered that some of the edges of the boundaries of the designated area needed to be clearly identified and that the adjoining Ward of Midsomer Norton Westfield be included in the consultation. The motion was put to the vote and it was:

- 1 To agree that Officers should undertake a formal consultation process in accordance with the Local Authorities (Alcohol Consumption in Designated Public Places) Regulations 2007 with a view to the Council designating the area of Midsomer Norton (as shown on the map in Appendix 1 of the report) under the powers given in the Criminal Justice and Police Act 2001 Section 13 (the Act)
- 2 To agree that the following steps should be taken as part of the consultation process:
 - 2.1 A public advertisement inviting comments [The public advertisement should give notice of the public meetings to be held at Midsomer Norton Town Hall where local stakeholders will be advised of the terms and likely impact of the designated Public Places Order (DPPO) and be invited to make their comments];
 - 2.2 A written invitation issued to the following interested parties to give their views: Avon and Somerset Police, Midsomer Norton Town Council, Radstock Town Council, residents living within the proposed area of the DPPO (Designated Public Place Order), representatives of the licensed trade and any other body agreed by the Members;
 - 2.3 Reasonable steps to consult owners/occupiers of the main public buildings and open spaces within the proposed designated area;

- 2.4 The designated area identified for consultation in Resolution 1 should take account of the risk of displacement;
- 2.5 Owners of private premises open to or accessible to the public within the proposed designated area are invited through the consultation process to state whether they are happy for the Police to use the powers within the Act on their property;
- 2.6 The Council reaffirms to the Police their support for the use of these powers only in situations where there is risk of crime, disorder or nuisance occurring;
- 3 That a further report be submitted once consultation has been completed to consider how to proceed; and
- 4 That the Licensing Committee be authorised to act on behalf of Council on matters relating to DPPOs and that its Terms of Reference be amended accordingly.

68 CONSTITUTIONAL ISSUES ARISING FROM NEW EXECUTIVE ARRANGEMENTS, AND OTHER MISCELLANEOUS CHANGES

The Council considered a report setting out various Constitutional amendments required as a result of the Local Authorities (Executive Arrangements) (Meetings and Access to Information) (England) Regulations 2012 which came into effect in September.

On a motion from Councillor Paul Crossley, seconded by Councillor Tim Warren, it was

- 1. To note the amended provisions regarding private business at Cabinet meetings, as set out in paragraphs 4.3 4.7 of the report;
- To note the arrangements regarding exceptions to advance notice of key decisions (paragraph 4.8 of the report) and agree that specific reference should be made to the relevant Policy Development & Scrutiny Panel Chair being informed;
- 3. To agree that the frequency for reports to Council from the Leader containing details of all key decisions taken under Special Urgency provisions shall be annually, or at another interval as decided by Council, as set out in paragraph 4.9 of the report;
- 4. To agree that the power to grant dispensations is delegated to the Monitoring Officer as set out in paragraph 4.11 of the report;
- 5. To note the additional provisions regarding overview and scrutiny rights to Cabinet documents, as set out in paragraphs 4.12 4.14 of the report;

- 6. To agree to the suggested change to the 'Recording of Meetings' Constitution section (paragraph 4.16 of the report) to enable more citizen blogging;
- To nominate the Wellbeing Policy Development & Scrutiny Panel as the Council's arrangement for delivering the health scrutiny function from April 2013 (paragraphs 4.18 - 4.20 of the report refer);
- 8. To agree the amendment to the Constitution regarding Council Rule 37, as set out in paragraph 4.21 of the report, to remove the ability for a Member, when seconding a motion or amendment, to reserve their speech until later in the debate;
- 9. To agree that all references to the Proper Officer shall be taken as being the Monitoring Officer and Divisional Director, Legal and Democratic Services, for the purposes of the Constitution; and
- 10. To delegate authority to the Monitoring Officer & Divisional Director, Legal and Democratic Services, to make any necessary changes to the Council's Constitution arising from decisions taken at this meeting and those required by law.

69 FUEL POVERTY - AGENDA MOTION FROM THE LIBERAL DEMOCRAT GROUP

An agenda motion was moved from the Liberal Democrat group regarding Fuel Poverty.

On a motion from Councillor Katie Hall, seconded by David Martin, it was

RESOLVED that:

This Council believes that:

- 1. Fuel poverty is a scourge on our communities; it harms the physical and psychological health, welfare and quality of life of our residents;
- 2. Local Authorities can play an important role in tackling fuel poverty through, for example, our new public health responsibilities, our strategic role in improving housing and our concern to encourage residents to contribute to the local economy and community.

And this Council therefore resolves to:

- 1. Work towards eliminating fuel poverty in Bath and North East Somerset;
- 2. Sign the "Local Authority Fuel Poverty Commitment" published by the End Fuel Poverty Campaign; and
- 3. Investigate practical schemes which could help residents who live in fuel poverty, such as working with partners to enable collective purchasing schemes for energy.

70 QUESTIONS, STATEMENTS, PETITIONS AND DEPUTATIONS FROM COUNCILLORS

Questions had been received from Councillors Brian Webber and Nathan Hartley. The responses had been circulated to Councillors and made available at the meeting (a copy of the document is retained in the Council's Minute book and linked online.)

The meeting ended at 9.25 pm

Chair(person)

Date Confirmed and Signed

Prepared by Democratic Services

Bath & North East Somerset Council				
MEETING:	Council			
MEETING DATE:	19 February 2013	AGENDA ITEM NUMBER		
TITLE:	Medium Term Service & Resource Planning 2013/14 – 2015/16, & Budget and Council Tax 2013/14			
WARD:	All			
	AN OPEN PUBLIC ITEM			
List of attac	hments to this report:			
Appendix	t 1 – Aims and Ambitions 2013/14 to 2015/16			
	• 2 – The Budget and Council Tax Proposal of the Cabir s a covering document, plus 8 Annexes	net 2013/14. This		
Annex 1	Draft Base Revenue Budget 2013/14 – individual service	e cash limits		
-	Section 25 of the Local Government Act 2003 - Chief Fin Adequacy of Balances and the Robustness of the Budg			
	Draft Capital Programme 2013/14-2017/18 including oth and programmes on an indicative basis - items shown in			
Annex 3i	Schools Planned Maintenance Programme 2013/14			
Annex 4	Minimum Revenue Provision (MRP) Policy			
Annex 5	Medium Term Service & Resource Plan (MTSRP) Savin	ıgs Detail		
Annex 6	Community Assets			
Annex 7	Council Policy Context			
Annex 8	West of England – City Region Deal			
Appendix Appendix Appendix	 A - Medium Term Service & Resource Planning Assumet A - Budget Setting Process – Advice of the Monitoring 5 – Council Pay Policy Statement 6 – Formal Council Tax Setting Resolutions (incorporation) 	Officer		
Parishes, Fi	re & Police) – TO FOLLOW			

1 THE ISSUE

1.1 This report presents the Cabinet's draft medium term financial plan, and revenue and capital budgets for the 2013/14 financial year together with a proposal for a Council Tax level for 2013/14.

2 RECOMMENDATION

- 2.1 That the Council approve:
 - a) The General Fund net revenue budget for 2013/14 of £123.359m with no increase in Council Tax.
 - b) That no Special Expenses be charged other than Town and Parish Council precepts for 2013/14.
 - c) The adequacy of reserves at Appendix 2 Table 9 with a risk-assessed level of £10.5m.
 - d) The individual service cash limits for 2013/14 summarised at Appendix 2 Table 5 and detailed in Annex 1.
 - e) That the specific arrangements for the governance and release of reserves, including invest to save proposals, be delegated to the Council's Section 151 Officer in consultation with the Cabinet Member for Community Resources and the Chief Executive.
- 2.2 That the Council include in its Council Tax setting, the precepts set and approved by other bodies including the local precepts of Town Councils, Parish Councils and the Charter Trustees of the City of Bath, and those of the Fire and Police Authorities.
- 2.3 That the Council notes the Section 151 officer's report on the robustness of the proposed budget and the adequacy of the Council's reserves (Appendix 2, Annex 2) and approves the conditions upon which the recommendations are made as set out throughout Appendix 2.
- 2.4 That in relation to the capital budget the Council:
 - a) approves a capital programme of £59.036m for 2013/14 and notes items for provisional approval in 2013/14 and the programme for 2014/15 to 2017/18 as shown at Appendix 2, Annex 3 including the planned sources of funding .
 - b) approves the Minimum Revenue Provision Policy as shown at Appendix 2, Annex 4
 - c) approves the Capital Prudential Indicators as set out in Appendix 2 Table 7.
- 2.5 That the Council notes the approach to Community Assets as set out in Appendix 2, Annex 6 and supports the progression of the Quick Wins identified in this Annex.

- 2.6 That the Council notes the Policy Development & Scrutiny review of Medium Term Service and Resource Plans and instructs the relevant officers to finalise these in consultation with the relevant Cabinet Member and in light of feedback from the PD&S reviews, and in line with the approved cash limits.
- 2.7 That the Council agree the Council's proposed pay policy statement as set out at Appendix 5.
- 2.8 That the Council approves the technical resolutions that are derived from the budget report, and all figures in that report, including the precepts for towns, parishes and other precepting bodies as set out in Appendix 6.

3 FINANCIAL IMPLICATIONS

3.1 The financial implications are contained within the body of the report.

4 COMMUNITY STRATEGY OUTCOMES

4.1 The framework for service and resource planning aims to ensure that resources are aligned to corporate priorities as set out in the Council's current Corporate Plan, which in turn reflects the Bath & NE Somerset Community Strategy.

5 CORPORATE IMPROVEMENT PRIORITIES

5.1 The framework for Service and Resource planning aims to ensure resources are aligned to corporate priorities and improving the delivery of services and cross-Council priorities, linking performance with resources and value for money.

6 THE REPORT

- 6.1 In this document the Cabinet sets out the following:-
 - Its medium term service and resource planning financial assumptions (Appendix 3) which set the basis for the draft budget proposal for 2013/14.
 - Its draft budget proposal for 2013/14 (Appendix 2). This provides the detail of the first year of the Medium Term Service and Resource Plans and recommends revenue and capital budgets for 2013/14, together with capital commitments for future years, and recommends a level of Council Tax for 2013/14.
- 6.2 The budget proposal builds on the prudent financial management of the Council and is designed to maintain front line services as far as possible whilst recognising the significant financial challenge facing the public sector. The budget proposals include:
 - A net £2.5m or 2.0% decrease in the non-schools budget (after allowing for the technical adjustments required as a result of changes in local government finance funding reforms) to £123.359m

- An overall increase in the DSG, estimated at £2.5m with a total funding of £116.6m (including academies). The Dedicated Schools Grant (DSG) increase compared to 2012/2013 is complicated by the transfer of additional responsibilities into the DSG from other funding routes. The majority of the additional funding relates to the transfer of funding for 2 year olds and the funding for post 16 special needs provision, that have transferred from the Early Intervention Grant and the Education Funding Agency respectively. Taking account of this, the overall total represents a cash freeze per pupil compared to the previous year. However as schools convert to academies the Department for Education (DFE) recoup the DSG payable to the Local Authority in order to make payments direct to the academies. The DFE estimate the recoupment will be £41.1m in 2013/2014 leaving £75.5m payable to the Local Authority.
- A freeze in the Council's level of Council Tax, which excludes Polices, Fire and Parish precepts.
- These budget proposals are set out in detail at Appendix 2.
- 6.3 It is important to be clear on the process to be followed in setting the 2013/14 Budget. The Monitoring Officer has given specific guidance which is set out at Appendix 4, and in particular the need for the Council to approve a balanced budget.
- 6.4 The Monitoring Officer has also highlighted the implications arising if it does not prove possible for the Council to set a budget at its meeting on 19th February and any decision having to be deferred until the reserve date on 28th February. This includes potential delays to the Council Tax billing process.
- 6.5 The Council's pay policy statement as required under The Localism Act 2011 legislation is set out at Appendix 5.
- 6.6 The Budget proposal set out at Appendix 2 also includes specific Annexes relating to the Council's approach to Community Assets and the on-going development of the West of England City Region Deal.

7 RISK MANAGEMENT

- 7.1 The report author (the Council's Section 151 Officer) has fully reviewed the risks related to this report and recommendations and consulted with the Lead Cabinet member.
- 7.2 In addition Appendix 2 to this report includes (at Annex 2) the Section 151 Officer's assessment of the Robustness of Estimates and Adequacy of Reserves. One of the considerations taken into account is the Directors' Review of Robustness of Estimates and Budget Risks/Sensitivities and the Corporate Risk Register. This is completed by all Directors in respect of their own services.

8 RATIONALE

8.1 The rationale for the recommendations is contained in the supporting paper to this report.

8.2 The Council's Section 151 Officer is the Divisional Director - Finance. This report reflects information already presented to Policy Development and Scrutiny by the Director. As Section 151 Officer his duties include ensuring a prudent and balanced budget is set on time which properly takes into account the financial constraints and risks facing the Council.

9 OTHER OPTIONS CONSIDERED

9.1 The supporting paper and appendices also contain the other options that can be considered in making any recommendations.

10 CONSULTATION

- 10.1 Meetings have been and will continue to be held with staff, trades unions and other stakeholders during the development of service and resource plans which have fed into this budget. This has again included three Budget Fair's (Bath, Keynsham and Timsbury), enabling cross service consideration of the range of proposals by a range of stakeholders.
- 10.2 Representatives of the business community were engaged in these consultations as part of the Budget Fair.
- 10.3 Comments received from consultation, including the Budget Fair, Policy Development and Scrutiny Panels and Trade Unions have been provided for consideration by the Cabinet.

11 EQUALITY

11.1 The Council is committed to ensuring that our financial decisions and the budgetary processes are subject to proportionate equality analysis. Bath & North East Somerset Council can demonstrate that it has analysed its decisions from an equality perspective by evidencing our analysis using a single template equality impact assessment toolkits. (B&NES was commended for this work and used as an exemplar authority for our proportionate approach to equality by the LGID.) Thus the Council has a written record of the equality considerations that have been taken; which include a consideration of the actions that would help to avoid or mitigate any negative impacts on people of particular protected groups. Our decisions have a stronger evidence base (through effective use of equality mapping and Joint Strategic Needs Assessment) assisting to make our decision-making processes more transparent.

12 ISSUES TO CONSIDER IN REACHING THE DECISION

12.1 These include Social Inclusion; Customer Focus; Sustainability; Young People; Equality; Corporate; Impact on Staff and Other Legal Considerations such as the requirement to set a budget and Council Tax.

13 ADVICE SOUGHT

13.1 The Council's Monitoring Officer (Council Solicitor) and Section 151 Officer have had the opportunity to input to this report and have cleared it for publication.

Contact person	Tim Richens, Divisional Director – Finance (01225) 477468			
Sponsoring Cabinet Member	Councillor David Bellotti, Cabinet Member for Community Resources			
Background papers	Medium Term Service & Resource Planning reports to Policy Development & Scrutiny Panels			
	Equalities Assessment –Report to Resources PDS 11th February 2013			
	Consultation Response Summary – Report to Resources PDS 11 th February 2013			
	Council Response to DCLG Document "50 Ways to Save" – Report to Resources PDS 11th February 2013)			
	Financial Settlement 2013/14 to 2014/15, CLG website			
Please contact the alternative format	e report author if you need to access this report in an			

Our Aims and Ambitions for Local People Explained 2013/14 to 2015/6

Bath & North East Somerset Council strives to deliver excellent services to local people and make our area *the* place to live, invest, and visit in the UK.

The foundations for how this vision will be achieved are a sensible management of our budget, effective partnership working with local, national, statutory and voluntary organisations, together with strong leadership from both elected councillors and our staff.

There are challenges the Council and its partners must overcome. These include Government funding reductions on a scale unprecedented since the Second World War to address the Country's financial deficit, the effects of the national recession, and demographic change.

This three-year budget reflects what matters most for local people as we plan for an area which offers opportunity and positive lives for everyone, whilst ensuring our most vulnerable residents are given the support that they need. In putting people first and communities at the heart of everything we do, our priorities are:

- Protecting frontline services;
- No increases in Council Tax;
- Investing in homes and jobs for local people.

Supporting the most vulnerable people and families

We want people to remain independent and living in their own home. The number of older people will grow incredibly over the next 20 years. From now until 2035, the over-65 population is likely to increase by 50%, and the over-85 population is likely to increase by 110% over the same timescale.

The £260 million over the next three years that we will spend on adult social care must be focused on the most vulnerable people, promoting their independence through improved community services to ensure that residents, where possible, stay at home. We must also recognise the increasing need for specialist care for people with dementia and those with the most complex needs, such as those with autism. We will continue to work closely with our partners to enable vulnerable people in our communities across Bath and North East Somerset to live fulfilling and rewarding lives.

This will mean refocusing some areas of our budget on people with the highest needs. An extra £3.25 million has been set aside for this purpose over the next three years.

In common with the national picture the Council is also experiencing significant population growth, which coupled with pupils generated from new housing developments, like the MOD sites, will require the provision of new classrooms at existing primary schools in areas where pressure is greatest. The capital funding available for schools will be used to provide an additional 375 places over the next three years.

The money available for Children's Services will be targeted to address the needs of vulnerable young people and families. Our Children's Centres and Youth Centres will strive for excellence to support those people most in need and give them opportunities for their future. We will be working hard with families who are struggling to help them to have successful lives.

New responsibilities for public health from April 2013 will provide us with a greater opportunity to offer people information and advice about how to improve their quality of life, whether that be through health, fitness, or diet.

A better quality of life in our communities

The Council must play a strong community leadership role in creating neighbourhoods where people are proud to live and we want our residents to make a difference to their own quality of life, as well as their health and wellbeing. We cannot achieve this on our own.

We will be investing £500,000 in cycle routes, completing the £32 million Bath Transportation Package and a new community cycle track at Odd Down. But local people must make the lifestyle choice to use sustainable methods of transport.

There will be also be times when residents are in the best position to make a difference to the place where they live. For instance, our eight library branches will be supported by our Community Library Programme, encouraging more people to play a part in running their library.

Taking this principle a step further, the Council is looking to transfer some of its assets to local communities if we think they would be better run by local people.

We will also be working closely with local communities to encourage more environmentally sustainable behaviour through promoting our Biomass Energy Efficiency Fund, Schools Energy Invest to Save Fund, and continuing our efforts to reduce the Council's own carbon footprint.

New places to live and work

We want to build a strong economy supporting the community improvements necessary to sustain new houses – including affordable homes – and new jobs and so that local people can find a wider range of jobs, residents can share higher levels of prosperity, and the children and grandchildren that our excellent schools educate don't have to move away to progress the careers they aspire to.

Our budget plans include:

• Progressing the Bath 'City of Ideas' Enterprise Area through a combined total of over £27 million to regenerate key sites along the River Avon with employment space and places to live, including affordable homes;

- Completing a £33 million revamp of Keynsham town centre that will encourage private sector investment into the community;
- Regenerating communities, including Radstock where an upgrade of the road network will support new homes and jobs, and London Road, Bath;

These plans will take full advantage of our place as a leading location for the digital, creative, and technology sectors.

We will be proactive alongside our partners to develop worker skills, improve connections between employers and job seekers, and support the network of apprentices, interns, and undergraduate placement schemes. The area's greatest resource for economic prosperity is local people.

Making the best of opportunity

The Council values the views of local people when making decisions about the services we provide. We listen and act upon the things residents tell us are important. The coming years will test the area to address the challenges we face – but they also provide significant opportunities. The Council looks forward to working with local communities every step of the way.

This page is intentionally left blank

THE BUDGET PROPOSAL OF THE CABINET 2013/2014

Budget Summary

Headlines

Despite a difficult national economic position and its impact on local government finances, our proposed budget protects high priority frontline services and invests in the future of our area, promoting new homes and jobs.

We are also proposing that Council Tax is frozen in 2013/14 for the third consecutive year, recognising the on-going pressure on household incomes. The proposed band D Council Tax for Bath & North East Somerset Council next year is £1,201.85 which represents no change on the previous year (no increase).

The proposed net revenue budget for Bath & North East Somerset Council next year, 2013/14, is £123.359m, a net cash reduction of £2.474m on the previous year. This reflects the impact of significant cuts in government grant funding amounting to 9.4% for 2013/2014.

The Budget Context

The proposed Council Budget for 2013/2014 presents a detailed Medium Term Service and Resource Plan for the three-year period from 2013/2014 through to 2015/2016. This will enable the Council to take a planned and structured approach to meet the significant financial challenge facing the Council.

2013/2014 represents the third year of financial planning prepared in the context of the Government's Comprehensive Spending Review (CSR) announced in October 2010. This CSR included a deficit reduction programme with 24% cuts to local authority spending spread over the four-year period from 2011/2012 to 2014/2015. Based on the most recent Local Government Finance Settlement we now know the equivalent reductions for this Council will actually be 40% over this same period. This reflects the below average Settlement received by this Council in each year of the CSR period and the Council's position as one of the lowest funded unitary councils in the country.

The table below illustrates the scale of the government grant funding reductions for the Council compared to the overall reductions indicated within the CSR 2010.

	2011/2012	2012/2013	2013/2014	2014/2015	Total
CSR Indicative Funding Reduction	-10.7%	-6.4%	-0.9%	-5.6%	-23.6%
Actual B&NES Funding Reduction	-13.5%	-8.3%	-9.4%	-8.7%	-39.9%

Table 1: Reduction in Government Grant Funding

It is also clear from the Government's Autumn Budget Statement in December 2012, that the reductions set out in this previous CSR will not be sufficient to meet the Government targets to reduce the fiscal deficit, as the on-going impact of economic uncertainty both across Europe and indeed worldwide, means the UK economy continues to fall short of previous expectations.

A range of significant changes impacting on local Government Finance will also take effect from 1 April 2013:

- The needs based Formula Grant funding system is replaced with a combination of localised Business Rates and a top up grant to be known as Revenue Support Grant.
- The new Localised Business Rates (National Non Domestic Rates) will provide for the Council to retain up to 49% of local business rates going forwards to incentivise growth.
- Responsibility for setting Council Tax Benefit passes to local authorities from 2013/2014 in the form of the new Local Council Tax Support Scheme. The Council considered and approved the revised arrangements for this scheme in November 2012.
- With effect from the 1st April 2013 the Government is abolishing elements of the Social Fund which is currently administered by the Department for Work and Pensions and allocating funding to Local Authorities based on their estimated awards for Community Care Grants and Crisis Loans.
- Public health responsibility and related services will pass to the Council from April 2013 from the PCT supported by separate ring fenced grant funding from the Department of Health.
- The full implementation of planned changes to Government Funding for the Local Education Authority and Academies through the Education Services Grant (ESG) will go ahead in 2013/14.
- Early years funding for 2 year olds will move from the LEA into the Dedicated Schools Grant together with responsibility for careers advice and guidance.

The implications for this Council will require estimated savings of at least £30M over the next three financial years (2013/2014 to 2015/2016).

The Budget Proposal

Each Directorate of the Council has prepared a Medium Term Service and Resource Plan (MTSRP) covering the full three-year period from 2013/2014 to 2015/2016. Each of these plans has been considered by the relevant Policy and Development Scrutiny (PDS) Panel throughout November 2012.

The MTSRP's set out the specific service and resource requirements for 2013/2014 and beyond, including growth requirements and savings proposals, and allowing for the anticipated financial implications of the grant settlement. Feedback from the individual PDS panels has been considered in arriving at the proposed Budget.

The proposed Budget for 2013/2014 recognises the very difficult financial challenge now facing the whole of the public sector and the increasing need to prioritise resources. The following principles have been used to support this:

- Protecting frontline services;
- No increases in Council Tax;
- Investing in homes and jobs for local people.

There are no longer the available resources to deliver the full range of services that have been provided in the past. New legislation and demographic changes similarly demand clear prioritisation and new approaches. This increasingly means difficult choices.

The development of the Budget has moved away from setting targets and budget top slices based on historic spending to and approach more focussed on prioritisation supported where appropriate by zero based budgeting. This approach has included: -

- Ensuring only essential cost pressures are taken into consideration, challenging all proposals for inflationary increases and additional spending.
- A continued focus on achieving efficiency savings within and across service areas.
- Maximising savings achieved through the continued development of the Change Programme with projects like Customer Services and Procure to Pay.
- Seeking to increase income from new and existing sources. Developing and investing in a diversified income base to help protect the Council from reductions in Government funding.
- Minimising costs of borrowing utilising Council cash flow balances where appropriate to provide funding for capital projects.
- Exploring opportunities to support Communities to enable them to be more resilient and self-sustaining (See Annex 6).
- Making better use of Council Assets, particularly council land and property, to reduce running costs and provide capital receipts.
- Where Government is cutting its grants to local authorities, or other external sources of funding are being reduced, these savings requirements may need to be passed on to the relevant service.

The proposed Budget also recognises service specific spending pressures and Council priorities that need to be addressed including impacts of national policy changes totalling some £9.2m. The most significant of these include:

- £1.1m for increasing demands on Adult Social Care and Health services.
- £0.1m for increased demand for Children's care services.
- £3.2m for inflationary costs particularly for care placements and external service contracts.
- £1.4m for the impacts of government legislation and initiatives, including reduced planning fees, waste taxes and treatments, and concessionary fares.
- £0.2m for increased school days in 2013/2014 due to term timings.
- £1.3m for impacts of economic downturn and increased competition.
- £1.0m to offset one-off funding used to support the 2012/2013 budget.

- £0.2m for capital financing.
- £0.2 m for IT licensing costs.
- £0.3m for other improvement priorities

Taking account of the reductions in government grant funding and the pressures outlined above, requires the identification £30m of budget savings over the three-years from 2013/14 to 2015/2016 as part of the proposed Budget. As part of this overall picture the specific savings required for 2013/2014 total £11.63M and these are set out in full at Annex 5 and can be summarised as follows:-

Table 2: 2013/14 Savings Analysis

Description	£'000	
Change Programme & Efficiency Savings	5,634	
Increases in Income from fees, charges and other grants	3,887	
Service Reductions	2,109	
Total Savings	11,630	

The majority of savings are being delivered from efficiencies and the Council's change programme. This has enabled prioritised service reductions to be kept to an absolute minimum. Indeed the reductions to services represent only around 18% of the total savings and around 1% of the gross Council budget excluding Schools. Details of the specific savings items and the impact on service areas are included within individual MTSRP's.

Capital Investment

The proposed Capital Programme provides for £151M of investment from 2013/2014 whist recognising the desire to reduce planned levels of external borrowing. Key elements of the programme include:-

- Continued investment in regeneration schemes including Keynsham Town Centre, Radstock and London Road Bath
- Delivery of the Bath Transport Package following the confirmation of funding support from the Department of Transport.
- An on-going contribution towards enablement of affordable housing across B&NES area, building on the commitment to the Bath Western Riverside Development.
- Significant investment in highways maintenance and transport infrastructure with on-going support for the provision of 20mph zones and new funding for improved cycle routes.
- Improvements to school buildings and facilities together with provision of additional capacity particularly for primary school places.
- Progressing the Bath Enterprise Area including investment to regenerate key sites for development.
- Supporting Energy Efficiency Schemes including a Schools Energy Invest to Save Fund and a Biomass Energy Efficiency Fund.

Schools Funding

Schools are funded by the Dedicated Schools Grant (DSG) which is initially allocated to the Council by the Department for Education. The DSG supports all expenditure in schools (who set their own budgets) and the activities that the Council carries out directly for schools. It does not cover the statutory responsibilities the Council has towards parents. These responsibilities are funded through the Councils main revenue funding and included as part of the proposed Budget.

The Dedicated Schools Grant (DSG) increase compared to 2012/13 is complicated by the conversion of several schools to academies, and the transfer of additional responsibilities into the DSG from other funding routes. The overall increase in the DSG is estimated for 2013/2014 at £2.5m with total funding of £116.6m (including academies). The majority of the additional funding relates to the transfer of funding for 2 year olds and the funding for post 16 special needs provision, that have transferred from the Early Intervention Grant and the Education Funding Agency respectively. Taking account of this, the overall total represents a cash freeze per pupil compared to the previous year.

Whilst the Dedicated Schools Grant (DSG) is being allocated to Local Authorities on a cash freeze basis. The Pupil Premium allocations for maintained schools are increasing from £623 to £900 per Free School Meals pupil in 2013-14. This increase will bring the total payable to B&NES schools to £2.448m in 2013-14, an increase of £760k. This represents approximately a further 1% increase in overall resources for schools.

As schools convert to academies the DFE take back the element of DSG payable to the local authority in order to make payments direct to the academies. The DFE estimate of this will be £41.2m in 2013/14 leaving £75.4m payable to the Council.

This recoupment by the DFE is based on 10 secondary, 1 special and 2 primary academies in 2013/14. It is difficult to assess whether there will be more schools converting to academies over the next year.

Further budget changes to council funding are generated by the creation of the Education Services Grant (ESG) to replace the recoupment process for resources for academies taking on the statutory responsibilities of the council. Schools converting to academies and the Local Authority will be allocated resources that were originally part of the local government settlement. The grant allocations are based on pupil numbers in the establishments with academies being funded directly and the Local Authority receiving the allocation for all maintained schools. As schools convert to academies the resources allocated to the Local Authority under the ESG would diminish on a per pupil basis.

<u>Council Tax</u>

The local government financial settlement included an announcement of financial support for councils who freeze their Council Tax for next year at the current level (i.e. a zero increase). This financial support (in the form of a grant) for 2013/2014 is stated as being included in baseline funding for 2014/15 and beyond. The income from the grant is equivalent to a 1% increase in council tax.

The proposed Council Budget utilises this grant and provides for a zero increase in Council Tax for 2013/2014.

Business Rates

The Council has historically been responsible for the collection of Business Rates (National Non Domestic Rates) effectively passing the funds raised directly through to national government. From 1 April 2013 as part of the new funding arrangements for local councils, up to 49% of the Business Rates will be retained locally. The aim of this new system will be to provide an incentive to local authorities to stimulate and encourage business growth in their area. However the Council will also share in the risk of non-payment, business rate appeals and business closure or failure. A national safety net is in place to provide some protection although this will only operate once business rates have reduced by 7.5%.

For the implementation of this new system the Government has set an initial baseline level of business rates to be collected. This baseline was initially stated to be set based on a 5 year average and this basis had been reflected in the Council's financial plans. A very late change by the Government to a 2 year average, effectively pushed this baseline higher for the Council adding a further £1M funding pressure to the Council's financial plans.

This system will not be reset now until at least 2020 and therefore the emphasis on business growth will be key to help support funding for frontline council services in the future. Immediate challenges do exist with the closure of the MOD sites in the Council area and the significant loss of business rates that will arise initially.

<u>Reserves</u>

The Council is maintaining its un-earmarked reserves at the appropriate risk assessed level. Some of these reserves (above a minimum level) will continue to be utilised on an Invest to Save basis. The assessed risk which has been reviewed as part of the Budget process, suggests reserves of £10.5M for 2013/2014 with a minimum level of £6M. These are deemed appropriate based on current financial planning assumptions.

Earmarked reserves are set aside to fund specific future commitments and have been fully reviewed as part of the budget process. The most significant of these are:-

- Revenue Budget Contingency £491K (to meet in-year revenue budget variations)
- Medium Term Financial Challenge Reserve £2,651K (to support the existing Change Programme)
- Restructuring Reserve £5,598K (to meet future severance costs arising from the MTSRP)
- Affordable Housing Reserve £3M (to fund the affordable housing contribution for Bath Western Riverside)
- A new Financial Planning Reserve of £6.6M is created following the review of reserves and the removal of the Capital to Revenue Reversion Reserve and the Capital Financing Reserve. Up to £3.8M of this is being allocated as one-off headroom in 2013/2014. The balance will be allocated as part of the Budget each year to support the specific medium term financial proposals and priorities of the Council.

The balances shown are for 2013/2014 and these reserves are anticipated to be fully committed over the medium term.

The Medium Term

The Government's Autumn Statement in December 2012 confirmed the need for public sector spending deficit reduction measures to continue beyond the period of the current CSR and at least until 2018. The Statement also confirmed an additional 2% average reduction for local government funding in 2014/15 over and above the position set out in the CSR.

The Council has, as far as possible, reflected this position in this proposed Budget covering the period until 2015/2016. We anticipate a further spending review in spring 2013 which may provide further clarity for the final year of this MTSRP period.

Structure of the Budget Proposal

Section 1 sets out the approach to the revenue and capital budget and the build-up of the recommended revenue budget for 2013/2014. Annex 1 provides the breakdown of the Budget for 2013/2014.

Section 2 sets out the position for future years 2014/2015 and 2015/2016 taking account of proposals developed in the medium term service and resource plans, in light of known pressures and assumptions about levels of funding. It also includes details of how the Council Change Programme will contribute to meeting the financial challenge.

Section 3 sets out the recommended capital programme for 2013/2014 including the indicative capital programme through to 2017/2018. Annex 3 provides more detail.

Section 4 sets out the current position on revenue balances taking into account the proposals for prudent use of reserves.

Section 5 sets out the implications of the revenue budget for Council Tax levels for 2013/2014.

	2013/14	2014/15	2015/16
Revenue Budget Funding:	£m	£m	£m
Council Tax	71.342	71.142	71.292
Council Tax Freeze Grant	0.740	0.740	0.740
Revenue Support Grant	31.106	25.943	21.243
Retained Business Rates (after tariff)	20.262	20.845	20.845
Reserves & Collection Fund Surplus	-0.091	0.000	0.000
Total Funding	123.359	118.670	114.120
Net Revenue Budget Spend	123.359	118.670	114.120
Capital Programme – for approval	59.036	19.972	2.631
Capital Programme - for provisional approval (subject to)	20.805	24.319	9.824

Note: Revenue Budget figures beyond 2013/2014 are indicative only.

Section 1 – The Revenue Budget for 2013/2014

Introduction & Process

The proposed Budget for 2013/2014 to 2015/2016 has been prepared using the Council's Medium Term Service and Resource Planning (MTSRP) process.

The purpose of the Budget is to allocate financial resources across the Council to deliver services to the community to required standards and performance targets. The detail of what is spent must therefore be seen within the context of the Council's corporate objectives.

Strategic Directors, in consultation with their Cabinet Members, have developed Medium Term Service and Resource Plans covering the years 2013/2014 to 2015/2016. These Medium Term Service and Resource Plans set out the financial pressures, savings and efficiency proposals that each service will be taking forwards to meet the overall level of resourcing available to the Council. This includes outlining the impacts of each of the proposals both in terms of service delivery and on staff.

The Medium Term Service and Resource Plans have been reviewed by the relevant Policy Development and Scrutiny Panels and feedback from these Panels has been considered as part of the process. A consolidated schedule of all the MTSRP savings is attached at Annex 5. Copies of the individual plans and feedback from the PDS Panels are available as background papers.

The Cabinet have now had the opportunity to further consider proposals from officers which develop the medium term financial planning and take account of the on-going work of Strategic Directors and the comments from PD&S panels. In that context the Cabinet has developed the Budget proposal to reflect this position. The Cabinet has also considered feedback from consultation with the, community, Trade Unions, and a range of other partners / stakeholders, particularly from the Council's three Annual Budget Fairs held in early November 2012.

Approach to the Budget

Corporate Assumptions

The Budget has been built up using a range of corporate financial planning assumptions for areas of income and expenditure which are subject to variation.

Assumptions relating to specific areas of cost increases or demand for individual services are separately identified within the Medium Term Service and Resource Plan for each service.

Appendix 3 sets out in more detail the corporate financial assumptions which underpin the Budget and the key financial planning assumptions are summarised below:

- A pay increases of no more than 1% in each year of the MTSRP 2013/14 to 2015/16.
- Continued low rates of interest from treasury management cash investments of 0.75% per annum. The Bank of England base rate is not expected to rise from the current level of 0.5% during the MTSRP period.
- A small 0.5% increase in the level of the employers' contribution to the Local Government Pension Fund following the next Actuarial Review in 2013, reflecting the impacts of wider changes to the Scheme.
- Balanced budgets are achieved for 2012/2013 with no related on-going funding pressures.
- No general inflationary provision specific inflation has been provided and identified as growth within service areas based on specific service circumstances and contractual commitments. These are set out in detail in the MTSRP's and provide for 2% inflation except Adult Social Care where the target is 1.75%.
- That capital funding is provided from Capital Receipts and internal cash flow. New borrowing will be kept to a minimum subject to market conditions.

Fees and Charges

The Council is reliant on a range of income streams from fees and charges, many of which are either set by statute or are subject to defined calculations under statutory guidance. Where the Council has discretion, the decisions are delegated to Officers, in consultation with the relevant Cabinet member, and increases in fees and charges are generally in line with the increase in the costs of the relevant service.

The Budget proposal and related Medium Term Service and Resource Plans assume any proposed changes in fees and charges will be in accordance with this policy with the following key exceptions:-

- **Car Parking Charges** the MTSRP includes some specific proposals for car parking charges although beyond these actions, no general increase is proposed in the rates for car parking during 2013/2014.
- **Park and Ride Charges** charges are now set by the operator within agreed contractual limits.
- **Planning Fees** assumption that fees will increase to the maximum level set out in the relevant Government Fee Schedule.

Government Grant

The Government originally set out its plans for tackling the significant budget deficit, in the Comprehensive Spending Review on 20th October 2010. This deficit reduction programme indicated 24% cuts to local authority spending over the Spending Review period from 2011/2012 to 2014/2015 with a significant element front loaded to the first two years.

The subsequent Local Government Finance Settlement announced in December 2012 showed the Council's funding baseline for 2013/2014 reducing by a further 9.7% and then by a further 8.7% in 2014/2015. This equates to a 40% reduction in funding for the Council, well in excess of the average set out in the CSR.

The funding baseline itself has been reassessed to take account of the start-up funding baseline for the new Localised Business Rate element. This Business Rate baseline position is fixed and only uprated in line with inflation for at least seven years.

The Government's Autumn Statement in December 2012 reflected the on-going market concerns surrounding the national debt position of the UK and a number of Eurozone countries. In particular the statement confirmed the need for public sector spending deficit reduction measures to continue beyond at least until 2018. The Statement also confirmed a ceiling on public sector pay of 1% per annum for the next two years.

This settlement confirms that B&NES Council will remain one of the lowest funded unitary authorities with the position set out for 2013/2014 now forming the baseline funding until at least 2020 when the system is next due to be reset. The Council has the 6th lowest spending power per dwelling when comparing all Unitary Authorities.

The grant funding baseline has been updated to remove the funding for LEA services which will now be paid directly to the LEA and Academies as the Education Services Grant (ESG) on a per pupil basis. We had anticipated this adjustment although the rate of schools transferring to Academies (particularly secondary schools) does present a funding challenge for the Council.

The funding baseline also reflects the change in responsibility for setting Council Tax Benefit which passes to local authorities from 2013/2014 in the form of the new Local Council Tax Support Scheme. The Government grant supporting this is being reduced by approximately 10% for 2013/2014 and this is reflected in the provisions within the new scheme which, was considered and approved by the Council in November 2012.

The New Homes Bonus Scheme which provides funding for new homes brought into use and included in the Council Tax Base, was provisionally confirmed at a total of $\pounds 1.977m$ for 2013/2014. This is the third year for funding for New Homes Bonus and includes an additional $\pounds 0.759m$ for 2013/2014. The New Homes Bonus scheme runs for 6 years from the year of initial allocation. This funding is un-ring-fenced and available for allocation as part of the overall revenue budget.

The Settlement included provision of financial support from the government for councils who freeze their Council Tax for next year – 2013/2014. The grant is equivalent to a 1% Council Tax increase and is confirmed as being including in the baseline funding calculations.

The Settlement announcement confirmed the requirements for Council tax referendums where any Council (excludes parishes again in 2013/2014) increasing council tax by more than 2% (or £5 for qualifying police and fire authorities). This is clearly irrelevant for councils accepting the council tax freeze grant.

Retained Local Business Rates

From 2013/2014 the new arrangements for funding of local government will provide for up to a maximum of 49% to be retained locally.

The actual amount to be retained is dependent upon a number of factors including a start-up baseline assessment, a top-up or tariff payment subject to the level of the baseline and a levy on future growth to support a national safety net.

The setting of the baseline is key and for the Council a late change in the bases if calculation from a 5 year to a 2 year average, effectively increased the Council's annual tariff payment to the Government by over £1M.

Going forwards the Council will retain 49% of the Business Rates less a tariff payment of \pounds 9.468m to the government based on the initial baseline. 49% of all future growth in business rates will be retained by the Council less a levy payment of 31p in the \pounds to provide for a national safety net.

Under this system the Council benefits from new growth but also bears 49% of the cost of reductions in business rates from for example, appeals, business closures or relocations, charitable reliefs etc. A national safety net will be provided where business rates reduce by over 7.5%, equivalent to approximately £2M for the Council.

The initial years of this new system will be very challenging as some large business closures will impact on the Business Rates collected, particularly MOD sites and Somerdale.

As part of the proposed Budget, reasonable assumptions have been made for likely future Business Rates although given these new arrangements have not been tested and current economic conditions continue to be challenging, there remain a significant risk associated with future receipts.

The Council has been seeking further changes to some elements of the new system, including the baseline calculation and the treatment of historic appeals, which may provide some flexibility in the overall Business Rates Collection Fund going forwards.

Social Fund

With effect from the 1st April 2013 the Government is abolishing elements of the Social Fund which is currently administered by the Department For Work and Pensions and allocating funding to Local Authorities based on their estimated awards for Community Care Grants and Crisis Loans.

Bath & North East Somerset Council has been awarded £249,260 for next financial year as its share of the Social Fund. There are no rules or restrictions on how this money should be used but it is allocated for the purpose of local Welfare Provision.

In addition to the Programme funding there is also an amount of £52,670 to support any administrative costs of a new scheme

The Council has already indicated that a scheme should target those most impacted by the introduction of the Local Council Tax Support Scheme. The Cabinet will shortly review a report on how to best target and administer these funds to assist the most vulnerable members of our community impacted by the Welfare Reforms.

Schools Funding

Schools are funded by the Dedicated Schools Grant (DSG) which is initially allocated to the Council by the Department for Education. The DSG supports all expenditure in schools (who set their own budgets) and the activities that the Council carries out directly for schools. It does not cover the statutory responsibilities the Council has towards parents. These responsibilities are funded through the Councils main revenue funding and included as part of the proposed Budget.

The Dedicated Schools Grant (DSG) increase compared to 2011/12 is complicated by the conversion of several schools to academies, and the transfer of additional responsibilities into the DSG from other funding routes. The overall increase in the DSG is estimated for 2013/2014 at £2.5m with total funding of £116.6m. The majority of the additional funding relates to the transfer of funding for 2 year olds and the funding for post 16 special needs provision, that have transferred from the Early Intervention Grant and the Education Funding Agency respectively. Taking account of this, the overall total represents a cash freeze per pupil compared to the previous year.

Whilst the Dedicated Schools Grant (DSG) is being allocated to Local Authorities on a cash freeze basis. The Pupil Premium allocations for maintained schools are increasing from £623 to £900 per Free School Meals pupil in 2013-14. This increase will bring the total payable to B&NES schools to £2.448m in 2013-14, an increase of £760k. This represents approximately a further 1% increase in overall resources for schools.

As schools convert to academies the DFE take back the element of DSG payable to the local authority in order to make payments direct to the academies. The DFE estimate of this will be \pounds 41.2m in 2013/14 leaving \pounds 75.4m payable to the Council.

This recoupment by the DFE is based on 10 secondary, 1 special and 2 primary academies in 2013/14. It is difficult to assess whether there will be more schools converting to academies over the next year.

Further budget changes to council funding are generated by the creation of the Education Services Grant (ESG) to replace the recoupment process for resources for academies taking on the statutory responsibilities of the council. Schools converting to academies and the Local Authority will be allocated resources that were originally part of the local government settlement. The grant allocations are based on pupil numbers in the establishments with academies being funded directly and the Local Authority receiving the allocated for all maintained schools. As schools convert to academies the resources allocated to the Local Authority under the ESG would diminish on a per pupil basis.

Resource Allocation including Use of Recurring and One-Off Funding Headroom

The development of the proposed Budget has moved away from setting targets and budget top slices based on historic spending to an approach more focussed on prioritisation supported where appropriate by zero based budgeting. This approach has included: -

- Ensuring only essential cost pressures are taken into consideration, challenging all proposals for inflationary increases and additional spending.
- A continued focus on achieving efficiency savings within and across service areas.
- Maximising savings achieved through the continued development of the Change Programme with projects like Customer Services and Procure to Pay.
- Seeking to increase income from new and existing sources. Developing and investing in a diversified income base to help protect the Council from reductions in Government funding.
- Minimising costs of borrowing utilising Council cash flow balances where appropriate to provide funding for capital projects.
- Exploring opportunities to support Communities to enable them to be more resilient and self-sustaining.
- Making better use of Council Assets, particularly council land and property, to reduce running costs and provide capital receipts.

The scale of the savings required over the next three years of the MTSRP period, is such that the Council will need to prioritise services and whilst every effort will be made to protect essential frontline services for local people, this will inevitably lead to reductions in some service areas which are considered a lower priority.

The proposals put forward in the MTSRP's provide clear actions to address the majority of the estimated £30M funding shortfall faced by the Council over this period, including a balanced Budget for 2013/2014. Future years will still be subject to government funding announcements and it is highly likely that further savings will be required.

Table 4 in this report shows how the budget rolled forward from 2012/2013 has then been built up with total growth and savings identified by services as set out in MTSRP's.

Budget Headroom

Each year the Council considers how any available headroom within the budget should be allocated. This headroom, which may be recurring or one-off in nature, is created from a number of sources and can only be finalised once all assumptions and calculations are completed for the proposed Budget. These sources include:-

- Changes in financial planning assumptions
- Variations in grant settlement
- Full year effect of prior year savings
- Adjustments to corporate finance items
- One-off funding provisions and variations, for example the council tax collection rate.

The Budget includes the following allocations of available headroom to meet specific commitments and priorities:

On-going Headroom Allocations - Nil

No on-going headroom is available for allocation at this stage over the MTSRP period. This is a reflection of the scale of the financial challenge and the need to utilise all available funding sources to minimise reduction in services.

One-off Headroom Allocations - £3,815K

These allocations are to be made from the Financial Planning Reserve.

- £3,000K to be allocated to establish an Earmarked Reserve to meet the costs arising from the likely timescale to implement recurring budget savings in 2013/2014
- £180K to meet additional costs of Children in Care Placement costs should they exceed baseline budget assumption.
- £150K to meet potential set up costs of a Tourism Levy.
- £150K to meet costs associated with a Trading Company acquisition.
- £100K to fund reduced savings in childrens preventative services.
- £50K to provide revenue maintenance grants to property owners in support of the London Road Regeneration Project.
- £60K for a contribution to the voluntary sector in order to provide business startup loans to the self-employed.
- £50K to establish a hardship fund for landlords as a result of the changes to the Council Tax Discount Scheme for empty properties (as agreed by Council 8th November 2012)
- £40K to provide a webcasting trial of Council Meetings for one year.
- £35K contribution to fund the community engagement work and advice line in respect of the Retrofit and Green Deal delivery partnership.

In addition to these specific proposals, under the Council's Invest to Save Scheme, the Section 151 Officer in consultation with the Cabinet Member for Resources, may authorise funding for robust and credible invest to save proposals from reserves (i.e. in the short term creating a 'negative ear-marked reserve' which is then repaid over time, usually 3 years, from the related savings.

Robust and credible 'invest to avoid' proposals (where investment can avoid future costs), can also be considered, but in addition there needs to be specific provision within business cases to replenish the reserves over a 3 year period.

Summary of the Revenue Budget Proposal

The proposed revenue budget for 2013/2014 represents:

- A net £2.5m or 2.0% decrease in the non-schools budget (after allowing for the technical adjustments required as a result of changes in the local government finance funding reforms).
- The Dedicated Schools Grant (DSG) increase compared to 2012/13 is complicated by the conversion of several schools to academies, and the transfer of additional responsibilities into the DSG from other funding routes. The overall increase in the DSG is estimated for 2013/2014 at £2.5m with total funding of £116.6m (including academies). The majority of the additional funding relates to the transfer of funding for 2 year olds and the funding for post 16 special needs provision, that have transferred from the Early Intervention Grant and the Education Funding Agency respectively. Taking account of this, the overall total represents a cash freeze per pupil compared to the previous year.
- Compared to 2012/2013. However as schools convert to academies the Department for Education (DFE) recoup the DSG payable to the Local Authority in order to make payments direct to the academies. The DFE estimate the recoupment will be £41.1m in 2013/2014 leaving £75.5m payable to the Local Authority.
- A freeze in the Council's level of Council Tax, which excludes Police, Fire and Parish precepts.

We are recommending a net revenue budget for 2013/2014 of £123.359m. Table 4 below, and Annex 1 to this Appendix, show the build-up of the recommended 2013/2014 revenue budget, compared to the rolled forward base budget from the current year.

Description	£'000
Total Base Budget rolled forward – 2013/14 (after removal of one-off items in 2012/13 Budget)	120,865
Technical Adjustments relating to Local Government Finance Funding Reforms	4,969
Contractual and Unavoidable Inflation	3,206
New Legislation / Government Initiatives	1,346
Increased Service Volumes	1,839
Impacts of Economic Downturn & Increased Competition	1,341
Other / Technical	1,423
Total including Growth	134,989
Change Programme & Efficiency Savings	5,634
Increases in Income from fees, charges and other grants	3,887
Service Reduction	2,109
Total Savings	11,630
Recommended Net Revenue Budget 2013/14	123,359

 Table 4: High Level Build-up of the 2013/14 Budget (detail in Annex 1)

In recommending the overall revenue budget to the Council, this also includes the individual service cash limits for 2013/14. These are shown in **Annex 1** to this Appendix.

Section 2 - Future Years

The Medium Term Service and Resource Plans were constructed to cover the 3 years 2013/14 - 2015/16 in line with the specific budget priorities and the council policy context (Annex 7).

The MTSRP's provide clear actions to address the majority of the estimated £30M funding shortfall faced by the Council over this period, including a balanced Budget for 2013/2014

The corporate financial assumptions and initial resource allocation as set out in Appendix 3 covered each of the next three financial years. Appendix 3 also explains that we cannot be certain at this stage about local government funding beyond 2014/2015 although we expect the financial challenge facing the public sector to continue until at least 2018 in line with the Government's Autumn Budget Statement in December 2012.

Table 5 below summarises the resource allocation from this Budget proposal together with the position for each of the following two years covered by the MTSRP's.

Whilst the MTSRP's provide for financial balance in 2013/2014, further work will be required in relation to 2014/15 and 2015/16. It is also the case, at this stage, that there is insufficient information available to fully identify future funding pressures including new government requirements, future demand changes and emerging issues. It is therefore expected that the requirements for savings in future years may increase.

Given the scale of savings already achieved and those outlined in the MTSRP's, it is likely that future savings will require prioritised changes to Council services. A rigorous process will therefore continue to be applied to the development of the Council budget and medium term financial planning process going forwards to enable resources to be prioritised between service areas.

The West of England City Deal

The City Region Deal is an agreement between Government, the West of England authorities and the West of England Local Enterprise Partnership giving increased local financial flexibility and freedoms in exchange for a focussed programme of investment to enable the region to achieve its full economic growth potential

Extensive preparatory work is taking place to enable options to be put before councils. At this stage each of the 4 West of England Unitary Councils and Government have agreed in principle only.

Key decisions are not required until June and July 2013. At this stage the background information set out in Annex 8 is just for noting as it provides important context for the medium term plans especially intended investment in regeneration.

Table 5: Resource Allocation 2013/14 to 2015/16

SERVICE AREA		20	13/2014		2014/15	2015/16
	GROWTH (£M)	SAVINGS (£M)	TECHNICAL ADJS (£M)	CASH LIMIT (£M)	INDICATIVE CASH LIMIT (£M)	INDICATIVE CASH LIMIT (£M)
Adult Social Care & Housing	3.148	1.836	3.537	58.377	59.388	60.083
Children's Services	0.663	0.895	3.506	24.455	23.688	22.025
Place	3.654	2.978	0.126	28.037	27.319	27.247
Regeneration, Skills & Major Projects	0.024	0.212	0.000	1.305	1.324	1.338
Resources & Support Services	1.167	3.080	0.000	3.279	0.026	(1.500)
Corporate & Agency	0.500	2.629	(2.200)	7.907	7.982	7.882
Totals	9.156	11.630	4.969	123.359	119.726	117.074
(Savings to be identified) / Headroom					(1.056)	(2.954)
Medium Term Financial Planning Total					118.670	114.120

Note: Some of the figures in this table are affected by rounding

The Cash limits for 2014/15 and 2015/16 are at this stage indicative and will be reviewed as part of the Medium Term Service & Resource Planning process for 2014/15 onwards.

The Medium Term Service and Resource Plans reflect the impacts and delivery of the Council's change programme. Further details of the change programme are set out below.

Council Change Programme

The Council's Change Programme was re-launched in 2010 and will have delivered substantial results over the period to 31 March 13. It takes account of the Government's austerity measures, the Academies Act, new requirements on the NHS, which affect our Health and Wellbeing Partnership and the various measures to promote Economic Growth.

The programme includes:

- Community Engagement where there is a clear understanding of the needs of local communities and how these may best be met either by the Council, one of its partners or increasingly from within the community itself. This also looks at how best to join up the engagement between communities and the public sector, not just the Council.
- Reshape Customer Services to be the public face of the Council, where 80% of questions are dealt with at the first point of contact, multi-agency one stop shops for face to face contact, plus improved web and selfservice. This will also take account of changes to the welfare and benefits system to Universal Credit and include a pilot on customer contact with the DWP with a significant emphasis on face to face contact.
- Bring together Support Services to deliver more effective and efficient services plus implement a new IT strategy with fewer systems, better support for flexible working, elected members and customer access.
- Meet the Government's requirements for Children's Services, Academies, Health and Social Care.
- Make further annual efficiency savings of at least £8m on top of other savings from the re-prioritisation of services.
- Make good use of Council assets rationalisation of office space, excellent sustainability standards in offices, encouragement of regeneration and transferring assets to community groups where appropriate.

There has been substantial progress over the last 2 years with more to come over the next 2 years. The recent activity and achievement includes:

- Creation of a new social enterprise (Sirona) to run community health and social care Council and PCT staff transferred in from October 2011 and the new organisation is operating successfully plus has a strong savings and improvement programme
- Customer Services improvements in many services such as highways, housing benefits, council tax, aspects of refuse collection, Bath one stop shop (an interim measure) and registrars. The Councils Website has also been updated and improved.

- Support services savings from simplification and standardisation of processes whilst also absorbing the effect of the creation of Sirona (which has mainly its own support services and some on-going support from the Council notably for property and IT – the property assets and networks having been retained by the Council)
- Changes in health and social care are being supported with the creation of a Health and Wellbeing board, which will deliver a Joint Health and Wellbeing Strategy for the area, plus arrangements to transfer public health to the Council.
- Academies have been supported to help their successful creation and a review of the LEA role is underway.
- Savings worth over £5M annually have already been directly achieved within the programme and the other efficiencies (to at least £8M in total) will be enabled by the programme.
- Council offices at Trimbridge House and Plymouth House in Bath have been closed (with the buildings returned to their owner) saving over £1M, flexible working has been rolled out.
- The Keynsham Regeneration project has been launched for completion in 2014 and work with the appointed delivery partner is now progressing.
- Records that took up valuable office space are now controlled through an efficient offsite archive and retrieval system.
- Lewis House in Bath has been modernised, as has the Hollies in Midsomer Norton. A new multi-agency one stop shop in Bath opened in May 2012.

Section 3 – The Capital Budget for 2013/14

Introduction

The Cabinet's proposals for the Council's capital programme are formulated in the context of:

- An ambitious yet prudent capital programme over a five year period.
- A desire to reduce the planned levels of external borrowing recognising the overall need to deliver value for money from the Council's external funding and treasury management decisions.
- The inclusion of significant government capital grant funding streams in relation to transport and schools which support the Council's strategic priorities and objectives.
- The prospective development and regeneration of various city centre and Employment Area sites across the Council area.
- The future financial challenge and the potential impact on future capital grant awards
- The increasing pressures on the revenue budget

This Capital Budget proposal:

- Limits new commitments to items which are in line with Council priorities and objectives and which are funded either from external sources, from anticipated future capital receipts, or where the potential costs of borrowing can clearly be met from within the proposed Budget.
- Recognises that careful consideration has been made by Officers and Members regarding future commitments and the direction of this programme.

Given that pressures on the revenue budget will increase over time, the Council maintains a prudent approach to revenue resource planning through the MTSRP's which take into account the associated revenue costs of the capital programme.

The MTSRP's recognise the intention to minimise new borrowing in the current market climate and fund the capital programme from a mixture of future capital receipts and internal cash flow. However the decision on the timing of new borrowing will be driven by market factors, particularly movements in interest rates to provide overall value for money to the Council.

The projected capital receipts for the period 2013/14 to 2017/18 are shaped by the Property Review and current proposals for development of Council owned sites.

All existing and future projected capital receipts will be utilised to support the general financing of the Council's Approved Capital Programme. The strategy of ear-marking capital receipts for specific areas of capital expenditure will be discontinued recognising the overall strategy to minimise and reduce the level of external borrowing going forwards.

The presentation of the Capital Programme retains the clear separation of schemes for **Full Approval** and those which are for **Provisional Approval**.

Items gaining **Full Approval** are clear to proceed to full scheme implementation and delivery, subject to appropriate project management and governance.

Items for **Provisional Approval** will require further Officer and Member scrutiny, including a formal Cabinet decision for Full Approval. The budget numbers for schemes shown for Provisional Approval are therefore included on an indicative basis, and as an aid to planning.

B) Recommended Programme for 2013/14

On this basis the Cabinet is recommending the Capital Programme as attached in **Annex 3** and summarised in the table below.

The proposed programme assumes total capital payments and funding in 2013/14, comprising both the programme for Full Approval of £59.036m and a programme for Provisional Approval (subject to) of £20.805m, as shown in Table 6 below. Table 6 also shows the indicative capital programme and funding at summary level for 2014/15 to 2017/18. **Annex 3** shows the total capital programme for 2013/14 to 2017/18 in more detail.

Table 6: Summary Capital Programme and Financing 2013/14 - 2017/18

For Approval

Capital Scheme	Budget 2013/14 £'000	Budget 2014/15 £'000	Budget 2015/16 £'000	Budget 2016/17 £'000	Budget 2017/18 £'000	Total £'000
Place	14,830	6,754	1,449	0	0	23,033
Regeneration, Skills & Major Projects	6,456	3,264	1,154	450	0	11,324
Children's Services	5,714	1,175	20	0	0	6,909
Adult Social Care & Housing	1,565	0	0	0	0	1,565
Resources & Support Services	28,971	8,779	8	159	134	38,051
Sub Total	57,536	19,972	2,631	609	134	80,882
Contingency	1,500	0	0	0	0	1,500
Total	59,036	19,972	2,631	609	134	82,382

For Provisional Approval (Subject to)

Capital Scheme	Budget 2013/14 £'000	Budget 2014/15 £'000	Budget 2015/16 £'000	Budget 2016/17 £'000	Budget 2017/18 £'000	Total £'000
Place	8,655	7,342	6,217	6,487	1,518	30,219
Regeneration, Skills & Major Projects	5,389	7,750	0	0	0	13,139
Children's Services	4,636	395	0	0	0	5,031
Adult Social Care & Housing	1,325	2,325	1,000	1,000	1,000	6,650
Resources & Support Services	800	6,507	2,607	2,307	1,757	13,978
Total	20,805	24,319	9,824	9,794	4,275	69,017

Grand Total	79,841	44,291	12,455	10,403	4,409	151,399

Funded By

Financing	Budget 2013/14 £'000	Budget 2014/15 £'000	Budget 2015/16 £'000	Budget 2016/17 £'000	Budget 2017/18 £'000	Total £'000
Grant	21,965	16,689	5,879	5,445	422	50,400
Capital Receipts/RTB	9,732	23,020	6,253	16,050	0	55,055
Revenue	823	2,401	1,297	563	478	5,562
Borrowing	44,943	766	-1,309	-12,070	2,420	34,750
3 rd Party (inc S106)	2,378	1,415	335	415	1,089	5,632
Total	79,841	44,291	12,455	10,403	4,409	151,399

<u>Funding</u>

The revenue budget for 2013/14 and the Medium Term Service and Resource Plans for 2014/15 and 2015/16 provide for the revenue consequences of the Council borrowing in support of capital expenditure where appropriate.

A desire to reduce the planned levels of external borrowing has been recognised within the funding arrangements for the Capital Programme. It is anticipated that all current and future capital receipts will be utilised, together with the Council's internal cash balances to fund the Capital Programme. However the decision on the timing of new borrowing will be driven by market factors, particularly movements in interest rates to provide overall value for money to the Council.

The Capital Programme assumes the following achievement of capital receipts:

- 2012/13 = £19.9M
- 2013/14 = £ 2.6M
- 2014/15 = £22.8M
- 2015/16 = £ 6.2M
- 2016/17 = £16.0M
- 2017/18 = £ Nil

This prudent assumption recognises the difficulty in accurately projecting the actual level of capital receipts over longer-term periods.

The £3m set aside in an earmarked revenue reserve for affordable housing and capital development in 2010/11 remains fully committed to support the provision of affordable homes in the Bath Western Riverside development.

Significant Elements of the Capital Programme

Bath Transportation Package

Fully Approved - the scheme will expand all three of the council's existing P&R sites, improve variable message signs (including car park count systems being installed into the main city centre and car park and ride sites), environmental improvements in the city centre and improve the 9 major bus routes into the city with new bus stops, bus priority and real time information.

The scheme received Full Approval from Department of Transport on 11th July 2012 and delivery commenced in August 2012. The work at Odd Down Park and Ride completed prior to Christmas and Lansdown Park and Ride will open by end of March 2013. Ducting work for VMS has been installed and Bus stop improvements, signal upgrades and car park counts have commenced.

Total budgeted costs are \pounds 31.87M (including \pounds 1.530m for High Street improvements), funded by a maximum grant of \pounds 10.958m from DfT with the remainder funded by the Council and other local contributions.

Highways Structural Maintenance

The Highways Structural Maintenance budget is included for **Full Approval** at £4.330m funded wholly from direct government grants including the recently announced additional Highway Maintenance Funding of £663K to recognise deterioration due to adverse weather.

This will be used to address a programme of priority works across all highway asset groups namely carriageways, footways, structures (bridges, retaining walls, embankments & culverts), drainage and electrical infrastructure as identified through on-going inspection, monitoring and evaluation.

Local Transport Improvement Schemes

The Local Transport Improvement Schemes budget is included at £1.264m funded from government grant, section 106 contributions and other 3rd party contributions. It is included for **Provisional Approval** and is the subject of a separate report on the Agenda for the Council meeting on February 19th 2013, providing further details of the individual schemes for full approval.

The 2013/14 schemes will initially be subject to agreement with the Cabinet Member for Transport and will focus on supporting the Joint Local Transport Plan approved with our WoE partner authorities.

The focus is around 5 key transport goals of Reducing Carbon Emissions, supporting Economic Growth, Promoting Accessibility, contributing to better safety, security and health and finally improve quality of life & a healthy natural environment.

Victoria Bridge

Fully Approved - capital funding of £2,587K is provided to design and install a permanent solution for Victoria Bridge. The approved solution is a refurbishment of the bridge involving its complete removal and reconstruction, with existing, retained members being refurbished, and new steel members added where required.

Ramps were installed in May 2012 and the route across the Bridge opened to the public in the same month, on a temporary basis. The route will be closed again when works on the permanent solution commence in 2013.

External engineers have been appointed to carry out the design, the work for which is on-going and is expected to be complete in Q2 2013. The work required to refurbish the Bridge is programmed to commence in Q3 2013 with completion currently envisaged in early 2014.

Haycombe Cemetery Florist / Cafe

There is an aspiration to broaden the range of facilities available to our customers at Haycombe Cemetery, some of whom travel long distances to pay their respects to loved ones. An ambition is to provide a cafe for customers and also a florist concession so that customers can purchase floral tributes on site. It is anticipated that we would provide this over the next 3 years.

Included for **Provisional Approval**, there is a need to develop a business plan and a cost plan for this project and review them in light of other priorities for the business which include the provision of a new canopy for the Crematorium entrance and courtyard. Other key dependencies include public consultation, identification of suitable site locations for the concessions, consideration of who should operate the concessions i.e. the Council or private operator and securing necessary planning permissions.

An indicative budget of £100k is included for provisional approval in 2013/14 subject to development of a detailed business case.

Cycle Routes

For **Provisional Approval** - £500K to provide road cycle lanes, contra flows in one way streets and some off road cycle paths. The aim is to make cycling safer for adults and children so that more will take up cycling. This programme will complement the Local Sustainable Transport programme to increase commuter cycling, the new off road track at Odd Down, the Two Tunnels project opening in April 2013 and the Batheaston Bridge.

River Corridor Fund

For **Provisional Approval** - £100K. This programme will consider schemes to improve river and river bank safety, improving walkways and facilities for those using boats. The Council will consider advice from the River Corridor group before deciding on the schemes to be funded.

Sydney Gardens

This scheme for the restoration of Sydney Gardens, involves an application to the Heritage Lottery Fund under their Parks for People Programme. The scheme is split into 3 phases:

Round 1 Feasibility

Round 2 Development

Round 3 Implementation

A project budget of £500k is included for **Provisional Approval** subject to the development of a detailed project plan and funding confirmation from the Heritage Lottery Fund

Odd Down Playing Field

Included for **Full Approval**, Odd Down Playing Field is funded by a £1.232m, S106 contribution from the Sainsbury's Supermarket development.

The playing fields are also to benefit from a significant regeneration project which will deliver a 1.5km cycling track (£600k funded by British Cycling) and the implementation of a scheme to deliver an artificial (3G) playing pitch and a new clubhouse and changing room facility.

The 2013/14 Capital Programme also includes a provisional Council funded budget of £200k, which it is proposed to be used for the provision of enhanced changing facilities on the site.

The project will be delivered in 3 phases; 1.5km cycling track by April 2013, the artificial pitch by January 2014 and the new clubhouse and facilities by September 2014.

Roman Baths Development – Phase 2

For **Provisional Approval**, the second phase of the Roman Baths Development will include further on-going investment to maintain, refresh and update the 'visitor offer'; this includes new step-free access throughout the museum and improvements to the East Bath.

Works will be spread over a number of years to reflect phasing and timing to fit around seasonal demand for the attractions. The project will also work closely with the Council's financial planning team to avoid potential impacts on the Council VAT position.

Bath Western Riverside

Fully Approved, the BWR development continues to progress. The current position is as follows:-

- The first 30 private residential units are now occupied or sold.
- 100 affordable homes are now occupied.
- Phase 2 occupation is due during 2013.
- The 2nd Phase of infrastructure procurement was commenced in November 2012.

Radstock Regeneration

Fully Approved - the scheme has undergone a re-designed to take on board feedback from the community. A new planning application is now being prepared. The revised project budget is now £1,575K of which £800K is confirmed funding via an HCA grant..

Public Realm

For Provisional Approval - £339K is included in the 13/14 capital programme for the improvement of streets and spaces within Bath.

Bath Quays South

£650K is included for **Provisional Approval** for the design development up to planning submission for a new commercial premises on Bath Quays South to form part of the Corporate Estate. The design development work will only proceed once an agreement to lease the facility has been reached with a tenant. The formal approval for the full capital expenditure will be subject to a full business case.

Growth Enabling Infrastructure - Bath Quays Footbridge, Bath Flood Risk Management, BWR Relocation of Gas Holders, BWR Replacement of Destructor Bridge.

For Provisional Approval –

Flood Risk Management

Strategic Flood Mitigation Phase 1: to enable the development of key river corridor sites in the Bath City Riverside Enterprise Area

BWR Relocation of Gas Holders

Decommissioning and decontamination of the Windsor Gas Station: to enable removal of the HSE restriction on development at Bath Western Riverside and other sites in the Windsor Bridge area

BWR Replacement of Destructor Bridge

The construction of a new road and pedestrian bridge to replace the Destructor Bridge at BWR: to provide access to the BWR western site

Bath Quays Footbridge

The construction of a new pedestrian bridge over the River Avon linking Bath Quays South and Bath Quays North: to enable the former site to be brought forward for development

Schools Capital Investment

Schools and Early Years capital grant funding for 2013/14 has not been confirmed by the Education Funding Agency (EFA). Indications are that allocations will remain at a similar level to 2012/13: £1.564m for Basic Need to support provision of additional pupil places where there is population growth: £2.403m for Capital Maintenance and an estimated £0.402m of Devolved Capital to schools.

Although indications from the DfE are that allocations will remain at a similar level it is possible that there will be a reduction in Capital Maintenance given that only one non church secondary school, Chew Valley, remains in the 'control' of the local authority. All others have opted to become academies thus removing the responsibility for maintenance at these schools from the Council.

Basic Need priorities will include provision of additional capacity at St Saviour's Junior, Oldfield Park Junior and Paulton Junior schools to reflect earlier expansions at St Saviour's Infants, Oldfield Park Infants and Paulton Infants which have previously been approved and are already in the approved capital programme. Feasibility studies will be undertaken to identify options and costs for providing additional accommodation at these junior schools.

Priority for Capital Maintenance funding will be to address the most pressing condition items in schools e.g. windows, roofs, boilers etc. through the Schools Planned Maintenance Programme (SCPM). However due a cautious approach to allocations in 2011/12 and 2012/13 it has been possible to fund the 2013/14 SCPM from within existing funding and £1m for this is already included in the capital programme for **Full Approval.** A detailed list of proposed priorities for this funding is attached at Annex 3(i).

Until the costs of the proposed basic need schemes above are identified it is not possible to identify whether DfE Basic Need funding for 2013/14 will be sufficient or whether it will be necessary to also contribute capital maintenance funding to these projects ensuring that the Council fulfils its statutory duty to provide sufficient school places.

Subject to confirmation of funding Officers will provide details of proposed projects and costs for initial consideration by the Cabinet Member for Early Years, Children and Youth.

In addition to planned future projects there are also several larger scale projects previously approved as part of the capital programme in various stages of development in primary and secondary schools, these include;

- Ralph Allen Applied Learning Centre and artificial turf pitch focussing on science with a learning environment to provide students with access to high technology equipment and practices mirroring those available in professional laboratories and industrial settings - Construction underway with completion Aug 2013 - £2.4m
- St Gregory's /St Mark's Post 16 Block new joint 6th form block on site adjacent to St Gregory's which will admit pupils from both St Gregory's and St Mark's. Construction underway with completion September 2013 -£2.375m
- Weston All Saints Primary –Basic Need to provide additional classrooms in two phases enlarging the school to 630 places overall. Phase 1 to be completed for 2014 of providing one classroom as an extension to the recently completed junior block with the remaining 6 classrooms to be delivered as a separate block by September 2015. Design phase underway £1.8m.
- Castle Primary Basic Need phase one to provide additional pupil places on a phased basis to provide a capacity of 420 from the current 210 places. This project will provide an additional 90 places by refurbishment of IT suite for September 2013 followed by construction of two classrooms for September 2014 - £800k
- Paulton Infants Basic Need -to provide 90 pupil places to meet rising birth rate and housing development to be funded via BN(£590k) and S106(£260k). Project phasing will see the refurbishment of kitchen space for September 2013 and construction of two classrooms for September 2014 - £850k

Schools Energy – Invest to Save Fund

For **Provisional Approval** - this project will provide financing for energy efficiency projects identified by schools. The funding is provided partly £290K by the schools forum from Dedicated Schools Grant funding. The remainder is funded by service supported borrowing. The sustainability team are working with schools to provide them with information to guide them into developing projects to be funded partly by the resource from the DSG and partly by service supported borrowing. The split between the two sources of funding will depend on the details of the projects.

The projects will be mainly changes to fabric of the school buildings in order to save energy costs and cannot start until the surveys in schools are completed as the sustainability team will need a full set of information in order to target the allocations.

The timing of the projects will be influenced by the desire of schools to complete work during holiday periods and by any linkages with other capital projects being completed by property services.

Disabled Facilities Grant

For **Full Approval** of £1M in 2013/2014 - this is the annual mandatory grant service administered by Housing Services for eligible applicants that satisfy the criteria of firstly, a necessary and appropriate home adaptation to enable them access to and use of the their home, secondly, the required adaption being reasonable and practical, and thirdly, meet the requirements of a test of resources. The assessment of need is carried out by the Council's OT service.

The scheme allows an eligible applicant to continue to live independently in their own home by providing a stair lift, suitable washing facilities or other relevant home adaptations.

Affordable Housing

Investment of £550k per annum in 2013/14 & 2014/15 is included for **Provisional Approval** subject to the provision of a detailed project plan and business case.

This provision is for supporting work on tackling empty homes, increasing the delivery of affordable housing and associated costs. Such provision will contribute towards attracting inward investment through the New Homes Bonus and help achieve aims within the Council's Core Strategy and the Housing and Well-being Strategy. Political and Corporate approval will be sought for each spending proposal to ensure value for money and purpose of outcome.

Gypsy and Travellers Site

A capital budget provision has been made for up to $\pounds 1.8m$ across 2012/13 - 2014/15 to provide a 14 pitch transit site for Gypsy & Travellers. This scheme is included in the capital programme for **Provisional Approval** subject to detailed project plans being submitted once appropriate sites are located through the planning process.

The Gypsy & Traveller Accommodation Assessment 2007, which covers the period up to 2016, identified a need for 22 residential pitches & 20 transit pitches. This is currently being updated. The provision of a 14 pitch site will make a significant impact in meeting need.

Workplaces Programme including Keynsham Regeneration

Fully Approved - the workplaces programme incorporates the entirety of the Council's office accommodation and rationalisation, better customer access, flexible working, carbon savings, supporting joining up public services including shared front office with key partners, substantial efficiencies (both cashable and non-cashable) and acts as a catalyst for regeneration.

The scheme includes:

- The Keynsham regeneration scheme to include Keynsham offices, retail, one stop shop, and library received planning consent last autumn. Demolition works of the former Keynsham Town hall, Library and retail shops is well underway. The new build is on programme and budget to open in the autumn of 2014.
- The main refurbishment of the corporate offices in Lewis House, Bath and the Hollies Midsomer Norton is now completed, with minor related projects nearing completion.

The workplaces programme is set to achieve a return of over 10% (after capital financing costs) and involves a total capital cost of £32M in future years (2013/14onwards) of which £28m relates to Keynsham

Victoria Hall

Fully Approved – subject to a cost & project plan, the objective is to bring Victoria Hall back to a safe and usable condition so that it can become a community space.

Until Feb 2012 the Hall was used as a community facility and an administrative base for Radstock Town Council. The building is in disrepair and officers were asked to carry out a feasibility study for a developed mixed use community, meeting and exhibition space with scope for locating the library into the building.

The Cabinet have approved continuing with this scheme.

The next steps will be to complete a business plan for the asset which includes the relocation of the town library. Sources of external funding such as grants will be further explored. A view will need to be taken to determine how best to maximise the financial return from the caretaker's house to facilitate the development. It is currently estimated that works will be completed by the end of the 2013/14 financial year.

Specific Developments for Full Approval – Saw Close and MSN South Road Car Park

Saw Close

The Saw Close site has been identified as a development possibility since the early 1990s. Property Services are currently in discussions with a local developer and adjoining landowner (the prospective developer) with the aim of achieving the best value deal possible for the Council in terms of capital generation and if possible maintaining revenue. This should also result in bringing a much neglected space back into full use and improving the public realm.

Negotiations with the prospective developer have progressed to the point that it is anticipated that the Council can reasonably expect to receive a capital receipt and appropriate costs associated with the disposal can be charged to the Capital Programme.

Midsomer Norton South Road Car Park

The overall objective is to bring forward this key Council freehold site for redevelopment in line with the draft Core Strategy and ERDP for Midsomer Norton.

In national planning terms South Road car park as an in-town location is sequentially the preferred site for food retail in the Somer Valley, supporting the role of Midsomer Norton as the market town for the area serving 46,000 people.

The location would provide increased footfall, together with linked shopping trips to the existing High Street. This will enhance the retail offering and contribute to the town's vitality and future viability and hopefully, stimulate further investment and regeneration.

Specific Developments for Provisional Approval – Grand Parade & Undercroft and Riverside Keynsham

Grand Parade & Undercroft

The key objectives for this development will include:

- Creation of a new fully accessible destination point for tourism and business within the City
- Bringing redundant space back into use for community and commercial benefit
- Enhancing the Bath Markets, as well as the museum and art gallery
- Increasing revenue income

The timeline will compliment the Workplaces Programme (due for completion in the autumn of 2014) with the assessment of development opportunities to include the Grand Parade and Undercroft, including Bath Markets, Victoria Art Gallery and the Colonnades.

Other potential linked development opportunities might include restoration of Pulteney Bridge, developing the Boat Dock, improving the Environment Agency (EA) radial gate and re-developing Bog Island.

The assessment will deliver a development specification that is based on the Council's priorities and objectives. This specification will then be used as an Invitation to Tender for suitably experienced developers, and a recommendation will be made to the Cabinet on appointing a developer(s).

Riverside Keynsham

This project is part of the overall Keynsham Regeneration scheme. Options and negotiations for the redevelopment of the Riverside building are continuing.

Ideas are being explored for a range of uses for the site and particularly housing, including town houses and retirement living for older people. There will be consultation with the community in the autumn of 2013 before any decision on the future of the site.

For this first phase, there are three options for the development scheme which are in part dependent upon the negotiations and will of the intermediary landlord of Riverside. The fees and costs involved vary depending on the option pursued. A more detailed business case will be brought back through the capital approval process following the consultation period.

Desktop As a Service – VDI Technology

For **Provisional Approval** – provides a virtualised desktop that meets the needs of WorkPlaces which will be deployed enterprise wide. Subsequently the virtual desktop will be available on almost any device with an internet connection, including personal devices.

Timescale – pilot 2012; gatekeeping sign off to deploy March 2013; completes Q3 2014 with indicative cost of £1.88M

Likely key deliverables include:

- thin client desktop in the office,
- virtual desktop at home, in the field/mobile,
- desktop available on almost any device including your own,
- reduced support costs
- lower carbon

Customer Services System

Fully Approved - this is a critical corporate system underpinning the whole Customer Service Work stream. It is the key enabler for new ways of working and significant benefits realisation across the organisation through standardisation and use of systems, greater automation reducing manual processes and increasing information flow as well as online capability.

Contracts were signed in 2012/2013 development is now underway. Supplier payment milestones have been agreed and have now been reflected in capital profiling.

ICT Strategy

The IT Strategy consists of the following project streams which are included for **Provisional Approval**:

<u>Pilots to Rollout Mobile Devices</u> - such as Smartphones, ruggedized devices and small form factor tablets to field workers to improve productivity and multidisciplinary workers. Approximately 300 field devices to be available to field based and mobile staff so that records can be accessed and updated when out and about enabling operational efficiencies in areas such as on-street presence, enforcement and on street reporting. Increases productivity for social workers, other case workers e.g. benefits workers - more visits per week, less dead travel time.

<u>Open Source</u> - investigate with a view to moving from a traditional licenced desktop to using open source desktop software to reduce annual licensing costs. In whole or in part.

<u>Collaboration</u> - The ability to collaborate effectively internally and with other agencies on cases e.g. Connecting Families, collaboration with partners e.g. CCG, Police, Health, DWP, Curo/RSL's etc and internally. This will build on technology the Council already owns.

These projects are subject to a pilot / feasibility leading to the production of a business case to support wider implementation.

People & Communities – IT System Replacement

For **Provisional Approval** and indicative budget of £1m phased between 2014/15 to 2016/17.

People and Communities currently have 3 main IT systems in use and other smaller applications. Each system does slightly different things and holds different parts of a record. The information is used by different staff for different purposes from reporting on attendance to care plans. The intention is to look to rationalise these and other applications in P&C to deliver better data integration, ease of use (less systems to learn) and a reduction in the costs. The new solution will fit with desktop as a service, support and enhance mobile working, increase staff productivity, embrace media rich functionality and have fully integrated reporting. A strategic review is underway and a full business case will be produced with a supporting cost benefit analysis.

The timescale will align to the strategic review of ICT needs during 2012/13 leading to the development of a business case in 2013 with a target for deployment during 2014.

Biomass Energy Efficiency Fund

For **Provisional Approval** - £500K over two years to fund the installation of Biomass boilers on the basis that the capital cost can be repaid by those participating through their energy savings over a period of time.

Minimum Revenue Provision (MRP) & Depreciation Policy

The Council is required to make revenue provision to repay capital spend that is financed by borrowing (either supported or unsupported). This is called the Minimum Revenue Provision (MRP). The Department of Communities & Local Government has issued regulations that require full Council to approve a MRP & Depreciation Policy in advance each year. The Council is recommended to approve the statement in Annex 4 which is unchanged from that agreed in 2008/2009 and defines how the Council will make a prudent minimum revenue provision for all new unsupported borrowing from 1st April 2008.

Prudential Indicators

The Capital Prudential Indicators are shown in table 7 below.

Table 7: Capital Prudential Indicators.

•							
PRUDENTIAL INDICATOR	2011/12 Actual	2012/13 Probable Outturn	2013/14	2014/15	2015/16		
Current Financial Plan Data							
Estimat	te of Capit	al Expendit	ure (£'000s)			
Actual/estimates of capital expenditure	45,838	55,665	79,841	44,291	12,455		
Net Increase in council t	ax (band [) per annun	n) Figures i	n £'s (not £	E'000's)		
The estimate of incremental impact of the new capital investment decisions on the council tax			£0.29	£2.25	£0.33		
Cumulative totals:			£0.29	£2.54	£2.87		
	ancing as	% of Net R	evenue Stre	eam			
Actual/estimates of the ratio of financing costs to net revenue stream			8.66%	10.54%	10.66%		
Memo: estimates of the ratio of financing cost to gross revenue stream			3.24%	3.85%	3.79%		
	Borrowi	ng Limits (£	m)				
Operational boundary – borrowing			£167m	£156m	£159m		
Operational boundary – other long-term liabilities			£2m	£2m	£2m		
Operational boundary - total			£169m	£158m	£161m		
Authorised limit - borrowing			£201m	£196m	£193m		
Authorised limit – other long-term liabilities			£2m	£2m	£2m		
Authorised limit - total			£203m	£198m	£195m		
Capital Financi	ng Require	ement (£'00		1 March)			
Actual/estimate of capital financing requirement	136,134	161,050	201,211	195,588	193,269		

Gross Debt and the Capital Financing Requirement

In order to ensure that over the medium term debt will only be for a capital purpose, the Council should ensure that external debt does not, except in the short term, exceed the total of the capital financing requirement in the preceding year plus the estimates of any additional capital financing requirement for the current and next two financial years.

After reviewing the capital programme and borrowing proposals, the Section 151 officer reports that the Council will continue to meet the demands of this indicator.

Borrowing limits

The Authorised limits for external debt include current commitments and proposals in the budget report for capital expenditure, plus additional headroom over and above the operational limit for unusual cash movements.

The Operational boundary for external debt (or planned borrowing level) is based on the same estimates as the authorised limit but without the additional headroom for unusual cash movements. This level also factors in the proposed use of the Council's longer term cash flow balances to "internally" meet the financing requirements for the Keynsham Town Centre Regeneration project, subject to overall treasury management decisions.

Section 4 – Revenue & Capital Reserves and Contingencies

A) Significant Earmarked Reserves

In developing the proposed Budget for 2013/2014 all the Council's earmarked reserves have been reviewed and the related potential liabilities re-assessed to establish the future requirements for each of the reserves. The reserves were allocated to one of three categories:-

- RED Reserve is required to meet known future liabilities or commitments.
- AMBER Reserve is required to meet anticipated future liabilities or commitments.
- GREEN Reserve is no longer required the liability or commitment no longer exists or can be met in other ways.

This review has identified a number of reserves that need to be maintained in the Red or Amber categories and also a number of reserves falling into the Green category which, can now be considered for alternative uses. The proposals for significant earmarked reserves will therefore be as follows:-

<u>The Revenue Budget Contingency</u> – the reserve has been utilised during 2012/13 to meet a range of in-year pressures and priorities and is anticipated to stand at £491K for 2013/14. This reserve will continue to be made available to meet in-year revenue budget pressures.

<u>The Medium Term Financial Challenge Reserve</u> – the Council continues to develop and progress a significant change programme both to redesign and reshape a range of services to recognise the future shape of the Council. A number of the specific changes are also progressing to respond to the initiatives imposed by the Government including the responsibilities for public health, changes in health and social care commissioning and the impacts of academy schools.

All these changes will require on-going investment in terms of financial and human resources to develop implement and embed them within the organisation.

Subject to the overall limits of the reserve, it may also be utilised to provide oneoff resources in support of the delivery of the specific savings plans set out in the MTSRP's subject to an overall business case and project plan.

This reserve will meet the costs associated with these changes and is fully committed. Based on current estimates, the reserve will be fully utilised by the end of 2014/2015.

<u>The Restructuring Reserve</u> – the significant financial challenge facing the Council has and will continue to lead to job losses as savings and efficiencies are delivered. The MTSRP's indicate that around 300 posts will be lost as a result including through redundancy. The associated severance costs will be significant, and based upon experience to date, it is proposed that this reserve will be available to meet these costs in full, avoiding the need for subsequent additional service savings purely to pay such severance costs. This approach will be regularly reviewed to ensure it is being applied appropriately. The reserve is projected to be fully utilised by the end of 2015/2016.

<u>The Affordable Housing Reserve</u> – this reserve is committed to fund the Council's contribution to affordable housing including within the Bath Western Riverside development.

<u>Capital to Revenue Reversion Reserve</u> – this reserve was established to recognise the potential for capital costs in respect of Bath Transport Package reverting to revenue in the event the scheme did not gain Department of Transport funding. This funding has now been confirmed and the scheme is progressing. This reserve is therefore no longer required and will be fully available for reallocation to other priorities.

<u>Capital Financing Reserve</u> – this reserve represents funding accumulated to finance the revenue borrowing costs of capital projects, where actual borrowing is delayed based on market conditions and historic use of the Council's internal cash flow. The reserve would then be applied to meet future borrowing costs. The current financial climate and market expectations for prolonged very low interest rates, together with the strategy to fund capital projects from Capital Receipts and internal cash flow indicate this reserve will not be needed over the period of the proposed 5 year Capital Programme. It is therefore reasonable to make the current balance of the reserve available for reallocation to other priorities.

<u>Financial Planning Reserve</u> – this reserve has been established to support the future medium term financial planning of the Council. It will be allocated as part of the Budget each year to support the specific medium term financial proposals and priorities of the Council. Funding for this reserve has been provided from the review of all Council reserves set out above, and any surplus on the Council Tax Collection Fund.

Table 8 below, sets out the projected level of earmarked reserves taking account of anticipated commitments over the next 3 years to 2015/2016. This reflects the review of reserves including reallocations during 2013/2014 to support One-Off Headroom Allocations and the establishment of the Medium Term Financial Planning Reserve. All other Earmarked Reserves are anticipated to be fully committed with the allocation of available Revenue Budget Contingency subject to new and emerging priorities.

 Table 8: Projected Significant Earmarked Reserves

	Revenue Budget Contingency £000	Medium Term Financial Challenge Reserve £000	Restructuring Reserve £000	Affordable Housing Reserve £000	Capital to Revenue Reversion Reserve £000	Capital Financing Reserve £000	Financial Planning Reserve £000
Currently Available	491	2,651	5,598	3,000	2,755	2,799	0
Transfers between Reserves					-2,755	-2,799	6,569 *
Estimated Reserves @ 1st April 2013	491	2,651	5,598	3,000	0	0	6,569
Allocation in 2013/2014	0	-1,691	-1,742	-	-	-	-3,815
Balance C/F	491	960	3,856	3,000	-	-	2,754
Allocation in 2014/2015	0	-960	-1,372	-1,778	-	-	ТВС
Balance C/F	491	0	2,484	1,222	-	-	
Allocation in 2015/2016	0	-	-2,484	-819	-	-	твс
Balance C/F	491	Nil	Nil	403	-	-	Nil (est)

*Includes balances from other small discontinued reserves and projected surplus from the Council Tax Collection Fund.

B) Unearmarked Revenue Reserves -

The Local Government Act 2003 contains a duty on the statutory finance officer (s151 Officer) to report to the Council, at the time the budget is considered and the Council Tax set, on the robustness of the budget estimates and the adequacy of financial reserves. The report of the s151 Officer on this subject is included as **Annex 2** to this report and is recommended to the Council. The conditions of the report by the Divisional Director - Finance are an integral part of our budget recommendations. This sets an unearmarked reserves target of £10.5 million based on the thorough risk assessment and in the context of the "one-off" funding proposals contained within this Budget proposal.

Table 9 below details the proposed movement in the level of unearmarked reserves over the period of the medium term service and resource plans. This analysis includes the proposed use of reserves to support invest to save proposals included within the Budget, specifically:

• The Workplaces Project – requires total cumulative revenue investment of £2.8m over the period to 31 March 2015 with reserves fully repaid by 2021/2022 and on-going revenue savings thereafter of over 10%.

This actual level of unearmarked reserves will also depend on the Outturn position for 2012/2013 and on future decisions by the Cabinet about any overspends. For financial planning purposes the Outturn estimate for 2011/2012 is assumed to be a balanced position. The figures are therefore only an estimate at this stage and are without prejudice to future Cabinet decisions.

C) Adequacy of reserves

The s151 officer's report on the robustness of estimates and adequacy of reserves is set out at Annex 2. This provides a reserves strategy to maintain non-earmarked General Fund reserves at £10.5m based on a thorough risk assessment. The projected reserve levels are set out in Table 9 below.

	2013/14 £'000	2014/15 £'000	2015/16 £'000
Estimated Reserves @ 1st April each year	9,741	8,844	7,728
2011/2012 Projected Outturn Underspending	-	-	-
Projected Invest to Save Movements	-897	-1,116	+134
Estimated Reserves @ 31st March each year	8,844	7,728	7,862

Table 9: Projected Non-Earmarked Revenue Reserves

Based on anticipated invest to save commitments mainly associated with the Workplaces project, the Non-Earmarked Reserves will be repaid in full by 2021/2022 and at no point will reduce below the risk assessed minimum level of $\pounds 6M$.

Under the Council's Invest to Save Scheme, the Section 151 Officer in consultation with the Cabinet Member for Community Resources, can authorise funding from within the non-earmarked reserve for robust and credible invest to save proposals (i.e. in the short term creating a 'negative ear-marked reserve' which is then repaid over time, usually 3 years, from the related savings). This is subject to the overall level of non-earmarked reserves being maintained above the risk assessed minimum level of £6M.

D) Capital Risk Contingency

There are three levels of risk provision in relation to the capital programme.

Firstly individual major projects within the capital programme hold their own contingency;

Secondly, the capital programme includes a funded corporate risk contingency currently forecast to be £2.144m. This includes a £1.5m top up in 2013/14. The level of this contingency will be reviewed regularly.

Thirdly the corporate risk assessment on which the general reserves target is based includes an element in the context of the capital programme based on the risks of the current programme.

As with all capital projects, relevant risks are being considered as part of the overall risk-assessed general reserves and the Corporate Risk Register.

E) Governance

The Council is requested to approve that the specific arrangements for the governance and release of Council reserves, including invest to save proposals, be delegated to the Council's Section 151 Officer in consultation with the Cabinet Member for Community Resources and the Chief Executive.

Section 5 – Council Tax

This section shows the implications of the recommended revenue budget for Council Tax levels for 2013/2014.

The financial settlement included an announcement of financial support for councils who freeze their Council Tax for next year at the current level (i.e. a zero increase). The proposed Budget provides for a zero increase in Council Tax for 2013/2014. **Table 10** explains the calculation of this figure:

Table 10: Council	Tax 2013/14	for Bath &	North	East	Somerset	Council
Services						

Description	Amount	Comments
Recommended Net Revenue Budget	£123,359k	See Annex 1
Less grant, retained business rates, reserves and estimate of Collection Fund surplus	£52,017k	See Annex 1 Sources of Funding
To be funded by Council Tax	£71,342k	
Tax base (Band D properties equivalent)	59,360.17	Approved by the Section 151 Officer in December 2012
Recommended Council Tax at Band D for 2013/14	£1,201.85	
2012/13 Council Tax Band D	£1,201.85	
Recommended Increase	£0.00	0% increase

The figures above exclude parish, fire and police precepts.

This Council collects Council Tax on the behalf of the parishes, Fire and Police Authorities and the final bills issued will include the Council Tax they have requested this Council to collect. These will form part of the Council's overall budget-setting resolution.

The Police & Crime Commissioner provided an updated draft of the Medium Term Financial Plan to the Avon & Somerset Police and Crime Panel on 16th January 2013. This update included the intention to freeze Council Tax for 2013/14. Final formal approval will be sought on 6th February 2013.

The Avon Fire Authority at its meeting on 14th December 2012 agreed to consult on the following three budget options:

- Option 1 a Council Tax Freeze
- Option 2 1.99% Council Tax Increase
- Option 3 8.92% Council Tax Increase (Council Tax equating to the average for combined Fire Authorities in 2012/13)

The Fire Authority will meet on 8th February 2013 to finalise its budget and set its Council Tax and precepts for 2013/2014.

The headline increase will be affected by the final decisions of the parishes, Fire and Police Authorities, and any decision made concerning special expenses (see below). Final figures will not be available until after Fire and Police meetings and decision dates highlighted above.

Table 11 sets out the composite Council Tax likely to be charged:

Council Tax charges (Band D) made by	Charge made now 2012/13 £	Proposed Charge 2013/14 £	% Change
Bath and North East Somerset Council	1,201.85	1,201.85	0% (£0.00 at Band D)
Avon and Somerset Police	168.03	ТВС	Final Decision to be taken on 6 th February 2013.
Avon Fire & Rescue	62.77	ТВС	Final decision to be taken on 8 th February 2013
Total excluding parishes	1,432.65	твс	
Parishes (average)	33.62	ТВС	Not known at time of writing
Total	1,466.27	ТВС	The 2013/14 figure will depend on decisions taken by the Police, Fire and Parish/Town Councils

Table 11: Potential Total Council Tax 2013/14 (Band D)

The precepts required by Parishes, Fire and Police will form part of the Council Tax setting resolution at Council on 19th February 2013, and so the necessary updated information will be set out in the report.

Special Expenses

As part of the 2012/2013 Budget preparation process no special expenses were declared (with the exception of Parish and Town Council precepts). It is proposed that this policy remains unchanged for the 2013/2014 budget.

Portfolio Cash Limits 2013/14 - Revenue Budgets

APPENDIX 2 ANNEX 1

MTSRP Technical Total 2013/14 Savings Adjustments Budget Changes	11 (60) 28 (100) 29 (110) 29 (110) 236 (73) 36 (73) 36 (377) 38 (377) 38 (377) (350) (550) (250) (250)	(500) (500) (1,400) (1,400) (1,400) (1,400) (300) (870) (170) (1,330) (759) (759)	1,237 (5,049) (2,200) (6,012) 6,000 3,134 (1,727) 3,537 4,944 55,694 (50) (50) (50) 53 53 3,134 (1,727) 3,537 4,944 55,694 3,134 (1,777) 3,537 4,894 56,243
MTS			12,013 1,2 50,750 3,1 598 3,1 51,349 3,1
201	(1,507) (1,507) (1,507) (1,507) (1,907) (1,907) (1,706) (1,907) (1,706) (1,706) (1,706) (1,706) (1,907) (1,907) (1,907) (1,907) (1,907) (1,907) (1,294		
Aemo offs (ir off vi			7
Current 2012/13 Cash Limits	1,530 1,530 738 738 1,530 1,969 2,616 5,346 1,706 96 5,346 1,706 1,706 1,706 684 684 684 684 684 684 684 684 684 684 684 684 684 684 684 1,222 1,223 1,76	305 6,118 1,709 3,494 (1,218) 17 17 305 205	10,711 55,438 598 56,036
Service	Policy & Partnerships Transformation Service Councils Retained ICT Budgets Council Solicitor & Democratic Services Improvement & Performance Improvement & Performance PORTFOLIO SUB TOTAL Finance Support Services Change Programme Support Services Change Programme Customer Services Services Risk & Assurance Services Risk & Assurance Services Risk & Assurance Services Corporate Estate Including R&M Commercial Estate Traded Services Strategic Director	Corporate items (Tourism Levy, Trading Opps, Community Use of Assets & Corporate Travel Plan) Hsg / Council Tax Benefits Subsidy Capital Financing / Interest Unfunded Pensions Corporate Budgets incl. Capital, Audit & Bank Charges Academies (LACSEG) Provision New Homes Bonus Grant Magistrates Coroners Coroners	PORTFOLIO SUB TOTAL Adult Services Adult Substance Misuse (Drug Action Team) Employment Development PORTFOLIO SUB TOTAL
CABINET PORTFOLIO	Leader	Association Beson Page 73	Wellbeing

Portfolio Cash Limits 2013/14 - Revenue Budgets

APPENDIX 2 ANNEX 1

Cabier Formotion Format										
	CABINET PORTFOLIO		Current 2012/13 Cash Limits	Removal of One- offs (including one- off virements in 2012/13)	2013/14 Base Budget	MTSRP Growth	MTSRP Savings		Total 2013/14 Budget Changes	2013/14 Proposed Budget
Homes Panning Panning Panning Panning Panning Panning Panning Panning Panning Panning Panning Panning Panning Panning Panning Panning Panning Panning Panning Panning Pann			000,3	000,3	000.3	000,3	000,3		£'000	000.3
Printing a limit of control turn C	-	Planning Services	2,283	(611)	1,672	877	(147)		730	2,402
	Homes &	Building Control & Land Charges	102		102	6	(18)		(6)	93
Detriction Sum Frontian 4564 (61) 3.833 900 (224) 676 Sustainable Revelopment Tartime Constrained France Tartime Constrained France (11) 252 (11) 253 (11) 253 (11) 253 (11) 253 (11) 253 (11) 253 (11) 253 (11) 253 (11) 253 (11) 253	Planning	Housing	2,179		2,179	14	(59)		(45)	2,134
		PORTFOLIO SUB TOTAL	4,564	(611)	3,953	006	(224)		676	4,629
Sustainable Memolecomment Torrism & Destination Management 1002 1002 1002 1002 10010		Arts	519	(11)	508	14	()		13	521
		Tourism & Destination Management	1,002	(80)	922	89	(201)		(111)	811
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	Sustainable	Heritage including Archives	(3,795)		(3,795)		(259)		(259)	(4,054)
	nevelopinent	Major Projects Support	571		571	17	(125)		(108)	463
Image: series between - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 -		Regeneration, Skills & Employment	921		921	7	(87)		(80)	841
		PORTFOLIO SUB TOTAL	(782)	(61)	(873)	127	(672)		(545)	(1,418)
Nate Service 10,918 10,208 17,028 1		Service Delivery - Overheads	928		928	78	(98)		(20)	906
Heile Protection Protection 1,028 31 (226) (296) (206) <td></td> <td>Waste Services</td> <td>10,918</td> <td></td> <td>10,918</td> <td>617</td> <td>(623)</td> <td></td> <td>(9)</td> <td>10,913</td>		Waste Services	10,918		10,918	617	(623)		(9)	10,913
Reighbourhoods Meighbourhood Services 5,306 (10) 5,286 (17) (240) (67) Duraries & Information 2,486 (103) 2,365 45 (240) (67) (90) Duraries & Information 2,1733 (113) 5,365 45 (240) (70) (90) Domunity Satey Transport Tarsportation Ensure 0,404 (45) 5,944 (43) (45) (45) (45) (45) (46)		Public Protection	1,028		1,028	31	(326)		(295)	733
$ \begin{array}{ l l l l l l l l l l l l l l l l l l l$	Neighbourhoods		5,306	(10)	5,296	173	(240)		(67)	5,229
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$		Libraries & Information	2,468	(103)	2,365	45	(246)		(201)	2,164
		Sports & Active Leisure	966	(44)	953	12	(103)		(06)	862
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	_	Community Safety	149		149					149
Tarsport Design & Frojects B42 664 911 7 (1) 7 (1) 7 (1) 6 1 Farsportation Planming (incl. Fubic Transport 5,944 (43) 5,942 6,943 7,370 316 (10) 1 0 3 1 <t< td=""><td></td><td>PORTFOLIO SUB TOTAL</td><td>21,793</td><td>(156)</td><td>21,637</td><td>956</td><td>(1,636)</td><td></td><td>(680)</td><td>20,957</td></t<>		PORTFOLIO SUB TOTAL	21,793	(156)	21,637	956	(1,636)		(680)	20,957
Transport Transportation Planning (no.l. Public Transport) 5,984 (43) 5,942 94 (213) (119) Park & Ride (684) $(7,13)$ (684) (3) $(7,13)$ </td <td></td> <td>Transport Design & Projects</td> <td>842</td> <td>69</td> <td>911</td> <td>7</td> <td>(1)</td> <td></td> <td>9</td> <td>917</td>		Transport Design & Projects	842	69	911	7	(1)		9	917
Tansport Park & Ride (684) 3 8 3 8 3 8 3 8 3 8 3 8 3	_		5,984	(43)	5,942	94	(213)		(119)	5,823
Highways - Network Maintenance 7,430 (69) 7,370 318 (90) 126 334 Transport Services Tansport Services (7,131) (46) 23 (122) 99) 973 Parking Services (7,131) (23) (7,153) 1,264 (291) 973 973 Peaking Services (7,131) (65) (6,332) 1,709 (717) 126 374 1 NET BUDGET 129,770 (8,905) (30,056) 120,665 9,156 1,118 1	-	Park & Ride	(684)		(684)	e			n	(682)
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $		Highways - Network Maintenance	7,439	(69)	7,370	318	(90)	126		7,724
$\begin{array}{c c c c c c c c c c c c c c c c c c c $		Transport Services	(46)		(46)	23	(122)		(66)	(145)
		Parking Services	(7,131)	(23)	(7,153)	1,264	(291)		973	(6,181)
$ \begin{tabular}{ c c c c c c c c c c c c c c c c c c c$		PORTFOLIO SUB TOTAL	6,404	(65)	6,339	1,709	(717)	126		7,457
Surplus (+) $ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		NET BUDGET	129,770	(8,905)	120,865	9,156	(11,630)	4,969		123,359
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		Sources of Funding								
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		Council Tax	77,447		77,447				(6,105)	71,342
Surplus (+) $ \begin{array}{ c c c c c c c c c c c c c c c c c c c$		Revenue Support Grant*	39,545		39,545				(8,439)	31,106
Surplus (+) 422 (422) (422) 3,872 3,872 3,872 8,483 (8,483) 3,872 129,770 (8,905) 120,865		Retained Business Rates							20,262	20,262
ax Freeze Grant 3,872 3,872 3,872 3,872 3,132 (3,132) 3,872 3,132 3,132 (3,132) 129,770 (8,905) 120,865 2,494 2,494			422	(422)					168	168
8,483 (8,483) (8,483) (259) 129,770 (8,905) 120,865 2,494			3,872		3,872				(3,132)	740
129,770 (8,905) 120,865 2,494		Balances	8,483	(8,483)					(259)	(259)
		Total	129,770	(8,905)	120,865				2,494	123,359

71,342 59,360.17 £1,201.85 **0.00%**

77,447 64,440.11 £1,201.85 **0.00%** Council Tax - Calculation Council Tax Debit £'000 Tax Base (No. of Band D equivalent properties) Band D Charge £ %age Increase

Previously Formula Grant in 2012/13

<u>Chief Financial Officer's Opinion on Adequacy of Balances and the</u> <u>Robustness of the Budget</u>

The Chief Financial Officer is required to make a statement on the adequacy of reserves and the robustness of the budget. This is a statutory duty under section 25 of the 2003 Local Government Act which states the following:

(1) Where an authority to which section 32 or 43 of the Local Government Finance Act 1992 (billing or major Precepting authority) or section 85 of the Greater London Authority Act 1999 (c. 29) (Greater London Authority) applies is making calculations in accordance with that section, the chief finance officer of the authority must report to it on the following matters-

(a) The robustness of the estimates made for the purposes of the calculations, and

(b) The adequacy of the proposed financial reserves.

(2) An authority to which a report under this section is made shall have regard to the report when making decisions about the calculations in connection with which it is made.

<u>Summary Report of the Divisional Director - Finance (as Chief Finance Officer</u> <u>for the Authority)</u>

I have examined the budget proposals contained in this report, and believe that whilst the spending, income and service delivery proposals contained remain challenging, they are achievable in terms of the requirement to set a balanced budget for 2013/2014.

I am satisfied that, in general, the requisite management processes exist within the Council to deliver this budget, and to identify and deal with any problems which may unexpectedly arise throughout the year.

The key points to highlight are:

- Un-earmarked reserves are at a prudent levels and this Budget proposes to maintain this position. This reserve will also be utilised for invest to save initiatives going forwards but at no point will the reserve fall below the risk assessed minimum level.
- This Budget provides some additional revenue reserve to recognise the challenges over the medium term financial planning period.
- Sound financial planning processes are in place, and there is member scrutiny at appropriate stages
- Revenue spend is closely monitored on a risk assessed basis and integrated finance and performance management reports are produced monthly.

- Challenging inflation targets have been set which will require a focus on commissioning and procurement to ensure they are fully delivered.
- There are a range of new and additional income targets which do increase the overall risk to the budget should they not be fully achieved.
- Changes to the local government finance system particularly the retained local business rates have been budgeted at prudent levels and whilst some flexibility has been allowed for, additional financial risk has been transferred from a national to a local level.
- Capital schemes are managed through an integrated project management, risk and financial management process.
- The intense external pressures on public finances require annually decreasing budgets and this will need to be closely managed.
- There is a need to use "invest to save" and "invest to avoid" approaches to enable the Council to maintain it's spend within acceptable parameters and to achieve requisite efficiencies.

The Medium Term Service and Resource Plans rely on the delivery of strategies and action plans within individual services and it will continue to be necessary to give a high priority to the monitoring and review the implementation of these plans as part of the performance management and monitoring processes of the Council.

The significant commitment in the capital programme in the coming years remains challenging. Sound monitoring, review and programming of schemes will need to continue, as will individual project management processes. Where projects form part of partnership arrangements, satisfactory partnership governance arrangements will need to be in place.

On the matter of unearmarked reserves, I have continued to evidence the requisite level by use of internal risk assessment. The Council is maintaining its unearmarked reserves at the appropriate risk assessed level. The 2013/14 recurring budget contains no reliance on the use of unearmarked reserves although some of these reserves (above a minimum level) will be utilised on an Invest to Save basis under the parameters set out in Appendix 2.

In view of the challenging financial climate, it remains essential that unearmarked reserves are maintained at risk assessed levels, the only prudent exceptions being to fund invest to save schemes (provided a minimum level is maintained) and to enable exceptional risks or contingencies to be funded where no other funding is available.

It will be essential in the event of any exceptional use of unearmarked reserves for the Council to continue to put in place arrangements in future years' budgets to recover the level of reserves within 3 years. From 2001/2, the Council adopted a risk management approach, which assesses the level of unearmarked reserves required against a corporate assessment of the risk being carried. The assessed risk suggests reserves of $\pm 10.5M$ for 2013/2014 with a minimum level of $\pm 6M$, excluding earmarked reserves.

My recommendation that the budget is reasonably robust and Unearmarked reserves are adequate is on the basis that the Council and the Cabinet:

- Recognise the need to continue to regularly review the level of reserves in the light of the regularly updated corporate risk assessment
- Recognise that in any use of reserves to fund 'one-off' corporate priorities on an invest to save basis, the Council needs to be clear that the overall level of reserves remains adequate and that the relevant business cases for such expenditure are fully scrutinised and monitored appropriately.
- Recognise that where there is a draw down on risk assessed reserves taking them below the prudent level, for whatever reason, this is repaid within 3 years.
- Maintain a rigorous approach to financial monitoring, particularly at this time when a significant level of savings is expected to be delivered in the financial year ahead.
- Ensures that capital schemes are funded prudently and do not rely excessively on revenue funding and do not create unaffordable revenue consequences including maintenance and other running costs
- Maintain a prudent approach to budgeting for capital receipts, given current market conditions and the need to optimise value over the medium term.
- The Cabinet Members, Strategic Directors and Assistant Directors achieve their cash limits for 2012/13.

Processes

Budget estimates are exactly that - estimates of spending and income made at a point in time. This statement about the robustness of estimates cannot give a guarantee about the Budget but gives members reasonable assurances that the Budget has been based on the best available information and assumptions at the time. The budget process aims to set challenging budgets while recognising the risk of this within its reserves strategy.

In order to meet the requirement on the robustness of estimates a number of key processes are in place, including:

• Specific guidance to Directorates on developing their budgets.

- A Council wide risk assessment.
- The continuing use of budget monitoring and financial outturn information to identify risks.
- The Council's s151 Officer and his staff providing advice throughout the process of budget preparation and budget monitoring.
- The Directors' review of the robustness of their budgets and budget sensitivities.

Notwithstanding these arrangements, which are designed to test the budget throughout its various stages of development, considerable reliance is placed on the Strategic Directors and Divisional Directors having proper arrangements in place to identify issues, project costs, service demands, to consider value for money and efficiency, and to implement changes in their service plans. This is supported by appropriately qualified financial support service staff.

Corporate and departmental processes will continue to develop over the MTSRP period to reflect the challenging financial position of the public sector. This will include the on-going implementation of risk assessed budget monitoring and enhancements to processes for monitoring implementation and delivery of savings.

Robustness of Estimates

The 2013/14 Budget and the MTSRP process continue the need to link financial resources to corporate priorities and risks. There are complex and difficult choices for the Council:

- To realise ongoing efficiencies
- To increase financial resources to meet demand and reduce risk; or
- To reduce where necessary service levels and standards, frequency of service delivery, and eligibility for services.
- To ensure all resourcing decisions reflect statutory and other external requirements, as well as Council priorities

As part of developing the Budget, Members of the administration have considered these options and they are reflected in the proposed Budget.

Most notably the Council has had to address significant reductions in government grant funding, unavoidable cost increases, and demand pressures as well as the corporate priorities including;

- Significantly reduced Government Grant funding
- Changes to government funding
- Welfare and benefits reforms including Local Council Tax Support

- Priorities as set out in the Corporate Plan
- Low levels of interest rates
- Demand and price pressures in Adult Social Care
- Demand and price pressures in Children's Social Care
- Realising and maximising capital receipts
- Need for capital investment in priority schemes
- A significant change agenda as a result of national policy changes e.g. Adult Social Care and Health, Academies Bill, Localism Bill etc.

The assumptions used for the 2013/14 and MTSRP period will require the forecasts for future years to be reviewed in light of actual circumstances. This will be undertaken early in each financial year leading to more detailed budgets being prepared for the next financial year and the medium term during the autumn of each financial year.

Given all these factors I, as the Council's Section 151 Officer, consider the estimates for 2013/14 to be sufficiently robust, and the reserves adequate, to be recommended for approval by the Council.

The medium term remains extremely challenging although the more detailed MTSRP's developed as part of this budget will place the Council in a relatively good position over this period. It is still highly likely that further service improvement and reasonable Council Tax levels, will only be achievable through continued development of different ways of working, and clear prioritisation between services.

The Capital Budget

Projects included in the capital programme for **Full Approval** were prepared by Divisional Directors and managers in line with financial regulations and guidance. All projects were agreed by the relevant Strategic Director and Cabinet Member and are fully funded, and reviewed through the Capital Strategy Group.

Projects have been estimated and costed at outturn prices with many subject to tender process after inclusion in the programme. This may lead to variance in the final cost.

Directorates are required to work within the given cash envelope so any under or over provision must be found within these limits.

In addition, I will require a clear commitment from the Council to:

• Ensure that all future commitments on the capital programme provide for a prudent source of funding in terms of revenue provision, including where investment will lead to future revenue savings.

- Carefully consider and balance the use of capital receipts to ensure they are prudently applied to help the council manage its resources effectively and achieve its priorities
- Review capital commitments in light of any future changes to Central Government support for capital projects where they are dependent on substantial Government funding.
- Be aware of the potential risks associated with capital spend before the scheme is completed i.e. the potential for costs charged to capital budgets to revert to revenue in the event schemes are discontinued prior to completion.

I will require the Council, Strategic Directors and Divisional Directors:

- To remain within their service budget for 2013/14 and to commence early work to identify options to balance medium term service and resource plans for future years while maintaining strict adherence to recovering overspends within future years' plans (i.e. services will need to absorb any overspends, pressures over the medium term).
- Repayment to reserves over 3 years should risks materialise, and need to be funded temporarily from reserves.

Estimated Available Revenue Reserves

Earmarked Revenue Reserves

The Council's earmarked revenue reserves have been reviewed as part of the 2013/2014 Budget process and are fully committed over the medium term financial planning period as set out in Appendix 2. This position will be regularly reviewed.

Non-Earmarked Revenue Reserves

Detailed in the table below is the estimated level of non-earmarked revenue reserves over future years, reflecting the specific elements within the Budget as set out in Appendix 2.

	2013/14 £'000	2014/15 £'000	2015/16 £'000
Estimated Reserves @ 1st April each year	9,741	8,844	7,728
2011/2012 Projected Outturn Underspending	-	-	-
Projected Invest to Save Movements	-897	-1,116	+134
Estimated Reserves @ 31st March each year	8,844	7,728	7,862

Table: Projected Non-Earmarked Revenue Reserves

Assessment of Adequacy of Reserves

Under the Local Government 2003 Act the Secretary of State has reserve powers to set a minimum level of reserves. The most likely use of this power is where an authority is running down its reserves against the advice of their s151 Officer.

Determining the appropriate levels of reserves is not a precise science or a formula e.g. a %age of the Council's budget. It is the Council's safety net for risks, unforeseen or other circumstances and must last the lifetime of the Council unless contributions are made from future years' revenue budgets. The minimum level of balances cannot be judged merely against the current risks facing the Council as these can and will change over time.

Determining the appropriate levels of reserves is a professional judgement based on local circumstances including the overall budget size, risks, robustness of budgets, major initiatives being undertaken, budget assumptions, other earmarked reserves and provisions, and the Council's track record in budget management. This judgement is subject to regular review as an integral part of the Council's financial reporting cycle, and annual review by full Council as an integral part of budget-setting and 3 year financial planning. Clearly, as circumstances change, the currently recommended level of reserves can be expected to change.

The consequences of not keeping a minimum prudent level of reserves can be serious. In the event of a major problem or a series of events, the Council would run a serious risk of a deficit or of being forced to cut spending during the year in a damaging and arbitrary way.

The recommendation on the prudent level of reserves has been based on the robustness of estimate information and the Corporate Risk Register. In addition, the other strategic, operational and financial risks taken into account when recommending the minimum level of unearmarked reserves include:

- There is always some degree of uncertainty over whether the full effects of any economy measures and/or service reductions will be achieved. Directors have been requested to be prudent in their assumptions and should have clear action plans to deliver such savings.
- The Bellwin Scheme Emergency Financial Assistance to Local Authorities provides assistance in the event of an emergency. The Local Authority is able to claim assistance with the cost of dealing with certain emergencies over and above a threshold set by the Government.
- The extent to which the Council is dependent on traded income.
- The risk of major litigation, both current and in the future.
- Risks in the inter-relation between the Council and other partner authorities and organisations.
- Unplanned volume increases in major demand led budgets, particularly in the context of high and accelerating growth.
- The need to retain a general contingency to provide for any unforeseen circumstances or emergencies, which may arise.
- The need to retain reserves for general day-today cash flow needs.

The recommendations of the Council's s151 Officer are:

- That the Council continues to maintain an absolute minimum prudent level of unearmarked reserves (excluding schools) of £6m at the end of any financial year, in addition to any specific earmarked reserves. The minimum level is designed to cope with risk and unforeseen circumstances that cannot be addressed by management or policy action within the year. Management and policy action should be the first actions taken before any resort to reserves.
- That an appropriate level of unearmarked reserves to provide resilience against day to day risks is £10.5m. This level of reserves is designed to allow the Council to withstand a measure of changes in circumstances during the year or minor variations in projected resources or spending over the period of the medium term service and resource plans.
- That the Council must restore reserves used to meet risks that crystallise within a period of 3 years.

Capital Programme - 2013/14 - 2017/18

APPENDIX 2 ANNEX 3

				\vdash	Budget	Budget	Budget	Budget	Budget	Budget			To	Fotal 5 Years Funding	pding	
	PY Spend pre 12/13	Forecast Outturn 12/13	Actual / Projected Spend pre 13/14	Frojected Rephasing from 12/13 to 13/14	Required 2013/14	Total 2013/14	2014/15	2015/16	2016/17	2017/18	Total Cost 5 Years	Overall Project Total	Borrowing / Capital Receipts	Grants / External Funding	RIF / Development Funding	Comment/Project Description
Project Title PLACE	000,3	000.3	000,3	000,3	000.3	000.3	000,3	000,3	000,3	000,3	000.3	000,3	000,3	000,3	000,3	
Planning & Transport Eull Approval BTP - Main Scheme BTP - Pre Construction RTP - Property Ansater Pand	7,156 1,315 216	4,806 389 208 86	4,806 7,545 1,523 302	340 - 1,532 -	9,615 - 766	9,955 110 766	4,954 766	1,449			16,358 - 1,532	21,164 7,545 1,633 1,834	11,316 - 1,532	ο, · · · Ω		Fully approved in July 2011 Fully approved in July 2011 Fully approved in July 2011 Fully approved in 2012/13
owier e-Purse Sub Total - Planning & Transport	8,756	322 5,811	391 14,567	ی 1,991	8,849	10,840	5,720	1,449		•	18,009	400 32,576	12,958	5,051		Fully approved in 2012/13
Environmertal Services Full Approval 20mph Schemes Wichola Bridge Bridge Brittel Services - Vehicles Naste Services - Vehicles Neighbourhood Services - Vehicles	714	259 575 320	259 1,289 	838	241 885 171 350 174	241 1,723 171 350 174	70 405 559				311 2,128 730 350 174	570 3,417 1,050 350 174	241 1,628 - 350 174	70 500 730		Fully approved in 2012/13 Fully approved in 2012/13 Fully approved in line with October 2012 Cabinet Report Proposed for full approval Proposed for full approval
Provisional Approval Highways Maintenance Block Highways Maintenance Block - Additional Funding Tarasport Improvement Programme Mattale Services - Vehicles Neighbourhood Services - Vehicles Highmann - Vehicles Public Programme Highways Maintenanco - Vehicles				75	3,667 663 1,264 82 225 15	3,667 663 1,264 157 157 225 15	3,435 3,53 1,723 133 695 225 225 110	3,300 1,723 278 151 115	3,300 1,723 539	558 110 85	13,702 1,016 6,433 411 157 1,943 565 565 565 220 220	13,702 1,016 6,433 411 157 1,943 565 565 360 220		13,702 1,016 6,433 - - -		Subject to March Cabinet Report subject to March Cabinet Report Subject to sparset February Council Report Business case required amunaly for 2014/15 onwards Business case and detailed project plan required Business case and detailed project plan required
Arking - Hadio System Replacement Arking - Pay & Display Replacement Programme Arking - Enforcement Hand Held Computer Terminal									50 50 80	350	45 400 80	4004 80	45 400 80			business case and detailed project plan required Business case and detailed project plan required Business case and detailed project plan required
Boatement Boatement Adoring Changes - Setup Costs Adoring Changes - Setup Costs Adoring Changes - Ray and Bench Replacement Bus Land Cannear Replacement Cycle Routes Reve Condor Fund					224 100 50 100	224 100 50 100	20	20	20 00	300	224 200 200 300 500 100	224 200 200 200 200 200 200 200	224 100 200 300 500 100			courses actuation of the project plant required Business case and detailed project plant required Business case and detailed project plant required Business case and detailed project plant required Detailed project plant availed Detailed project plant availed
Sydney Gardens Sub Total - Environmental Services	714	1.154	- 1.868	913	250 8.961	250 9.874	250 8.023	5.617	5,787	1 418	500 30.719	500 32,587	500 8.268	2.451		Detailed project plan awaited & awaiting funding confirmation from Heritage Lottery Fund
court court - Entrational control - Entrational - Entrationa - Entrational - Entrationa - Entrational - Entrational - Entrational - Entrational - Entrationa - Entrationa - Entrationa - Entrationa - Entrationa - E	<u>:</u>	101	101	2	1,131 200	1,131		20	5		1,131	1,232	200			Fully approved in line with Jan 13 cabinat report Availing funding confirmation from external sources
Provisional Approval Beau Street Coin Hoard Heriage Infrastructure Development Amana Baths Development Phase 2 Visitor & TiT Management System	50		50	100	170 320 750 100	270 320 750 100	153 100 -	100 500	200 500	100	423 820 1,750 200	473 820 1,750 200	- 820 1,750 200			Business case and detailed project plan required Business scase and detailed project plan required Business case and detailed project plan required Business case and detailed project plan required
Sub Total - Tourism, Leisure & Culture	50	101	151	100	2,671	2,771	353	600	200	100	4,524	4,675	2,970	1,554		
	9,520	7,066	16,586	3,004	20,481	23,485	14,096	7,666	6,487	1,518	53,252	69,838	24,196	29,056		

	Comment/Project Description			Fully Approved - Continuation of Phase 1 of project	Fully Approved - Continuation of Phase 1 of project Fully Approved - Continuation of Phase 1 of project	Fully Approved by Cabinet in January 2012 Fully Approved - Continuation of project	Fully Approved - Continuation of project	r uny Approved - Continuation of project Fully Approved - Continuation of project Fully Approved - Socontinuation	ruing approved in 2012/13	Detailed project plan required	Detailed project plan required Detailed project plan required	Detailed project plan required	Detailed project plan required Subject to March Cabinet Report	Provisional approval pending Cabinet report in April Provisional approval pending Cabinet report in April					Fully Approved - completion of project Fully Approved - completion of project	Proposed for full approval	Fully Approved - completion of project Fully Approved - completion of project	Fully Approved - completion of project	Froposed for full approval - Subject to confinitiation of grant funding Fully approved in 2012/13	Fully approved in 2012/13 Fully approved in 2012/13		Detailed project plan required & confirmation of grant funding	Detailed project plan required & continuation of grant tunding Detailed project plan required & confirmation of grant funding	Detailed project plan required Business Case & detailed project plan required & subject to external	funding opportunities			Fully Approved	Froposed incruit approvat Fully Approved	Date and and some some some some some some some some	Business Case & detailed project plan required	Business Case & detailed project plan required	
Di	RIF / Development Funding	000.3									2.500	3,000		4,000	11,200	11,200																					
Total 5 Years Funding	Grants / External Do Funding			468	2,009 3,758			,	070	,					6,858	6,858			1,131	1,000	42 26	51	402	725 690		2,403	72	202	290	10,390		220 F	488	1000	-	5,565	15,955
Total 5	Borrowing / Capital Receipts	+		472	1,821 279	690 424	80 75	70	000	339	- 650	- 010	340		6,405	6,405			1 050	-									200	1,550					1,100	1,550 2,650	4,200
	Overall Project Total	000.3		1,888	6,900 7,499	690 1,529	247 346	1,400	c/c/1	339	650 2.500	3,000	340	4,000 1 700	35,603	35,603			2,447 0 375	1,000	26,498 3,450	102	1,800	800 850		2,403	76	202	290	44,759		150	588		1,200	8,738	53,497
	Total Cost 5 Years	000,3		940	3,830 4,037	690 424	80 75	202	1,1/0	339	650 2.500	3,000	340	4,000 1 700	24,463	24,463			1,131	1,000	42 26	51	1,670	725 690		2,403	+204	202	290	11,940		200 F	488		1,100	8,215	20,155
Budget		000,3																												•				1 000	000	1,000	1,000
Budget	2016/17	000,3			450										450	450																		1 000	000'-	1,000	1,000
Budget	2015/16	000,3		154	1,000										1,154	1,154								20						20				1 000	000	1,000	1,020
Budget Budget Budget Budget	2014/15	000,3		234	1,000 1,800	230					2.250	2,750	067	2,500	11,014	11,014							680	475 20					395	1,570				1 000	550	2,325	3,895
		000,3		552	1,380 2,237	460 424	80 75	202	1,1/0	339	650 250	250	360 340	1,500	11,845	11,845			1,131	1,000	42 26	51	402 990	230 670		2,403	72	202	395	10,350		2000 F	488		550	2,890	13,240
Budget Budget	Required 2013/14	000.3		234	1,000 2,200	230		07E	6/6	339	650 250	250		1,500	8,728	8,728			52 1 050	1,000		007	404 990	230 670		2,403	72	202	395	9,030		12 12	000,1		150	2,002	11,032
	Projected Rephasing from 12/13 to 13/14	000,3		318	380 37	230 424	80 75	02	000			096	340 340		3,117	3,117			1,079	1 9	42 26	51								1,320			488		400	888	2,208
Г	Actual / Projected Spend pre 13/14	000,3		948	3,070 3,462	- 1,105	167 271	1,330	180			-			11,140	11,140			1,316		26,456 3,424	51	130	75 160			4			32,819		73	100		100	250 523	33,342
	Forecast F Outturn S 12/13	000,3		237	1,070 935	1,105	167 271	233	077			000	080		4,628	4,628			1,144		163 21	51	130	75 160						2,872		73	100		100	523	3,395
	PY Spend pre 12/13	000.3		711	2,000 2,527			1,097							6,512	6,512			172 75		26,293 3,403						4			29,947							29,947
_		Project Title	REGENERATION SKILLS & MAJOR PROJECTS	Full Approval BWR - Council Project Team	BWR - Affordable Housing BWR - Infrastructure	BDUK Public Realm High Street	Public Realm Northumberland Place	r dolor heatin r attent book City Information Scheme NDB Informations		Provisional Approval Public Realm Improvements Programme	Bath Quays South Bath Quavs Footbridge Construction	Strategic Flooding Solution	London Hoad Hegeneration Radstock Regeneration	BWR - Relocation of Gas Holders BWR - Renjacement of Destructor Bridge	Sub Total - Regeneration Skills & Major Projects	Total REGENERATION SKILLS & MAJOR PROJECTS	PEOPLE & COMMUNITIES	Children's Services Full Approval	Balph Allen ALC	Chools Capital Maintenance Programme		Early Years S106 Twerton	Story Devolved Capital	Castle Primary - Basic Need Paulton Infant - Basic Need	Provisional Approval	Schools Capital Maintenance Programme	Schools basic need schemes Short Breaks for Disabled Children	Early Years - 2yr Olds Funding	School Energy Investto Save Fund	School Places Provision Sub Total - Children's Services	Adult Social Care & Housing	Supported Housing Development	Disabled Facilities Grant PSS Grant Unallocated	Provisional Approval Disabled Escrittice Grout	Affordable Housing	Gypsy & Traveller Sites Sub Total - Adult Social Care & Housing	Total PEOPLE & COMMUNITIES

		ľ	ſ		Budget	Budget	Budget	Budget	Budget Budget	Budget			Total	Total 5 Years Funding	lind	
	PY Spend pre 12/13	Forecast Outturn 12/13	Actual / Projected Spend pre 13/14	Projected Rephasing from 12/13 to 13/14	Required 2013/14		2014/15	2015/16	2016/17	2017/18	Total Cost 5 Years	Overall Project Total	Borrowing / Capital Receipts	Grants / External Funding	BIF / Development Funding	Comment/Project Description
Project Title	000,3	000,3	000,3	000,3	000,3	000,3	000,3	000,3	000.3	000,3	000.3	000,3	000,3	000,3	000,3	
Resources & Support Services																
Property Services Full Approval Workplaces Programme Delivery Kerynsham Regeneration & New Build Lewis House (Inc Comms Hub & OSS) The Holles	2,648 1,480 4,438	995 4,339 1,116 100	3,643 5,819 5,554 1,877	- 3,781	339 19,275	328 23,056	3,114 5,300	4			3,446 28,356 -	7,089 34,175 5,554 1,877	3,446 28,356 -			Fully Approved Fully Approved Fully Approved
Corporate Estate Planned Maintenance Disposals Programme (Minor) Key Disposal Programme Commercial Estate Investment Fund					825 200 250 200	825 200 250 200	250				825 200 500 200	825 200 500 200	825 200 500	200		Proposed for Full Approval Proposed for Full Approval Proposed for Full Approval Proposed for Full Approval Pronosed for Full Aproval
Victoria Hall Saw Close Development South Road Car Park MSN	1	38 38	, 38 , 38	125 242	715	840 242 155					840 242 155	875 280 155	840 242 155			reposed for Full Approval Proposed for Full Approval
Provisional Approval Corporate Estate Phanned Maintenance Equality Act Works Deposate Programme (Minor) Grand Parade & Underrott Riverside Davelopment (in C PO) SUD Total - Property Services	10,343	6,623		4,137	400 22,359	 400 26,496	905 552 200 4,000 14,321	905 552 200 1,661	905 552 200 1,657	905 552 200 1,657	3,620 2,208 800 4,400 45,792	3,620 2,208 800 4,400 62,758	3,620 2,208 800 4,400 45, 592	200 200		Amnual detailed project plan required Amnual detailed project plan required Amnual detailed project plan required Amnual detailed project plan required Subject to feasibility study & business case Subject to feasibility study & business case
Support Services Full Approval Destrop As Service- VDI Technology Windows 7 Ubgrade Agress Ohani System Update Performance Reward Grant	175	7 371 203	- 182 371 203	6 704 297	1,468 400	1,468 400 6 704 297	115	4	159	134	1,880 400 6 704 297	1,880 400 1,075 500	1,880 400 704	297		Proposed for full approval Proposed for full approval Proposed for full approval Proposed for full approval Fully Approved
Covisional Approval Or Mobile Devices - Pilot Or Mobile Devices - Pilot Ortopie & Communities - IT System Replacement Bornass Energy Efficiency Fund Sub Total - Support Services	175	581	- - 756	1,007	150 - 250 2,268	150 - 250 3,275	350 250 25 0	450 500 954	400 250 809	100 234	150 1,300 1,000 6,237	150 1,300 1,000 6,993	1,300 1,300 1,000 5,940	- - - 297		Business Case & detailed project plan required Indviout detailed project plans required Business Case & detailed project plan required Business Case & detailed project plan required
Total Resources & Support Services	10,518	7,204	17,722	5,144	24,627	29,771	15,286	2,615	2,466	1,891	52,029	69,751	51,532	497		
Capital Contingency Grand Total	56,497	22,293	78,790	13,473	1,500 66,368	1,500 79,841	44,291	12,455	10,403	4,409	1,500 151,399	_	1,500 87,833	52,366	11,200	Proposed for full approval

This page is intentionally left blank

APPENDIX 2 ANNEX 3i

Schools C	apital Planned Maintenance F	Programme 201	13-14
School	Project	Cost Est.	Cumulative Total
Moorlands Infant School	Repair Roof	£136,235	£136,235
Whitchurch Primary School	Flooring and damp proofing	£29,755	£165,990
Moorlands Junior School	Renew roof lights & renew roof, Upgrade/install ventilation canopy in kitchen & kitchen refurbishment	£234,850	£400,840
St Michaels CofE Junior School	Ceiling cover and roof lights	£18,000	£418,840
Chew Valley School	Gas heating in sports hall	£57,015	£475,855
Chew Valley School	roof works to the tech block	£104,577	£580,432
St Philips CofE Primary School	Replace boiler	£43,417	£623,849
Moorlands Infant School	Rewire power and lighting	£143,245	£767,094
Batheaston CofE Primary School	Refurbish pupil toilets	£45,550	£812,644
	Fee	£120,000	£932,644
	Contingency	£67,356	£1,000,000
	Total	£1,000,000	

This page is intentionally left blank

Council MRP Policy

Bath and North East Somerset will make a prudent minimum revenue provision for all new unsupported borrowing from 1st April 2008.

For all Government Supported Borrowing

a) The Council will determine that its MRP is equal as the amount determined in accordance with the former regulations 28 and 29 of the 2003 Regulations, as if they had not been revoked by the 2008 regulations.

For all new schemes of Council supported borrowing after 1st April 2008 it will use the Asset Life Method

b) This will be calculated where capital expenditure on an asset is financed wholly or partly by borrowing or credit arrangements, MRP is to be made in equal annual instalments over the life of the asset, in accordance with the following formula:

Where-

A is the amount of the capital expenditure in respect of the asset financed by borrowing or credit arrangements

B is the total provision made before the current financial year in respect of that expenditure

C is the inclusive number of financial years from the current year to that in which the estimated life of the asset expires.

c) Subject to paragraph f below, MRP will normally commence in the financial year following the one in which the expenditure was incurred.

d) Asset life. The estimated life of the asset will be determined in the year that MRP commences and not subsequently be revised.

e) Freehold land. If no life can reasonably be attributed to an asset, such as freehold land, the life will be taken to be a maximum of 50 years. However, in the case of freehold land on which a building or other structure is constructed, the life of the land will be treated as equal to that of the structure, where this would exceed 50 years.

f) Construction period. When borrowing to construct an asset, the authority will treat the asset life as commencing in the year in which the asset first becomes operational. It may accordingly postpone beginning to make MRP until that year. "Operational" here has its standard accounting definition. Investment properties will be regarded as becoming operational when they begin to generate revenues.

This page is intentionally left blank

TAILS
DE DE
VIN
– S/
LAN
URCE PLAN
OUR
RESC
CE &
SERVIC
TERM
ШМ

CHILDRENS' SERVICES

S
×
₩
Ź
A
2
×
Ξ
Z
ш
Δ
Δ.
◄

2013-14 Saving £000	2014-15 Saving £000	2015-16 Saving £000	Risk to Delivery	Impact on staff	Impact on Assets and Property	How saving to be achieved	Impact to Service Delivery	Additional Information (Inc. PDSP Feedback)
50	58	4	Σ	1.7fte	None	Savings identified from the customer services workstream which looks at the redesigning the customer pathway making better use of IT systems and implementing streamlined processes a (including family information)	Yet to be determined. Service will transfer work to the Customer Stervice equivalent to this reduction. Savings will need to be made available to the School Forum as partly DSG funded.	
31			Σ	1.5 fte	None	P2P Efficiency savings	Restructure of administration in relation to Invoice payment and purchase orders	
51	28	4	Sub Total – Change Programme	0,	bavings 5 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1			
2013-14 Saving £000	2014-15 Saving £000	2015-16 Saving £000	Risk to Delivery	Impact on staff	Impact on Assets and Property	How saving to be achieved	Impact to Service Delivery	Additional Information (Inc. PDSP Feedback)
_{ଞ୍ଚ} Page 91		100	L	3/4 fte	None	Management restructure. Deletes 1 X Divisional Director post, 1 X Service Manager and 1 X Team Manager post from 1 April 2013	Leaves four Divisional Directors to lead the re-structured Department across 2013-15. Merges a number of management posts across these services.	
200	0	100	Sub Total – Other	Sub Total – Other Cashable Efficiency Savings	sy Savings			
2013-14 Saving £000	2014-15 Saving £000	2015-16 Saving £000	Risk to Delivery	Impact on staff	Impact on Assets and Property	How saving to be achieved	Impact to Service Delivery	Additional Information (Inc. PDSP Feedback)
16	5		Γ	anon	None	Growth in sales of childcare vouchers		
16	5	0	Sub Total – Additional Income	onal Income				

2015-16 Saving £000	Risk to Delivery Impact on staff and Property How saving to be achieved and Property	Impact to Service Delivery	Additional Information (Inc. PDSP Feedback)
c	M 1.77 FTE N/A School Improvement and Achievement Services	Following a 30% (£373k) reduction over 2011-12 and 2012-13 and the move towards more commissioned support for schools, further reductions will have a significant impact on our ability to deliver our statury duties of promoing school improvement and challenging underperformance. This saving would require whole team restructure and realignment of responsibility with some tasks moved to managers. This would reduce our advisory work in schools to only those schools identified as failing or likely to fail and remove early advice/support. This further shifts the balance to commissioned support.	
	Further re-structuring of our support to schools and consequential restructuring of administrative support	pport The LA will hold further discussions al with schools about the respective roles and responsibilities for school improvement.	
0	13/14 = L 13/14 = L Children Missing Education Service 1 FTE None In 13/14 cease remaining activity to secondary schools.	Service reductions 2012/13 to a Primary only and prosecution (on behalf of schools in most extreme cases) service. This will reduce ity to support for schools to achieve high levels of attendance and pupil tracking of more vulnerable children, where they are removed from a	
	In 14/15 £26k reduction plus income target of £4k	school roll or where no school place is immediately available. Further reduction of 1 FTE, limiting service to statutory work and little preventative work on behalf of schools. This could result in a higher level of absence, which could impact on attainment, more schools categorised as having increasing levels of persistent absence and adverse inspection (Ofsted)	NB as Primary schools become academies funds must be delegated to them, reducing central funds.

2013-14 Saving £000	2014-15 Saving £000	2015-16 Saving £000	Risk to Delivery	Impact on staff	Impact on Assets and Property	How saving to be achieved	Impact to Service Delivery	Additional Information (Inc. PDSP Feedback)
30			L	0		Restructure Educational Psychology Service to undertake only statutory work	Restructure of the Educational Psychology service as part of the People and Communities restructure. Move to Statutory work only related to pupils with statements. Schools could commission work from this service (generating income to provide more capacity) or from other providers.	
20	37		×	3 FTE posts	None	Inclusion Support Service restructure of administrative support	Restructure of the Inclusion support service as part of the People and Communities restructure to remove posts. Impact on workload and efficiency of the wider team.	
≌ Page 9			L	-0.2 FTE (25%)	N/A	Post 16 Education Training and NEET Reduce staff FTE working on raising participation	Majority of funding directly on projects for more vulnerable 16-19 year olds. Reduction in staffing (currently 0.8ftb) would mean reduced capacity to support raising the participation age and to work with young people Not in Education Employment or Training. (NEET).	
₽ 23			L	ī	Ē	Post 16 Education, Training and NEET Reduce commissioning spend	Reduced capacity to commission work to reduce the numbers of young people Not in Education Employment or Training. (NEET).	

Additional Information (Inc. PDSP Feedback)		
Impact to Service Delivery Add	Connexions Service would be merged with Youth Service would be focused provision for 16-25 yr olds. The new combined service working with referred vulnerable young people. Connexions contract with Learning Partnership West to be terminated as our statutory duties are reduced and remaining statutory duties (such as Section 139A Assessments for Young People with LDD and monitoring and support for most vulnerable young people likely to become NEET) brought in-house. LA would have to meet redundancy and pension costs. Youth work to continue the move to increased targeted services. Work in rural areas where deprivation is lower (although there are issues of transport and accessibility) would ceases or be much reduced. Reduced number of sessions that centres would be open with less access for non-referred young people.	The savings from capital team will result in less resources be available through the capital programme for maintenance and improvement in schools. Some officers of the team may be charged to the capital programme. This will have a limited impact upon the number of projects undertaken.
How saving to be achieved	Connexions – Discontinuation of current form of service	Schools Capital and Reorganisation Team Reallocation of costs to capital programme
Impact on Assets and Property	NA	
Impact on staff	7 FTE	o
Risk to Delivery	ΗŴ	Σ
2015-16 Saving £000		
2014-15 Saving £000	480	60
2013-14 Saving £000	Page 94	

ce Delivery Feedback)		Most significant retaining one or BME, CYP.			hat the range of nal services vulnerable du. Core du. Core du there will be valiable which fe for some e and their e and their care pathway.	hat the range of nal services vulnerable d. Core but there will be raitable which fe for some e or their e or hidren may om the more care pathway.	hat the range of nal services vulnerable d. Core but there will be valiable which fe for some e and their e and their e and their care pathway. om the more care pathway. thereak thereak thereak thereak thereak thereak thereak thereak thereak
Impact to Service Delivery	(i) 2013-14	Impact will be small. Most significant will be SARI. We are retaining one key support service for BME, CYP.	ă	(ii) 2014-15 ices	Sig ancor gro gro ser tam tam tam 'aci		
and Property How saving to be achieved	Reductions in commissioned services.	(i) 2013-14	Removes additional funds to commission extra CAMHS type services and cuts spend on IT Software Licences. Reduces lower	priority elements of service such as Contact Centre (private law cases) and Relate. Reduces spend on BME/Equalities advice, CYP services and requires providers to make efficiency savings.	priority elements of service such as Contact Centre (private law cases) and Relate. Reduces spend on BME/Equalities advice, CYP servic and requires providers to make efficiency savings. (ii) 2014-15	priority elements of service such as Contact Centre (private law cases) and Relate. Reduces spend on BME/Equalities advice, CYP services and requires providers to make efficiency savings. (ii) 2014-15 (ii) 2014-15 (ii) 2014-15 (ii) 2014-15 Parafise for a range of additional /complementary services for a range of groups: Anti- bullying; Family Support (Southside); Play Rangers; Short Breaks (CYP with disabilities); CYP Participation and commissioning capacity.	priority elements of service such as Contact Centre (private law cases) and Relate. Reduces spend on and requires providers to make efficiency savings. (ii) 2014-15 Reduces funding available for a range of additional /complementary services for a range of groups: Anti bullving; Family Support (Southside Play Rangers; Short Breaks (CYP with disabilities); CYP Participation and commissioning capacity.
	s S/L						
	U.5 F IE Internal. Not known how many posts lost in volumtary sector.						
	175						
_	178				Page 95	Page 95	Page 95

Children's Centres will coase all Council-funded 'universal' access services with services targeted upon the most vulnerable/deprived children and families.Lack of good universal services will mean that children may not be identified in a timely way and therefore could put increasing pressure on Children's Social Care. Parental mental health will not be identified early and add pressure to Adult Services or Health. Early Years Service contributions to prestilist services e.g. breastfeeding, specialist services e.g. breastfeeding, specialist services e.g. breastfeeding, specialist community focus and health and post-natal parent/child attachment will cease. The Children's Centres also provide an increasing community focus and access point and a move to targeted service and partial opening will affect this aspect of work. Children's Centre budgets will be reduced by beween	hole service re-organisation of nildren's Centre and Early Years arvices with work starting in nuary 2013 to achieve complete re- ucture and consideration of ferent models of service delivery to farent models of service delivery to fully implemented by April 2015.	Whole service re-organisation of Children's Centre and Early Years Services with work starting in January 2013 to achieve complete re- structure and consideration of different models of service delivery to be fully implemented by April 2015.	30 FTE & 2 FTE modern modern apprentice) be fully implemented by April 2015.	Щ	30 FTE & 2 FTE (modern apprentice)
je .	2013-14 and 2014-1 irvice areas and corr rvices will be reduce nese will be in additio mmissions being reo nildren's Commissior	In 2013-14 and 2014-15 a number of service areas and commissioned services will be reduced or stopped. These will be in addition to those commissions being reduced by the Children's Commissioning Team.	In 2013-14 and 2014-1 service areas and corr services will be reduce These will be in additic commissions being red Children's Commission	In 2013-14 and 2014-1 service areas and com services will be reduce These will be in addition commissions being red Children's Commission	In 2013-14 and 2014-1 service areas and com services will be reduce These will be in additic commissions being red Children's Commission

Additional Information (Inc. PDSP Feedback)				
Impact to Service Delivery		Council funding for YOS is only 2261,000 out of a total budget of 2700,000 with Youth Justice Board grant and other partners contributing two thirds of funding. Therefore reductions risk partnership arrangements.	Net cost to Council has been reduced to minimal level £50K after Government Grant. This funding leverages in national grant of £750k per annum for Bath NES to be a regional music hub. Removal of Council funding therefore risks loss of grant. Cosis to parents would increase impacting adversely on those least able to pay.	This would significantly reduce the Service's capacity to provide targeted services to prevent family breakdown (in accordance with its statutory duties under section 17 Children Act 1989). This would impact upon the Service's ability to divert children and young people age 11-16 years from care, and would inevitably lead to an increased demand for care placements and budgetson average care placements cost £20k- £40k per annum.
How saving to be achieved	A reduction in savings for Children's preventative services - to be prioritised	Management restructure in children's services will merge responsibilities at tier 3 and reduce management costs.	Music service	Safeguarding, Social Care and Family Support Services Review 117 Service and redesign overall provision of Family Support Services with reducing staffing capacity.
Impact on Assets and Property				
Impact on staff		1 fte	Increased charges not reduced Staffing as staffing directly relates to music provision	2
Risk to Delivery		Σ	Σ	т
2015-16 Saving £000		25	20	37
2014-15 Saving £000				28
2013-14 Saving £000	-100		Page 97	92

2013-14 Savind	2014-15 Savind	2015-16 Savind			Impact on Assats			Additional Information (Inc. PDSP
0003	5000	0003	Risk to Delivery	Impact on staff	and Property	How saving to be achieved	Impact to Service Delivery	Feedback)
છ	27	35	т	4fte		Family Support.	This would significantly reduce the Service's capacity to provide targeted services to prevent family breakdown (in accordance with its statutory duties under section 17 Children Act 1989). This would impact upon the Service's ability to divert children aged 5-11 years from care, and would inevitably lead to an increased demand for care placements and budgetson average care placements cost £20k- £40k per annum.	
					/	Review Specialist Child and Family Support Service and redesign overall provision of family support services with reduced staffing capacity.		
H 628	1130	1983	Sub Total	Sub Total – Reduced Service Levels	e Levels			
2003-14 Saving 0 £000 60	2014-15 Saving £000	2015-16 Saving £000	Risk to Delivery	Impact on staff	Impact on Assets and Property	How saving to be achieved	Impact to Service Delivery	Additional Information (Inc. PDSP Feedback)
8								
0	0	0	Sub Tot	Sub Total – Discontinued Service	Service			
895	1163	2087		TOTAL SAVINGS				

Additional Information (Inc. PDSP Feedback)		Additional Information (Inc. PDSP Feedback)					Additional Information (Inc. PDSP Feedback)	
Impact to Service Delivery		Impact to Service Delivery	Already accommodated in service planning	Yet to be determined. Service will transfer work to the customer service equivalent to this reduction	Restructure of administration in relation to Invoice payment and purchase orders		Impact to Service Delivery	Limited service impact as there is less need to fund out of area residential treatment as a consequence of improvements to the care pathway and effectiveness of community treatment. The treatment system is dependent on national performance related funding. Further reductions in service from council or NHS locally will put performance and therefore treatment system investment at risk. Lack of effective substance misuse services can adversely impact on the community, including escalation of dug and alcohol related crime and anti-social behaviour. Loss of commissioning capacity will increase workload pressures.
How saving to be achieved		How saving to be achieved	Decrease in Sirona contractual values as agreed.	Savings identified from the customer services workstream which looks at redesigning the customer pathway making better use of IT systems and implementing streamlined processes (including family information)	P2P Efficiency savings		How saving to be achieved	Savings in commissioning substance misuse services to be achieved primarily through reduced commissioning staff capacity with a small saving to be achieved by reducing spend on residential treatment by "holding" people in treatment by "holding" people in treatment by "holding" people in treatment by "holding" people in treatment by the treatment the now achieving significantly improved outcomes following pathway redesign.
Impact on Assets and Property		Impact on Assets and Property	None	None	None	mme Savings	Impact on Assets and Property	a N
Impact on staff		Impact on staff	None	2.6 fte	0.75 fte	 Change Programn 	Impact on staff	1.0 FTE
Risk to Delivery		Risk to Delivery	н	×	M	Sub Total –	Risk to Delivery	Σ
2015-16 Saving £000	NG	2015-16 Saving £000	296	O		296	2015-16 Saving £000	
2014-15 Saving £000	ADULT SOCIAL CARE & HOUSING	2014-15 Saving £000	293	8		332	2014-15 Saving £000	
2013-14 Saving £000	ADULT SOCIAL	2013-14 Saving £000		ŝ	15	24 24	20 ෆ 3-14 Saving රි £000	Ω

Additional Information (Inc. PDSP Feedback)	Change will require short-term investment in change management and investment in targeted advice and information, including to self- funders.		
Additional Info	Change will require short-term investment in change managemen and investment in targeted advice and information, including to self- funders.		
Impact to Service Delivery	Any service impacts would need to be assessed in light of the detailed savings plans, to be developed and agreed during 2013/14. The Audit Commission report suggests that savings can be made without adversely impacting on quality.	If implemented in the right way, this change could impact positively on service users as a) some service users would self-assess or be signposted to services with no requirement for an assessment; and b) people who 'self-fund' their care services would be able to access advice (particularly financial advice) and, also a 'brokerage' service that would enable them to choose the provider of their service in light of up to date, accurate information on value for money, quality etc.	
How saving to be achieved	In partnership with Sirona Care & Health further efficiency savings from the contract with 'Sirona' Care & Health. This would be in addition to the E9.0m savings already built into the five year contract between Sirona, the Council and the Primary Care Tust. A recently published Audit Commission report " <i>Reducing</i> <i>the cost of assessments and</i> <i>reviews</i> " based on 2010/11 benchmarking information, which pre- dates the establishment of Sirona, suggests that efficiencies from social care processes could be achieved in the medium term. Target is based on bringing BaNES costs closer to the national benchmark.	Delivery of the saving would need to be supported by: i) improved access to signosting, provision of advice tund information (including to self- funder); ii) policy and process redesign, including increases in self- assessment; and funder; iii) pathway redesign, culture change and skill-mix review.	
Impact on Assets and Property	None		ficiency Savings
Impact on staff	TBC (Primarily Sirona)		Sub Total – Other Cashable Effic
Risk to Delivery	×		Sub Total – Ot
2015-16 Saving £000	575		575
2014-15 Saving £000	575		575
2013-14 Saving £000		Page 100	50

2013-14 Saving £000	2014-15 Saving £000	2015-16 Saving £000	Risk to Delivery	Impact on staff	Impact on Assets and Property	How saving to be achieved	Impact to Service Delivery	Additional Information (Inc. PDSP Feedback)
20	o	0	L	None	None	Additional income from administration of Homesearch.	None. Additional income from Curo and other Registered Providers to fund advertising of social rented properties via the Homesearch Scheme.	
ଞ Pag	0 9		Σ		an N	The Fairer Contributions policy, which is based on national guidance and determines individuals' personal contribution to the costs of their community based personal care services. The policy requires that individuals are left with basic minimum income thresholds, which are nationally prescribed. Further protection is provided by a nationally prescribed 25% "buffer", which in B&NES is set above the required minimum at 30%. A very small amount of additional income could be generated by reducing this buffer back down to the statutory 25%.	Impact on the income of service users subject to the Fairer Contributions Policy, though these service users would continue to receive the income protection prescribed through national guidance. Some impact on commissioning and finance capacity to implement change.	
e 101	-1,000		L			Utilise s256 12/13 carry forward to delay recurring impact (one off). In line with Department of Health Guidance, it has been agreed by the Council and Primary Care Trust (to become the Clinical Commissioning Group (CCG) in April 2013) that a proportion of s256 funding can be utilised to offset demand-led pressures in adult social care purchasing budgets (including of Personal Budgets).		
200			L			Utilise s256 funding to meet pressures on adult social care purchasing budgets arising from demographic growth – particularly in placements, packages and Personal Budgets for older people and people with mental health needs, including dementia.		
1,580	-940	0	Sub Total – Additi	 Additional Income 				

2013-14 Saving £000	2014-15 Saving £000	2015-16 Saving £000	Risk to Delivery	Impact on staff	Impact on Assets and Property	How saving to be achieved	Impact to Service Delivery	Additional Information (Inc. PDSP Feedback)
o	51	0	т	1.5 FTE	None	£25k saving from ceasing the voluntary Accreditation Scheme for private rented accommodation.	We are changing our approach to ensuring quality standards in HMOs – this is currently being consulted on.	
						£26k saving from a reduction in staffing capacity in Housing Services.	The Accreditation Scheme provides landlords & tenants with reassurance that a property meets minimum standards. Proposed additional HMO licensing areas cover a significant proportion of the accreditation properties. – as a result, the voluntary scheme will be stopped.	
							Reduction in staffing capacity is likely to result in increased waiting times for some housing services.	
Page 102	375	455	т	None	None	A planned reduction of spend on purchasing the provision of personal care and support for older people, including those with dementia, adults with mental health needs, adults with learning difficulties and disabled adults, including those with sensory impairment.	Some service users and their families/carers view admission to residential or nursing care as the "safe" (low-risk) option. Our staff will work to ensure that any concerns about community-based alternatives about community-based alternatives are addressed effectively. In order to reduce such concerns and mitigate any risks, it would be critical to ensure strong, effective preventative and early intervention services, pathway redesign, and improved signposting and access (including to self-funders) to financial advice.	Proposals made in light of high- level review by PC, which suggests that savings could be achieved by further reductions in admissions to residential care.
						Primarily achieved by reducing admissions to residential care, particularly for older people, including those with dementia, by improving access to preventative and early intervention and also, by ensuring that signposting, access to universal services and advice to all, including self-funders, is effective. This saving aligns with investment plans to develop preventative services.	Further investment of Section 256 funding as well as a strategic shift in the investment of a proportion of Supporting People & Communities Funding would be appropriate in supporting the further development of this approach, which is in line with current national and local health and social care strategies.	IPC review makes it clear that these savings can only be realised if part of a strategic shift, including pathway redesign, improved access to preventative services and culture change – see also comment against savings in relation to assessment & care management.
							Proposal will increase pressures on Commissioning Team and will require culture change programme for practitioners.	
0	426	455	Sub Total	al – Reduced Service Levels	ce Levels			

Additional Information (Inc. PDSP Feedback)	Additional Information (Inc. PDSP Feedback)					
Additi	Additi		e z e	of r	o Id Alt	ž
Impact to Service Delivery	Impact to Service Delivery	Proposals represent a shift in the focus of Supporting People & Communities funding away from lower level support and towards delivery of more mainstream adult social care objectives.	There will be an impact on the people who currently use these specific services, such as older people, people who need support to enter or re-enter the workplace, people who ineed support to avoid/prevent homelessness, people who are socially excluded because of multiple/complex vulnerabilities such as mental ill health, disability, poverty, poor educational achievement & poor housing.	There will be an impact on a range of services which community organisations, as well as independent sector organisations, provide on our behalf.	However, as we continue to target our services towards more vulnerable people, there will still be an important part for the independent/ community sector to play in respect of delivering some of the £500k reinvestment in targeted advice/information; preventative services; and "pump- priming" third-sector organisations to recruit and support volunteers.	Managing the de-commissioning of services represents a significant challenge to commissioning capacity.
How saving to be achieved	How saving to be achieved	Over the coming years, the Council will focus the money it has available on care for the most vulnerable adults to support their independence.	As a result of this focus, there will be a reduction in the level of services which are not directly discharging defined statutory duies under Community Care legislation. Detailed proposals for 2014/15 will be worked up during 2013/14 and the estimated savings by "sector" set out below should, therefore, be treated with caution.	Detailed proposals to be worked up during 2013/14 will enable consideration of:	a) alignment with the Council's priorities;	 b) service performance, utilisation and value for money;
Impact on Assets and Property	Impact on Assets and Property	None				
Impact on staff	Impact on staff	None				
Risk to Delivery	Risk to Delivery	×				
2015-16 Saving £000	2015-16 Saving £000					
2014-15 Saving £000	2014-15 Saving £000	689				
2013-14 Saving £000	2013-14 Saving £000	152	P	age 103		

Impact to Service Delivery Additional Information (Inc. PDSP Feedback)	e to examine this ne various seek to bring 2013-14 if
	Officers will continue to examine this area of spend and the various contracts in place to seek to bring forward settings into 2013-14 if possible.
How saving to be achieved	c) engagement with providers, including views on how they might help deliver savings by for example working together more effectively to avoid duplication; d) the overall picture including how targeted investment is made to intigate the impact of delivering the savings and, indeed, help deliver the savings; and
ff and Property	‡⊕ <u>%%31</u> 40, ≱≰ <u>7,5</u> ,0
£000	
0003	
2013-14 Saving £000	

Additional Information (Inc. PDSP Feedback)	
Impact to Service Delivery	
How saving to be achieved	
Impact on Assets and Property	
Impact on staff	
Risk to Delivery	Î
2015-16 Saving £000	
ng 2014-15 Saving £000	
2013-14 Saving £000	Í

PLACE

Additional Information (Inc. PDSP Feedback)				Additional Information (Inc. PDSP Feedback)			
Impact to Service Delivery	Better customer service and lower processing costs.	Arising as part of other restructures.		Impact to Service Delivery	Pending further work. £212K of 2013/14 savings from D&MP.	Minimal	Introducing admission charges will require physical alterations; the reduction in building maintenance could impact on the quality of decorations and the fabric of the building.
How saving to be achieved	Customer Services Project within the Change Programme	P2P savings - consolidation of income functions across council		How saving to be achieved	All Directorate - Management Structure Changes across directorate, including DD level, 3rd tier and DMP division.	Transport Policy - Focus on transport planning and strategy.	Heritage Services - Victoria Art Gallery - reduction in net subsidy of £50k per annum, including the introduction of admission charging in 2013/14, together with a reduction in planned building maintenance; the operating model for the museum will be fundamentally reviewed in order to achieve a similar level of savings in subsequent years
Impact on Assets and Property			ne Savings	Impact on Assets and Property			
Impact on staff	Up to 19.5 ftes	Up to 3.5 ftes	Sub Total – Change Programme Savings	Impact on staff	Estimated at 8 ftes	Up to 2 ftes	Introducing admission charges will require greater use of volunteers and changes to staff duties required ; the fundamental review of the operating model in years 2 and 3 could impact further on staff numbers
Risk to Delivery			Sub Total –	Risk to Delivery			
2015-16 Saving £000	99	0	66	2015-16 Saving £000	300	0	20
2014-15 Saving £000	260	08	290	2014-15 Saving £000	300	0	20
2013-14 Saving £000	274	29	303	2013-14 Saving £000	86	se Page	ദ 105

y Additional Information (Inc. PDSP Feedback)	seum dating rating of ons e yet e yet		delays pport s on ding.	luding rvice y
Impact to Service Delivery	There is a risk of a reduction in footfall. The introduction of museum admission charges typically leads to drop in footfall of between 50 and 75%. The measure will render the Gallery less accessible to seniors and 'hard to reach' groups; the fundamental review of the operating model for the museum is likely to significantly change the offer available to visitors. The introduction of charging will mean a change to the way exhibitions are sourced, requiring greater emphasis on finding crowd-pulling exhibitions in addition to exhibitions where works are for sale ; the review of the operating model could have more significant impacts that are yet to be determined.	Minimal	Reduction in capacity, causing delays to delivery of work. Reduced support to Members and Parish Councils on traffic management requests. Reduction in capability to respond to prolonged emergency, e.g. flooding. Some equalities issues as traffic management supports minority groups, including people with disabilities.	Improved offer to customers including taxis and reduction in overall service costs. Improvements enabled by recent move of premises to Locksbrook Road
How saving to be achieved		Transportation Overheads - Seek to reduce management costs - joint review with DD Environmental Services	Highways (reduced management costs through structure changes) - potential to further consolidate and reduce service provision. This is the net saving after existing and emerging pressures absorbed.	Fleet Workshops - increase MOT / servicing income through reorganising service schedules and reliability to release capacity. Generate additional income through releasing in-house Passenger Transport capacity and reducing
Impact on Assets and Property				
Impact on staff		Up to 1 fte	Up to 7 ftes	Reduced overtime
Risk to Delivery				
2015-16 Saving £000		0	0	0
2014-15 Saving £000		0	0	0
2013-14 Saving £000	Pa	_ଷ age 106	ő	50

Additional Information (Inc. PDSP Feedback)					
Impact to Service Delivery	Contract extension likely to be required in order to secure a rate freeze. Failure to achieve a rate freeze may result in reductions to statutory services (with impacts on assets). Future opportunity to repackage services and include revised working methods in retendering. Risk that any new tender may bring in higher rates, creating further budget pressures.	Lease and / or licence agreements to be put in place; no proposal to transfer assets. Externalisation of TLC services into a Trust needs to be done against the background of achieving financial benefit. It does have the potential to generate a net annual saving of c. £100k. £200k. One-off set up costs are likely to be in excess of £500k. These calculations are heavily dependent upon a large number of assumptions.	Increased visitor numbers are likely to cause increased attrition on the site. Deterioration to the ancient monument cannot be reversed and therefore increased measures are needed to prevent this.	Taking into account our investment in the Roman Baths over the coming years, this is an excellent opportunity to improve the reputation of Bath to a wider number of people.	Stops use of B&NES Recycling Centres by non- residents and mirrors approach in neighbouring authorities Potential for shorter queues with nobody from outside the area using our facilities. Potential for increased fly tipping by people we no longer serve.
How saving to be achieved	Highways Maintenance - potential to achieve further contract savings	Delivery Trust option for Parks, Heritage Services, Libraries, Arts, Festivals - Review potential to establish a Trust to manage and deliver a range of services and functions including Heritage, Parks etc.	Heritage Services - Additional net profit over three years in line with the Heritage Services Business Plan.	We will improve our offer through the Roman Baths Development Phase 2 which has £4.188 million of capital allocated.	Waste - Introduce residents only permits at RC sites (using existing ANPR)
Impact on Assets and Property					
Impact on staff	Ē	All staff would be covered by TUPE arrangements	The business plan provides for a sustained increase in visitor numbers to the Roman Baths, which will impact on staff workloads		Ē
Risk to Delivery					
2015-16 Saving £000	o	O	200		0
2014-15 Saving £000	O	200	200		0
2013-14 Saving £000	20	⊳ Page 1	07		09

Additional Information (Inc. PDSP Feedback)							Additional Information (Inc. PDSP Feedback)	
Impact to Service Delivery	Process efficiencies, including potential for reducing overall costs of service management, therefore no significant impact on users.	Redevelop facilities at Old Welton & install new accommodation. Releases Radstock Road Depot. Removes potential to use Old Welton as transfer station contingency site = risk. Risk that full saving not achieved after all risks and costs are factored in	Improved customer service, increased focus and hours of enforcement. Reduced net cost through savings and income. More "localised" approach	Reduction of duplication but also capacity. Improved focus on developer funding.	Should improve fluidity of resources across boundaries. Net saving after supervision costs		Impact to Service Delivery	Increases in a number of areas proposed. A majority would require changes to Traffic Regulation Orders. Some charging proposals likely to be opposed by local residents resulting in delays to implementation. Significant risk of reduction in numbers using affected car parks, renewing permits or visiting affected locations.
How saving to be achieved	Reduce Transport Services management, supervision and admin teams through integration of teams and reduction of processing	Radstock Rd Depot to be closed and put portacabin facilities on Old Welton for cleansing / grounds staff	Reorganise Licensing and Environmental street-based inspection and enforcement staff into multi-functional teams which focus on geographical areas and are deployed to reflect varying levels of demand.	Restructure Estates Management / Play function	Reduce Neighbourhood Managers by combining cleansing areas		How saving to be achieved	Increase / introduce Car Parking Charges
Impact on Assets and Property						ency Savings	Impact on Assets and Property	
Impact on staff	Up to 2 ftes	ī	Up to 1 fte	1 fte	1 fte	Sub Total – Other Cashable Efficiency Savings	Impact on staff	Ē
Risk to Delivery						Sub Total – Otl	Risk to Delivery	
2015-16 Saving £000	0	O	o	0	0	550	2015-16 Saving £000	o
2014-15 Saving £000	0	0	20	0	0	770	2014-15 Saving £000	O
2013-14 Saving £000	02	20	40	₀ Page 1	50 50	822	2013-14 Saving £000	250

2014-15 2000 E0000	2015-16 Saving £000	Risk to Delivery	Impact on staff	Impact on Assets and Property	How saving to be achieved	Impact to Service Delivery	Additional Information (Inc. PDSP Feedback)
15	40		nil		Additional advertising income (13/14 = roundabouts, 14/15 = billboards & 15/16 = bus shelters)	Risks of lack of market interests and possible planning constraints due to National and Council Planning Policy. Income targets to be reviewed.	
0	0		nil		Planning - additional income from Government fee increase (note - this is one side of the net £700k pressure with £800k growth to remove base budget assumption on full cost recovery)	None - based on 15% extra allowed. Subject to government changing the fee schedule. Requires government decision.	
o	o		ni		Inflation on external income across Environmental Services (assumed at 2% for every income stream that is not a) set by statute, b) subject to a separate decision c) internal).	Risk that total income may not increase as demand is reduced due to increases in charges	
0	0		nil		Review of Bus Subsidies	Results from re-tender of existing subsidised bus contracts	
60	0		ni		Waste - Reduce subsidy for garden waste fees per bin in line with neighbouring authorities	Minimal	
75	40	Sub Total – Additional Income	onal Income				
2014-15 Saving £000	2015-16 Saving £000	Risk to Delivery	Impact on staff	Impact on Assets and Property	How saving to be achieved	Impact to Service Delivery	Additional Information (Inc. PDSP Feedback)
255	0		Up to 10 ftes			Involves non replacement of one mobile library (estimated cost of replacement c.£90,000)	
					Libraries & Information - To prioritise the retention of our eight Library branches we propose to deliver the service differently in the future with support from our emerging Community Library Programme.	Fewer visits and for less time at stop- offs by the one remaining Mobile Library – this will be offset by the development of our Community Library Programme to provide services in communities without a library branch; restructuring of service; reduction in materials, and some PC's and services. Consultation last year indicated how strongly people felt about provision of library branches and willingness to be involved in shaping their local library service	

2013-14 Saving £000	2014-15 Saving £000	2015-16 Saving £000	Risk to Delivery	Impact on staff	Impact on Assets and Property	How saving to be achieved	Impact to Service Delivery	Additional Information (Inc. PDSP Feedback)
200	200	500		0.5 fte		Destination Management - Reduction in subsidy for significant elements of activity, combined with increased Spa income and reduced water monitoring costs	Includes continued growth in Spa income, combined with reduced water monitoring costs; significant reduction in support for Destination Marketing phased over years 2 & 3 alongside action to increase external finance; reduced project budgets for World Heritage site management; deletion of residual budget for support to Future Bath Plus. Managed reduction in Council direct subsidy for destination marketing activity alongside action to increase external finance	
୍ଜୁ Page 110	o	o		up to 5.5 ftes		Restructure / reduction in service offer. Services will be at statutory minimum. Likely cessation of business support model with reduction in informal approach to securing compliance at a time of increasing business numbers. Reduction in enforcement. Cessati A routine sampling. Reduction in Public Protection (Food Safety and Standards, Environment al Monitoring at routine sampling. Reduction in partnership working. Possible dela services, review elements for delivery with a view to minimising service request in esponse times. Reputational issues - service reductions - failure to deliver statutory services. Reputational issues - service requiring project work support change in working method and consultation / redundancy processes.	Restructure / reduction in service offer. Services will be at statutory minimum. Likely cessation of business support model with reduction in informal approach to securing compliance at a time of increasing business numbers. Reduction in enforcement. Cessation of routine sampling. Reduction in partnership working. Possible delays in response to corporate deadlines and complaint/service request response times. Reputational issues - service eductions - failure to deliver statutory services. Reputational issues - service eductions - failure to meet event change and develoment requiring project work to support change in working methods and consultation / redundancy processes.	

Additional Information (Inc. PDSP Feedback)			
Impact to Service Delivery	Revised pest control service to move from Radstock Rd depot and free up space (store relocated to RVP). No requirement for biomass boiler in RVP nursery. Saving in RVP energy consumption (gas) and therefore reduction in carbon. Significant change in Pest Control Service with reduced staffing and offer focussed around providing means-tested service for people in need (Year 1). Parks management streamlined to improve coordination, efficiency and effectiveness and reduce costs (Year 1). RVP Nursery will not close but savings made through stopping costs of heating area through stopping costs of heating and reducing volume of planting and reducing volume of planting and reducing volume of planting and reducing volume of planting and reducing staff within Parks service standards (Year 1 onwards). Significant organisational change and development to achieve proposed savings whilst ensuring service continuity.	Assumes reduction in Planning Policy work once these workstreams are complete. Rated as M rather than L because political priorities keep changing	15 public conveniences to be closed and re-invest some of the proceeds to improve the remainder of Council- run toilets. This will allow the public convenience budget to focus on the remainder to a higher standard than people currently experience.
How saving to be achieved	Neighbourhoods Services - reduce Environmental Protection services, cease in-house plant production, reduce parks maintenance and planting.	Planning Policy & Environment teams - Reduction in level of spending within Planning Policy and Environment teams to be achieved after the major elements of Planning Policy are completed, including core strategy, ypsy & travellers DPD, Placemaking and CIL	
Impact on Assets and Property			
Impact on staff	1 fte plus some redeployment anticipated within service area	up to 3 ftes	up to 2 ftes and agency staff
Risk to Delivery			
2015-16 Saving £000	O	0	o
2014-15 Saving £000	ξ	O	120
2013-14 Saving £000	∝ Page 111	0	o

Additional Information (Inc. PDSP Feedback)		
Impact to Service Delivery	Currently provide 29 public conveniences in total. Better facilities for residents, visitors, and workers. Fewer Council-run toilets will be offset by the existing provision of non- offset by the existing provision of non- centre, including SouthGate shopping centre and the new toilets agreed through the planning process at the Railway Vaults Maintenance of remaining facilities transferred to other organisations where possible. Service is non- statutory but affects residents, tourists, visitors, mobile workers. Possible impacts across the equalities range but elderly and disabled particularly depending on the location of the toilets we may sell. Proposa two und retain key Provision Strategy for Public Toilets (2011) to be reviewed/revised to viable alternative provision (in shops, cafes, bars, Council facilities) and build in higher cleansing standard at remaining sites. Mote: concurrent tendering exercise in progress. Savings assumed all from yr 2 but could be delayed. Service likely to stagnate and unable to improve if decision left in limbo.	Increased fly tipping; capacity issues at existing sites when open. Significant reduction in customer service
How saving to be achieved	Public Conveniences - Reprioritise our spend on public conveniences to improve quality standards and service to the public	Waste - 2 recycling centres to be closed 2.5 days each in mid-week.
Impact on Assets and Property		
Impact on staff		up to 5 ftes
Risk to Delivery		
2015-16 Saving £000		o
2014-15 Saving £000		o
2013-14 Saving £000	Page 112	50

Additional Information (Inc. PDSP Feedback)	ر ل			Additional Information (Inc. PDSP Feedback)	= 0		be "De			Additional Information (Inc. DDCD	Additional Information (Inc. PUSP Feedback)			
Impact to Service Delivery	Possible asset transfer. Move away from dependency on Council managed sites. Risks of dispute and of low take-up	Possible asset transfer. There is currently over-provision of bowls facilities in Bath. Can offer facilities elsewhere		Impact to Service Delivery	No direct impact. Highly successful events will not take place (Sky Ride and Wheels for All); at least one voluntary organisation would be at risk if grants are eliminated; Less funds available to finance one off issues within Leisure. Proposals impact across the Sports Development and Active Leisure service	Reputational risk through failure to properly discharge statutory duty - atthough private sector providers of pest control are available and some services could be recommended by us. Statutory obligations remain	Increased complaints. Save staff and fuel costs. Most bins are not "missed" due to service errors	Increased complaints due to changes. Recycling collections not affected. Net saving through reduction in premium rates / extra time off.			Impact to Service Delivery	subject to announcement by end October 2012		
How saving to be achieved	Transfer allotment management to user groups (i.e. provide them but do not manage them)	Stop maintaining bowls at Sydney Gardens		How saving to be achieved	Sports Development & Active Lifestyles: No longer provide Sky Ride and Wheels for All events; Deletion of grants to voluntary organisations; Reduce Performance Swimming Budget; Remove contingency budget.	Cease pest control service (extra reduction beyond that proposed above)	Waste - stop collecting "missed bins" except where this is the fault of the service.	Waste - cease 5 bank holiday refuse collections except Xmas / New Year			How saving to be achieved	Weekly Collections Fund		
Impact on Assets and Property			e Levels	Impact on Assets and Property					services (1997)	Immact on Accate	impact on Assets and Property			
Impact on staff	nil	nil	Sub Total – Reduced Service Levels	Impact on staff	īz	2 ftes	possibly reduced overtime	possibly reduced overtime	Sub Total – Discontinued Services		Impact on staff	nil	Sub Total	TOTAL SAVINGS
Risk to Delivery			Sub Tota	Risk to Delivery					Sub Tot		Risk to Delivery			
2015-16 Saving £000	0	0	300	2015-16 Saving £000	O	O	0	O	0	2016_16 Caving	2015-16 Saving 2000	0	0	956
2014-15 Saving £000	20	10	636	2014-15 Saving £000	0	08	0	08	60	2011-15 Caving	2014-15 Saving £000	0	0	1,831
2013-14 Saving £000	0	0	758	2013-14 Saving £000	ق Page	9 113 ⁹	20	0	132	2012-11 Caving	2013-14 Saving £000	400	400	2,978

	Impact to Service Delivery Feedback)	This saving is the balance remaining	Delivery is dependent upon ICT transferring in-house on 1 August 2013 & then centralisation and rationalisation of all ICT spend from 1 April 2013.	Reduced spend on ICT will mean ICT will be provided strictly to meet priority business needs	Statfing impacts cannot be assessed until the service has returned in house.	The effective completion of this programme requires continued investment in the change from the relevant reserve and links to the various items in the capital programme.	Link to Sirona ICT strategy to be evaluated	£60K Change target relating directly to Resources procure to pay and £40K similar target for consolidation of income functions across the Council including from rationalising cash income system & systems, which support web based payments facility.	The effective completion of this programme requires continued investment in the change from the relevant reserve.	P2P will deliver further savings within Place (£92k) and People & Communities (£65k) which will require staff restructuring & consolidation within these
	Impact to \$	This saving is th	Delivery is dependent upon ICT transferring in-house on 1 Augu 2013 & then centralisation and rationalisation of all ICT spend f April 2013.	Reduced spend on ICT will me will be provided strictly to meet priority business needs	Staffing impacts until the service house.	The effective completion of this programme requires continued investment in the change from t relevant reserve and links to the various items in the capital programme.	Link to Sirona IC evaluated	260K Change target relating dir to Resources procure to pay ar 240K similar target for consolid. of income functions across the Council including from rationalis cash income system & systems which support web based paym facility.	The effective completion of this programme requires continued investment in the change from t relevant reserve.	P2P will deliver further savings Place (£92k) and People & Communities (£65k) which will require staff restructuring & consolidation within these Directorates
	How saving to be achieved	ICT - Corporate	Rationalisation of systems, improved procurement & implementation of new ICT Strategy plus consolidation of budgets.	This saving will apply across all departments and so is described as corporate even though will be led in the Resources Department.	New centralised or federated staffing model required			Finance	Business & Financial Services: P2P	Rationalising procure to pay (P2P) will deliver further savir and income systems in finance either using centralised or federated staffing model
	Impact on Assets and Property									
	Impact on staff							3 P2P	1 BFS	
	Risk to Delivery									
	2015-16 Saving £000	150								
	2014-15 Saving £000	500						27		
RESOURCES	2013-14 Saving £000	250				Page 114		٤2		

2013-14 Saving £000	2014-15 Saving £000	2015-16 Saving £000	Risk to Delivery	Impact on staff	Impact on Assets and Property	How saving to be achieved	Impact to Service Delivery	Additional Information (Inc. PDSP Feedback)
		170		σ	-	Customer Services	This is a year 3 savings from simply stopping a service development function on the basis it will have become business as usual - The Business Development team in Customer Services supports the customer services change programme – removing this team will mean the programme comes to an end in 15/16 – the schedule suggests a one off source of resources in 15/16 may be needed to sustain the programme through that year and then stop.	
						Business Development		
						Stopping the business development service once the new way of working has been fully rolled out in 2015/16.		
		100		ى ب		HR & Payroll – delivery model	New model for HR/Payroll service delivery agreed, consistent with future function and size of the Council. Currently subject to finalising the business case.	
						New methods with less date inputting and more self service.	Savings are dependent upon insourcing of transactional and payroll functions from Mouchel and new software platform to support manager / employee self-service.	
							Full implementation will take 2 years. Benefits will also include accurate & up-to-date management information.	
250	250					Management Structure	Consultation on a new management structure will start at the beginning of 2013.	
					_ **	Resources Department management structure to be reviewed	Following agreement of principles a more streamlined structure will be phased in.	
							Staff impact to be assessed. Savings include related support.	
100	100					ICT	Relates to assumed overheads of current contract	
						In-sourcing of ICT function	Staff impact to be assessed.	

Additional Information (Inc. PDSP Feedback)										Additional Information (Inc. PDSP Feedback)		
Impact to Service Delivery	To be delivered as per Release 1 of customer services change plan.	Face to face customer service options retained but self serve and web improved. Also affects e billing and document management,	New open systems Drupal web site to require less resources to maintain.	No further staff impact expected	Some mitigation may be achieved by splitting the delivery and commissioning roles and combining some of delivery with project delivery elsewhere in the Council.	A less responsive service will result including ability to take forward development opportunities unless one off development budgets can be identified. The rolling development fund helps with this but will need to be topped up from time to time.	Business to be run at scaled back level to test viability of this approach.	For more specialist work a new framework contract procured through the procurement partnership will be available.		Impact to Service Delivery	Gradual removal of all project budgets much of which is spent on lean systems thinking reviews, project support to Council services and ICT projects. Replacement with one off funding from reserves as required.	
How saving to be achieved	Customer Services	New standardised and simplified customer contact arrangements	Improvement & Performance	Web site administration efficiencies	Property	Property Services - Reduction in staffing levels to facilitate 20% reduction target	Property	Print procurement and delivery efficiencies.		How saving to be achieved	Transformation	Removal of recurring budget
Impact on Assets and Property									ne Savings	Impact on Assets and Property		
Impact on staff	9				50		ε		Sub Total – Change Programme Savings	Impact on staff		
Risk to Delivery									Sub Total -	Risk to Delivery		
2015-16 Saving £000	21						20		457	2015-16 Saving £000		
2014-15 Saving £000	9		44		230		20		1177	2014-15 Saving £000	200	
2013-14 Saving £000	155		21		377	Page 116	10		1236	2013-14 Saving £000	100	

Additional Information (Inc. PDSP Feedback)									
Impact to Service Delivery	The Corporate Travel Plan together with the Workplaces project are intended to support changes in the business travel practices of staff reducing journeys and carbon emissions.	Managers are currently assessing the impact on service delivery although no adverse impact is anticipated based on experience in other authorities.	The saving arises from introducing HMRC travel rates and associated changes to mileage allowances but is subject to consultation.		Review to primarily affect democratic services, policy and partnerships plus improvement and performance to integrate functions and reduce systems and processes.	Staff impacts unknown at this stage but assumption made based on level of saving	There may be some corporate aspects to this review and potential links to the Resources management review.	There are distributed arrangements in place that could be combined to achieve savings from the management of capital schemes and project delivery.	A spilt of some commissioning roles from project delivery and support is also appropriate in aspects of property and project delivery.
How saving to be achieved	Policy & Partnerships	Travel Plan – Corporate	Review of business travel & reimbursement arrangements	This saving will apply across all departments and so is described as corporate even though will be led in the Resources Department.	Policy, Performance & Administration in Resources Department	Consolidation of support areas across Resources including democratic meetings support, aspects of information management, performance management and reporting.		Property	Consolidation of capital programme support, property & project management.
Impact on Assets and Property		·							
Impact on staff									
Risk to Delivery									
2015-16 Saving £000									
2014-15 Saving £000					200				
2013-14 Saving £000	500				Page 117				

Additional Information (Inc. PDSP Feedback)										
Impact to Service Delivery	This will potentially link with the Resources management review and in addition there will be corporate aspects to this review. Staff impact to be assessed.	Similar to the above but in respect of commissioning and procurement support. Scope to centralise or federate staff plus potential links to the Resources management review.	Staff impact to be assessed.	Similar to the above but in respect of ICT & Information Governance roles following in sourcing of IT contract. Potential links to the Resources management review	Staff impact to be assessed.		Reductions in corporate estate budgets to deal with one off work. No direct staff impact.	The main savings relate to workplaces and have already been accounted for as part of that project.	Separate savings relate to R&M for which see below.	A 15% cut in legal support (split over 2 years) that will require clearer prioritisation of support based on impact and risk as assessed by the Council on a corporate basis. Any substantial legal challenges will need to be funded from project budgets or Council central contingency reserves.
How saving to be achieved		Audit Risk & Assurance	Consolidation of Procurement & Commissioning Support across the Council to increase resilience and sustain acceptable level of capacity and skills.	Audit Risk & Assurance	Information Governance	Rationalisation with ICT Service	Property Corporate Estate – efficiencies			Legal & Democratic
Impact on Assets and Property										
Impact on staff										4
Risk to Delivery										
2015-16 Saving £000							9			
2014-15 Saving £000		20		25			16			75
2013-14 Saving £000				Page	e 118	}	5			75

2014-15 Saving £000	2015-16 Saving £000	Risk to Delivery	Impact on staff	Impact on Assets and Property	How saving to be achieved	Impact to Service Delivery	Additional Information (Inc. PDSP Feedback)
					Reduced structure for Legal Services	Shared working on specialist aspects of the service with other authorities to be accelerated to see if further efficiencies can be found.	
	200		3		Finance	This level of savings will need to be measured carefully.	
					More savings in management of finance to reduce the service to a core service	As with legal and HR the risk is that the services will start to decline below acceptable levels and in this case qualification of accounts becomes a risk.	
						To mitigate this, a very effective risk based approach will be needed.	
						All other support services will be assessed to ensure no one service is disproportionately affected in year 3. Metrics and benchmarking will be important.	
25					Property	A slightly reduced cleaning specification, which could start to impact noticeably on building cleanliness - so will be kept under review especially in busy areas and busy building with flexible office space. This represents a 20 % reduction.	
				-	Cleaning		
					Reduction in Cleaning spec across corporate offices		
					Resources	There is now no funding available to support 'one-off' change projects within Resources	
					Project Support Removal of Resources Change Project funding		

Additional Information (Inc. PDSP Feedback)								
Impact to Service Delivery	Option to fully remove the service has previously been considered and savings are now available as a result of absorbing workload within Customer Services team and co- location within One Stop Shop.			Review of long term capital financing and financing reserves.	Investment interest, Capital & debt inherent in the treasury management costs – Corporate financing arrangements.	No provision required for rising interest costs as loans will be at fixed rates but opportunity cost of cash flow funding may become more expensive over time which has not been allowed for.	The figure also includes £200,000 for carbon tax and this is no longer required as it has been possible to charge schools appropriately and the rates are slightly better than expected. A further £100,000 relates to saving in external audit fee following the abolition of the Audit Commission and appointment of Grant Thornton as external auditors. No direct staff or service impacts	Review of use of Council owned assets based in communities to see which ones might be better run by the communities themselves, also which should be consolidated and which sold off. Also affected by the new statutory 'right to bid'.
How saving to be achieved	Customer Services	Shop mobility	Maintain Shopmobility service through current Customer Services team and greater use of volunteers	Finance	Investment interest, Capital & debt costs – Corporate	This saving will apply across all departments and so is described as corporate even though will be led in the Resources Department		Policy & Partnerships + Property
Impact on Assets and Property								
Impact on staff								n/a
Risk to Delivery								
2015-16 Saving £000								500
2014-15 Saving £000								500
2013-14 Saving £000	۷			1,700		Page 120		

2014-15 Saving £000	2015-16 Saving £000	Risk to Delivery	Impact on staff	Impact on Assets and Property	How saving to be achieved	Impact to Service Delivery	Additional Information (Inc. PDSP Feedback)
					Community Use of Assets – Corporate	There is a risk that aspects of this saving might be double counted with savings elsewhere in medium term plans but this will not be evident until the review progresses.	
					This saving will apply across all departments and so is described as corporate even though will be led in the Resources Department	The review does not include the Councils main offices, which have been reviewed and are being rationalised as part of the workplaces programme. The savings from this have already been accounted for.	
1,091	706	Sub Total – Oth	Sub Total – Other Cashable Efficiency	ency Savings			
2014-15 Saving £000	2015-16 Saving £000	Risk to Delivery	Impact on staff	Impact on Assets and Property	How saving to be achieved	Impact to Service Delivery	Additional Information (Inc. PDSP Feedback)
250	200		n/a		Property	This arises from the potential to invest in properties with secure returns that exceed those available through usual treasury management opportunities.	
					Commercial Estate	The acquisitions need to be linked to the enhancement of the Council's estate and to therefore achieve wider property objectives.	
					Increased revenue income through investment in Commercial properties.	One example of such an acquisition has already been achieved in the current financial year and represents an early move towards this target. A and support for regeneration are possible additional benefits.	
					The Council as a whole benefits from the income in this estate, which presently is a net £13M.	Over the next three years, the capital borrowed to fund the acquisitions is intended to be repaid using new capital receipts to be generated from lease restructuring which has no significant impact upon existing revenue streams.	

Additional Information (Inc. PDSP Feedback)											Additional Information (Inc. PDSP Feedback)	
Impact to Service Delivery	Income to be generated through increased advertising using the Council's web site and other means such as bus stops, traffic islands, and events.			There is potential to introduce some form of tourism levy so that the Council's day-to-day investment in tourism and visitors can continue to be supported.	Any such levy would be subject to widespread consultation and appropriate due diligence.		The Council has a number of trading partners and the aspiration is to review those relationships to assess the potential to increase Council income.	These reviews will initially need to be treated as commercially sensitive.			Impact to Service Delivery	The new local Council Tax Support scheme (formerly Council Tax benefits) is being introduced in a way that will offset the 20% reduction in Government funding for working age claimants. Pensioners are protected.
How saving to be achieved	Improvement & Performance - corporate	Additional Advertising income	This saving will apply across all departments and so is described as corporate even though will be led in the Resources Department	Finance - corporate	Tourism Levy	This saving is described as corporate and will be led in the Resources Department	Finance – corporate	Trading Opportunities	This saving will affect a number of departments, is described as corporate and will be led in the Resources Department		How saving to be achieved	Customer Services
Impact on Assets and Property										Icome	Impact on Assets and Property	
Impact on staff	n/a			'n/a						Sub Total – Additional Income	Impact on staff	
Risk to Delivery										Sub T	Risk to Delivery	
2015-16 Saving £000	62			500			250			1,012	2015-16 Saving £000	
2014-15 Saving £000	29			200			200			1,279	2014-15 Saving £000	
2013-14 Saving £000	19					Paç	je 122			569	2013-14 Saving £000	

Additional Information (Inc. PDSP Feedback)								
Impact to Service Delivery	This major change therefore shows as a nil net cost.	Dues to the timetable for billing and tax base setting this change is to be considered by the November meeting of Council and in advance of budget setting.	To try to avoid the reduction in access to face-to-face advice the Council is participating in as a pilot for the Department of Work & Pensions (DWP) and attracting funding for that purpose.	The new system will be gradually introduced between October 2013 and 2017.	The Council will cease to provide the service as the caseload migrates to the DWP but may be able to provide a local front office service if the pilot is successful.	The financial arrangements for such a service are not yet clear and there will be implications for the voluritary sector who already provide advice claimants as well as money advice.	A nil net cost has been assumed. There are about 10-15 staff that work in this area (HB) and their roles will potentially transfer or disappear over time, subject to the above.	HB Admin Government grant equates to the cost of running the existing service. The benefits are currently funded by Government subsidy. This relates to complex subsidy rules and certain risk that remain with the Council. It is assumed that this will still be needed in future when Universal Credits are implemented as a means of funding a front office service.
How saving to be achieved	Council Tax Support Scheme		Customer Services	Housing Benefits to Universal Credits				
Impact on Assets and Property								
Impact on staff								
Risk to Delivery								
2015-16 Saving £000								
2014-15 Saving £000								
2013-14 Saving £000					Pag	e 123		

Additional Information (Inc. PDSP Feedback)					
Impact to Service Delivery	Savings relate to	 Removal of dedicated budgets for local initiatives and events. Currently £19k used to complement local projects, including Christmas events and some co-ordination of local partnership activity. Budget has been reduced over the years and communities have been assisted in obtaining funding from other sources such as Ward Councillors Initiative 	2 Savings from Christmas lighting budget in first year leading to ending of this budget at end of current contract. Budget currently £81k will reduce by £11k in year one, Year 2 and 3 savings assume that we have been able to generate income to fund any Christmas lights and or the level of provision has been reduced or efficiencies made	3 Saving from community safety budgets which will be achieved through continuing the change programme the Partnership began last year in the light of the advent of PCCs. Saving proposed £30k from a £40k budget set up in the 2012/13 budget to provide some flexibility in the transitional period	These savings coupled with savings delivered in 2012/3 leaves the service commissioning in 2013/14, local Heatthwarch (new statutory responsibility) £72k, the reducing Christmas lights budget £70k, a contract on Voluntary and Community Sector support 265k, Street Marshalls work with the BID £21k, the Councils contribution to the Student Community Partnership £18k and some capacity to support some work with Equalities groups £40k
How saving to be achieved	Policy & Partnerships	Partnership Delivery	Commissioned services and local events budgets		
Impact on Assets and Property					
Impact on staff					
Risk to Delivery					
2015-16 Saving £000	45				
2014-15 Saving £000	20				
2013-14 Saving £000	60		Page	124	

Additional Information (Inc. PDSP Feedback)								
Impact to Service Delivery		The savings will be achieved through a series of efficiency measures within Democratic Services. The level of Overview and Scrutiny and democratic accountability will be protected	A significant reduction in the budget of about 20%, which will mean that the already highly prioritised approach will be restricted further. Condition of assets will need to be monitored. Rationalisation and renewal of corporate estate, especially. offices will help.					
How saving to be achieved	Legal & Democratic	Efficiency measures within Democratic Services	Property	Repairs and Maintenance	Reductions in R&M budgets			
Impact on Assets and Property								
Impact on staff	1					ed Service Levels		
Risk to Delivery						Sub Total – Reduced Service Levels	TOTAL SAVINGS	
2015-16 Saving £000						45	2,220	
2014-15 Saving £000			83			105	3,652	
2013-14 Saving £000	35		344		F	D 439	ር መ 4,780	125

Additional Information (Inc. PDSP Feedback)	
Impact to Service Delivery	
How saving to be achieved	
Impact on Assets and Property	
Impact on staff	
Risk to Delivery	
2015-16 Saving £000	
013-14 Saving 2014-15 Saving £000 £000	
2013-14 Saving £000	

REGENERATION, SKILLS & MAJOR PROJECTS

2013-14 Saving £000	2013-14 Saving 2014-15 Saving 2015-16 Saving 2000 2000	2015-16 Saving £000	Risk to Delivery Impact on staff	Impact on staff	Impact on Assets and Property	How saving to be achieved	Impact to Service Delivery	Additional Information (Inc. PDSP Feedback)
212	0	0	-	3 posts to be deleted	auou	All Directorate - Management Structure Changes across directorate, including DD level, 3rd tier and DMP division.	No impact as saving achieved by change in Strategic reporting and management input in line with Council policy of moving to 3 Directorates.	
212	0	0	Sub Total – Oth	Sub Total – Other Cashable Effici	fficiency Savings			
212	0	0	TOTAL SAVINGS					

CORPORATE

2013-14 Saving £000	2014-15 Saving £000	2013-14 Saving 2014-15 Saving 2015-16 Saving 2000 2000	Risk to Delivery Impact on staff	Impact on staff	Impact on Assets and Property	How saving to be achieved	Impact to Service Delivery	Additional Information (Inc. PDSP Feedback)
Page 1	o	0	L	IN	none	Removal of remaining budget set aside to cover reductions resulting from central educational services funding changes (LACSEG)	No impact	
2 170	0	0	Sub Total – Other Cashable Efficiency Savings	Cashable Efficienc	:y Savings			
0 759	200	200	L	IIN	none	New Homes Bonus grant income	No impact	
759	200	200	Sub Total – Additional Income	onal Income				
929	200	200	TOTAL SAVINGS					
11,630	8,428	7,289	OVERALL SAVINGS	ŝ				

COMMUNITY ASSETS TRANSFER PROGRAMME

1 CONTEXT

- 1.1 The challenging financial situation increasingly requires the Council to use to the full the assets available to help communities tackle the local issues of importance to them.
- 1.2 Since the publication of the Quirk report in 2007, councils and central government have seen greater community involvement in community assets and buildings as potentially delivering some or all of the following benefits:
 - (1) Better community outcomes- by releasing an organisation from its liabilities to the Council it can grow its work in the community. The Quirk Review noted that the crucial driver for asset transfer is not the disposal of assets per se but the empowerment of communities; any approach to community assets must therefore be seen in the context of the Council's aim to be an open and engaging Council that empowers communities and works in partnership with a range of public, private and voluntary organisations.
 - (2) **Better Value for Money** cutting costs through new, localised ways of working and involvement of volunteers as well as drawing in more resources, for example from parish councils.
 - (3) **Greater support for our voluntary and community sector** with opportunities for long-term planning by the community organisation involved in the building, including the ability to draw down external investment as well as greater freedom to tailor services to need. This support for local community organisations is urgent in the current climate and can make a real difference to long-run viability.
- 1.3 The new Community Right to Bid provides an additional focus for involvement in community assets, including those not in Council ownership.

2 COMMUNITY ASSETS IN BATH AND NORTH EAST SOMERSET: SUGGESTED APPROACH

2.1 It is proposed that a programme begin of transferring Council community based assets and services to voluntary groups and parish councils, beginning with a series of "Quick Wins" alongside the development of a more detailed framework and criteria. "Transfer" is defined as "passing ownership or management of a building or piece of land from a public sector body to a third sector organisation". This may or may not involve transfer of the freehold and the most appropriate arrangements will be put in place depending on the nature of the asset and organisation concerned and the outcomes to be achieved.

3 COMMUNITY ASSET TRANSFER- "QUICK WINS"

- 3.1 To make progress on this issue, the Council will initially focus on finalising proposals for asset transfer where discussions are already well- advanced. These are set out in Table 1(below) and reflect the diversity of community organisations delivering services in our area, both in terms of structure and of the types of outcomes being achieved. However, each of these schemes has been built around:
 - (1) An appropriate, not-for-profit legal entity, and governance structure to ensure effective management of the asset
 - (2) Alignment between the work and objects of the organisations and the Council's own Corporate Plan objectives.
 - (3) Opportunities for enhancing the capacity of local voluntary and community organisations and empowering them to deliver their objects
- 3.2 As asset transfer terms are finalised with these organisations, the Council will also put in place arrangements to ensure that the wider community interest is safeguarded and that the overall objectives agreed will continue to be delivered in the longer term. The Council will consider impacts on its overall priority of enabling groups to grow their services in the community as well as the overall financial implications of proposals. The learning from this work will be used to develop the wider criteria for Community Asset Transfer set out in 4 below.

4 DEVELOPING CRITERIA FOR ASSET TRANSFER

- 4.1 As well as progressing the "Quick Wins" identified in Table 1, the Council wishes to develop a wider offer to voluntary and community groups and parish councils in relation to community assets. A report will be brought to Cabinet recommending the process and criteria for this, building on experience from the work on the "Quick Wins". It is expected that this report will be built around the following key principles
 - (1) **Listening-** an open and transparent invitation to voluntary/community organisations and parish councils to propose changes to arrangements relating to community assets
 - (2) **Prioritising** being clear about the types of assets which will be considered for transfer
 - (3) Joining-Up- checking the impact of proposals on other plans for assets, bringing services together to reduce costs and duplication where possible
 - (4) **Working With You** engaging constructively on proposals and encouraging phased approaches where practical

- (5) Sharing Ideas- learning from successful schemes
- 4.2 In response to requests arising from this process, the Council will make positive suggestions including the potential for a phased approach and providing appropriate support to local groups.
- 4.3 Further work will be undertaken on more detailed criteria but the tests to be adopted will be proportionate to the degree of change in the relationship being proposed and the benefits to the community. The criteria will include appropriate safeguards to ensure that any asset is used in accordance with the objects agreed in any leasing arrangement, to ensure effective governance and to address any financial and value for money implications.
- 4.4 The Resources Policy Development and Scrutiny Panel intends to undertake a review of the Council's approach to community assets and a report is being presented to its meeting of March 18th. This will be linked to the proposals for policy development set out in this Appendix.

5 LINK WITH MEDIUM TERM SERVICE AND RESOURCE PLANNING

- 5.1 The Council and its partners hold a significant numbers of assets across our area. It is proposed that further work be undertaken to identify where costs can be reduced by removing duplication, for example, through colocation of services. The Council has in its Resources MTSRP a proposal to save £500k in 14/15 and an additional £500k in 15/16 through reviewing the use of Council owned assets. An analysis of 90 Council community assets suggests a total running cost of £7.5m, although since this includes service assets such as library and leisure facilities it is important to avoid double-counting.
- 5.2 Again, this work will be linked to the work being undertaken by the Resources Policy Development and Scrutiny Panel.

Table 1 - "Quick Wins"

NOTE: All proposals below will contribute to the following outcomes through the process of asset transfer:

- Where local people actively lead the delivery of improvements in their community
- Where decisions are made as locally as possible
- Where there is easy access to public services and local amenities
- People have developed their skills and used them to improve their community

Asset	Summary of outcomes provided by the organisation	Corporate Plan Outcomes directly delivered
Percy Community Centre – New King Street, Bath	A well-maintained, accessible community resource for other voluntary sector and community groups and to support the economic activities of individuals with a range of services. The Centre now has over 50,000 users a year and hosts a regular schedule of around 25 different activities each week.	 The people most in need are supported to live full active lives Children and young people enjoy their childhood and are prepared for adult life There are activities and opportunities to help young people to make a positive difference to their lives and communities Everyone has the opportunity to participate in sports, leisure and cultural activities Reduced inequality between communities across Bath & North East Somerset
Chapel Arts Centre, St James Memorial Hall, Bath	A community & culturally based arts centre presenting a wide range of the performing arts, including music), theatre, dance, comedy, film, mime, poetry and multimedia	 Everyone has the opportunity to participate in sports, leisure and cultural activities
Whisty Community Centre, Stoneable Rd	The Whisty community Association provides community facilities and services tailored to meet local needs in Whitelands Springfield and Tyning. It works with statutory authorities, schools, colleges and voluntary organisations to improve the quality of life of the inhabitants of the WHISTY area. Services available include a pre-school playgroup and parent toddler group.	 The people most in need are supported to live full active lives. Children and young people enjoy their childhood and are prepared for adult life Everyone has the opportunity to participate in sports, leisure and cultural activities Reduced inequality between communities across Bath & North East Somerset

Fairfield House, Bath	The Friends of Haile Selassie I Fairfield House has been established with clear objects relating to respecting and preserving its heritage and its legacy for the older people of Bath as for different community groups intended by HIM Haile Selassie 1 bequest as well as developing Fairfield House as a sacred space. It is in the process of preparing an appropriate legal entity and Business Plan.	 The people most in need are supported to live full active lives. Older people are supported to live independently Everyone has the opportunity to participate in sports, leisure and cultural activities
Midsomer Norton Railway Station	The Somerset & Dorset Railway Heritage Trust operates a heritage railway from its station at Midsomer Norton and is currently extending its operation.	 There are activities and opportunities to help young people to make a positive difference to their lives and communities. Everyone has the opportunity to participate in sports, leisure and cultural activities
YMCA Broad St, Bath	Bath YMCA is an autonomous charity that affiliates to YMCA England. It has its own board of management made up of representatives from local communities. As well as accommodation, the organisation provides a wide range of other services including breakfast clubs, after school clubs, childcare and youth projects.	 The people most in need are supported to live full active lives Children and young people enjoy their childhood and are prepared for adult life There are activities and opportunities to help young people to make a positive difference to their lives and communities Everyone has the opportunity to participate in sports, leisure and cultural activities
Genesis Trust, James St West, Bath	A charity in Bath that currently runs a range of to help the homeless and vulnerable in the city. These include a furniture project, Foodbank and the Street Pastors scheme.	 The people most in need are supported to live full active lives. Where people feel safe Reduced inequality between communities across Bath & North East Somerset

Scouts, Kelston Rd	The aim of The Scout Association is to "promote the development of young people in achieving their full physical, intellectual, social and spiritual potential, as individuals, as responsible citizens and as members of their local, national and international communities". It achieves this through a programme of activities for young people from the ages of 6 to 25.	 The people most in need are supported to live full active lives Children and young people enjoy their childhood and are prepared for adult life There are activities and opportunities to help young people to make a positive difference to their lives and communities Everyone has the opportunity to participate in sports, leisure and cultural activities
Beacon Hall/Beacon Field, Peasedown St John	This purpose-built Hall and nearby children's play area, were provided for the village of Peasedown St John as a result of the construction of over 1000 new homes in the village. The site of the Hall and its surrounding gardens, Beacon Field and car park is now leased to Peasedown St John Parish Council by Bath & North East Somerset Council. This will bring benefits to the Hall and the management of the adjacent space.	 The people most in need are supported to live full active lives Children and young people enjoy their childhood and are prepared for adult life Everyone has the opportunity to participate in sports, leisure and cultural activities Clean streets and open spaces
Midsomer Norton Town Trust	A new Midsomer Norton Town Trust has been established to hold the leases for buildings in trust for the town, be responsible for maintaining the fabric of the buildings and to develop a capital sinking fund over time to cover the cost of repairs. The 11 Town Councillors will be the trustees. The Midsomer Norton Community Trust will be a Community Interest Company responsible for running the Town Hall and Orchard Hall buildings on a daily basis.	 Clean streets and open spaces The quality of the environment is maintained or enhanced

Policy Context to the Budget 2013/2014

Context to our plans

Bath & North East Somerset Council (B&NES) strives to deliver excellent services to local people and make our area *the* place to live, invest, and visit in the UK. We are committed through our strategies to making this happen.

The environment in which we are working has changed and we are now facing new challenges from cuts to public sector funding, new legislation and shifts in our local population numbers and profiles.

Using a comprehensive evidence base, called the Joint Strategic Needs Assessment, we are ensuring that we have a good understanding of the challenges facing us and the needs of local people. This understanding is shaping our strategies and the way we deliver front line services.

Our strategies and plans describe how we will be investing in the future with projects to promote new jobs and homes, keeping Council Tax bills as low as possible and protecting high priority frontline services, whilst ensuring that our most vulnerable residents are given the support that they need.

What our plans say

Our Corporate Plan sets out the Councils vision for the area. This vision puts people first and communities at the heart of everything we do, and is framed around 3 objectives: *Promoting independence and positive lives for everyone (quality of life); Creating neighbourhoods where people are proud to live (people); Building a strong economy.* The Plan goes on to describe how we will make this vision a reality, whilst making the most of our resources. Our Corporate Plan 2012 – 2015 can be found at www.bathnes.gov.uk/services/your-council-and-democracy/policies-and-plans/corporate-plan

Our Sustainable Community Strategy (SCS) builds on our vision for the area and sets out the commitment from the wider B&NES partnership. It seeks to ensure that public services in B&NES are coordinated, and that our themed multi-agency partnerships such as the Health and Wellbeing Board and the Economic Partnership are joining up on key issues such as unemployment, skills and education.

We are in the process of reviewing the Sustainable Community Strategy to make sure that it is up to date and relevant; a new version of the SCS is expected in spring 2013. The B&NES Public Services Board is leading the development of the SCS. This Board is chaired by Paul Crossley, leader of the Council, and includes chief officers from B&NES public sector agencies including the police, fire and rescue, housing and health.

Engagement with local people and communities is an important part of our strategy and service design process and our new Local Engagement Framework is helping us to develop a coordinated, transparent and efficient way of engaging. This new approach will help us to reduce duplication, whilst making the most of new ways of working. The following paragraphs summarise some of the key challenges identified in the Joint Strategic Needs Assessment and priorities of our Corporate Plan and Sustainable Community Strategy.

Population change: Aging population

The number of older people is growing. Statistical projections suggest that this will continue and that by 2026, people over 75 will have increased by 40%, representing 11% of the total population. This will create challenges in services such as social care, health and housing.

Responding to this population change is a priority of our Health and Wellbeing Board and through the Joint Health and Wellbeing Strategy the Board has been planning how to meet the needs of our older population. Our commitment is to provide good quality services for older people that include choice, dignity and the respect to make their own decisions, as well as assistance to live independently.

Population change: School age population

The number of school age children is growing and, combined with new housing and families moving into the area, we will need to provide more primary school places. We have embarked upon an investment programme and new school places are proposed in north Bath, Paulton, Peasedown and Keynsham to meet this demand.

New housing developments are being considered in Bath Western Riverside, Bath and Somerdale Keynsham which will create additional demand on schools. We are planning for this now, including the need for new schools. We will continue to invest in children and schools ensuring that we continue to create a learning environment that is nurturing, safe and sets high expectations.

Promoting new jobs

We want to do more to support local people into work. Investing in economic growth to create new jobs, new homes and new opportunities for our residents is a crucial part of our local plans and our developing Economic Strategy will further outline this commitment.

Climate change

Climate change poses significant challenges for the area and we need to move towards a more sustainable future. Our Environmental Sustainability and Climate Change Strategy outlines our approach to achieving this including how we are working with the community to reduce our carbon consumption by 45 per cent by 2026, through a range of programmes that promote resource efficiency.

Connecting families

The government estimates that there are more than 200 families living within B&NES which have complex needs. These families experience five or more of the following problems; unemployment, poor quality or overcrowded housing, no parent has any qualifications, mental ill-health, longstanding limiting illness, disability or infirmity, low income, or unable to afford food and clothing items. This is not acceptable, and we are

working to ensure that these families enjoy the same quality of life experienced by others, through our Connecting Families programme.

Reducing health inequalities

Despite relatively low levels of social inequality, there are small geographical areas that experience issues. Social inequality has a significant relationship with a wide range of health and social care needs including life expectancy and long term conditions. In partnership with the B&NES Clinical Commissioning Group we will seek to reduce these inequalities through our Joint Health and Wellbeing Strategy.

Legislative changes

There have been many changes in legislation in recent years and these represent significant shifts to the way we work and how we deliver services.

The **Academies Act** has affected the way we manage and fund our schools. We have responded well to this change and are working collaboratively with our new academies to ensure that every child receives a high standard of education.

In response to the **Localism Act**, the Council is considering a range of mechanisms which will facilitate greater community involvement in the planning of local services, such as neighbourhood plans, a new local engagement framework and the new community rights to buy and challenge.

We have established a new Health and Wellbeing Board, which was a requirement of the **Health and Social Care Act**. This Board gives us a greater say in the way health services are delivered, and increases the opportunity for greater integration between health and social care. The overall aim of the Board is to improve health and wellbeing and reduce health inequalities.

Forthcoming changes to **state welfare provision** will impact on people and families within our communities. We are putting mechanisms in place now to support people impacted by this change. We are also working with the new **Police and Crime Commissioner** to ensure community safety initiatives in B&NES continue to receive investment and that crime remains low. This page is intentionally left blank

City Region Deal - Growth Incentive Background

Background

- 1 This note provides context for the budget and medium term plans. The impact does not start until 2014/15. Key decisions are not required until June and July 2013. At this stage the information is just for noting as it provides important context for the medium term plans especially intended investment in regeneration.
- 2 The City Deal offers potential additional financial benefits and investment as a result of a commitment to Growth. Extensive preparatory work is taking place to enable options to be put before councils. At this stage each of the 4 West of England Unitary Councils and Government have agreed in principle only.
- 3 The City Region Deal is an agreement between Government, the West of England authorities and the West of England Local Enterprise Partnership giving increased local financial flexibility and freedoms in exchange for a focussed programme of investment to enable the region to achieve its full economic growth potential.
- 4 The City Region Deal is made up of five key elements, most of these have already been agreed by the respective councils. The five elements are:
 - A Growth Incentive Proposition
 - A Transport Devolution Agreement
 - A People and Skills Programme
 - A City Growth Hub
 - A Bristol Public Property Board (Bristol City Council Only)
- 5 This briefing note focuses on the first of these, the Growth Incentive proposition, which is still under development and negotiation between the partners and Government. It is intended that these negotiations will be completed by Easter, enabling formal decisions to be taken by each Council in July, with implementation starting from April 2014. Given the significant nature of this element of the deal approval will be required by Full Council.
- 6 The Growth Incentive deal will support the strategic objective of delivering accelerated economic growth by establishing and delivering a £1bn Economic Development Fund which will be targeted to projects that will maximise economic returns. It will also be used to manage local demographic and service pressures arising from economic growth. This proposition will be key to the delivery of some of the other elements of the City Region Deal (e.g. transport).
- 7 The deal looks to enable the four West of England Councils to keep a larger proportion of their business rates growth locally than would otherwise be the case (rather than passing this money back into the

national system) over the next 25 years. In return, the Authorities will agree to pool this additional business rate income, and invest it in part to support the £1 billion Economic Development Fund referenced above, and also in supporting any extra demographic pressures each council may face.

What does the Growth Incentive Deal mean in practice?

- 8 The new Local Government Finance system sees the launch from 2013/14 of the Business Rates Retention scheme as the principal form of local government funding. A council's overall funding will be dependent on it being able to meet its baseline level of business rate generation in any year. However, over and above this, councils will be able to keep up to c50% of any additional business rates they generate until the next national 'reset' date (currently 2020 then every 10 years), less where relevant for some council's the payment of a 'Levy'.
- 9 The Growth Incentive proposition seeks a 'licensed exemption' from certain elements of the national local government finance system (notably the 50% share, levy, and reset) for any growth delivered across the 5 West of England Enterprise Areas. The 5 Enterprise areas are:
 - Avonmouth/Severnside
 - Bath
 - Weston Junction 21
 - Filton
 - Emersons Green
- In practice this would mean any growth in business rate income across these enterprise areas from April 2014 would wholly be kept by the West of England for the next 25 years. For these areas the respective councils would not be required to pass back into the national system the 50%, would get to keep all growth for the full 25 years (not lose it at each national reset), and where relevant would not need to pay a Levy. The council's would however agree to pool this element of the business rates. It is important to note that this is only in respect of the Growth in business rates in these areas, any current business rates income being collected would still be subject to the national system.
- 11 Whilst detailed financial modelling work is currently still being undertaken, and therefore any figures should be treated with extreme caution, this could mean up to an additional £0.5 billion of income including the Enterprise Zone could be retained for investment within the West of England rather than being lost into a national system. This is clearly cumulative over 25 years, and builds more quickly in the latter years.
- 12 As referenced above, the income from the enterprise areas would need to be pooled across the 4 Unitary Councils into one business rates pool. The income from the Bristol Temple Gate Enterprise Zone would then be added alongside this pool to create a substantial investment pot.

- 13 The four councils would then, in consultation with the Local Enterprise Partnership, agree how this total pot, as outlined in 10 above, would be invested. It would have three primary calls upon it.
 - To make sure, as far as is reasonably and practically possible, that no individual council is worse off under the growth incentive deal than they would be had it not taken place. To meet the costs of operating the pool and the growth incentive proposition.
 - To invest in the £1 billion Economic Development Fund, and through it unlocking the infrastructure investment required to generate additional economic growth.
 - To support each council's budget in meeting some of the additional demographical pressures growth may bring with it.
- 14 The exact list of investment to be delivered through the Economic Development Fund (EDF) and supported by the business rates pool is still under consideration, and is likely to evolve over time. This will be determined through the Local Enterprise Partnership following engagement and agreement with each of the Local Authorities. As the investment into the EDF will not be possible in the first place unless the business rates have either been generated (or there is a high likelihood they will be in the future), schemes supported by the pool and approved through the EDF are likely to be prioritised on the basis of:
 - Critical infrastructure which unlocks enterprise sites which are entirely dependent on growth incentive funding.
 - Bringing EA/EZ infrastructure schemes that are already funded forward in the programme to generate economic growth and business rates earlier in the cycle than would otherwise have been the case.
 - Other infrastructure schemes within the EA/EZ areas which demonstrate a positive impact on the economy and business rates.
 - Other economic development schemes within or outside the EA/EZ areas.
- 15 Whilst detailed financial modelling is still underway, the above programme does generate the option of borrowing in advance of business rate take to pump prime and release the investment in sites earlier than would otherwise have been the case. If this scenario is followed each council will want to assure itself of the potential financial risks (and rewards) this generates. This is likely to require each Local Authority being responsible for underwriting its own borrowing so that no Local Authority is responsible for another Council's borrowing, despite the pooling of extra business rates income should that income not be sufficient to meet the costs of repayment. The opportunities and parameters around this will be further developed over the next few months.

Conclusions

- 16 The Growth Incentive Proposal brings with it significant opportunity to secure additional funds for the West of England. Equally these opportunities do not come without risk. Much work is still required before final proposals can be put before members for decision in July. This will require significant negotiation not only between the unitary councils and the LEP, but also with central government.
- 17 A robust project governance structure is in place to ensure the project delivers to time and also negotiates the best possible deal for the West of England. More detailed proposals will be available for consultation in good time to enable firm recommendations are made during June for final decision by each Council in July.

MEDIUM TERM SERVICE & RESOURCE PLANS – 2013/14 to 2015/2016

FINANCIAL PLANNING ASSUMPTIONS

1. <u>Context – The Financial Challenge</u>

The Council's Budget for 2013/2014 presents a full and detailed Medium Term Service and Resource Plan for the three-year period from 2013/2014 through to 2015/2016. This enables the Council to take a planned and structured approach to meet the significant financial challenge facing local government.

2013/214 represents the third year of financial planning prepared in the context of the Government's Comprehensive Spending Review (CSR) announced in October 2010. This CSR included a deficit reduction programme with 24% cuts to local authority spending spread over the four year period from 2011/2012 to 2014/2015.

However as we approach the next Comprehensive Spending Review in 2013 it is clear that the reductions set out in the previous CSR are not sufficient to meet the Government targets to reduce the fiscal deficit, as the on-going impact of economic uncertainty both across Europe and indeed worldwide, means the UK economy continues to fall short of previous expectations.

The financial implications for the Council over the next two years (2013/14 and 2014/15) were set out in the Provisional Local Government Financial Settlement announced on 19th December 2012. This showed that over the case CSR period, the Council will see a 40% reduction in government funding.

This overall position is further complicated by a range of significant changes impacting on local Government Finance as set out below.

- The needs based Formula Grant funding system (the Four Block Model) for local government is being replaced by a combination of localised Business Rates and a top up grant to be known as Revenue Support Grant.
- The new Localised Business Rates (National Non Domestic Rates) will provide for the Council to retain up to 49% of local business rates going forwards to incentivise growth. The Council will also share the financial impact of non-payment, business contraction/cessation and business rate appeals. There will be a safety net where business rates decrease by 7.5% or more. This system will be reset from time to time to allow an element of rebalancing – the first such reset being scheduled for no earlier than 2020.
- The new Revenue Support Grant uses a baseline needs assessment for 2013/2014 and has been set broadly at a level to cover the gap between baseline funding need and the initial share of local business rates. The RSG will then be reduced to reflect Government savings requirements from 2013/2014 onwards.
- Responsibility for setting Council Tax Benefit passes to local authorities from 2013/2014 in the form of the new Local Council Tax Support Scheme. At the same time the funding from Government will be

reduced by 10% resulting in a shortfall of around \pounds 1.5M, which to be met from adjustments to the new scheme as approved by the Council in November 2012. The elderly and most vulnerable claimants will be protected.

- Anticipated reforms to the Planning System to provide for full cost recovery did not progress as expected. Some fee increases are being permitted and factored into the MTSRP, but this falls far short of the anticipated levels that had been factored into the base budgets.
- Public health responsibility and related services will pass to the Council from April 2013, together with an appropriate budget transfer from the PCT. It is assumed the grant received will fully cover all related costs of this service.
- The full implementation of planned changes to Government Funding for LEA and Academies through the Education Services Grant. The resulting reductions in funding in excess of current levels of spending have been factored into the current MTSRP although this will become increasingly challenging as more schools move to become Academies.
- Early years funding for 2 year olds will move from the LEA into the Dedicated Schools Grant which is primarily a technical change however the a significant element of the remaining funding for Early Years within LEA's has been held back by the DfE This reduction in funding is reflected in the MTSRP.

These issues are reflected within the Medium Term Service and Resource Planning process for 2013/2014 to 2015/2016 to the extent the impacts can be reasonably anticipated.

There are also a range of service specific cost pressures including impacts of national policy changes. The most significant of these include:

- Rising elderly population placing significant demands on Adult Social Care and Health services.
- Increased demand for Children's care services.
- Contractual inflationary costs particularly for care placements and external service contracts.
- Local impacts of the economic downturn and increasing competition e.g. car parking income.

Taking account of the known and anticipated reductions in government grant funding and the pressures outlined above, it is estimated that at least £30m of budget savings are required over the period 2013/2014 to 2015/2016.

2. The Budget approach for 2013/2014 to 2015/2016

The sound financial management of the Council over the years means it is in a better position than many other councils to face the continuing financial challenges arising as a result of the national economic situation.

The Council Budget covers the period from 2013/2014 to 2015/2016, recognising the very difficult financial challenge now facing the whole of the

public sector and the increasing need to prioritise resources. The following principles have been used to support this:

- Protecting frontline services;
- No increases in Council Tax;
- Investing in homes and jobs for local people.

There are no longer the available resources to deliver the full range of services that have been provided in the past. New legislation and demographic changes similarly demand clear prioritisation and new approaches. This increasingly means difficult choices.

The development of the Budget has moved away from setting targets and budget top slices based on historic spending to and approach more focussed on prioritisation supported where appropriate by zero based budgeting. This approach has included: -

- Ensuring only essential cost pressures are taken into consideration, challenging all proposals for inflationary increases and additional spending.
- A continued focus on achieving efficiency savings within and across service areas.
- Maximising savings achieved through the continued development of the Change Programme with projects like Customer Services and Procure to Pay.
- Seeking to increase income from new and existing sources. Developing and investing in a diversified income base to help protect the Council from reductions in Government funding.
- Minimising costs of borrowing utilising Council cash flow balances where appropriate to provide funding for capital projects.
- Exploring opportunities to support Communities to enable them to be more resilient and self-sustaining.
- Making better use of Council Assets, particularly council land and property, to reduce running costs and provide capital receipts.
- Where Government is cutting its grants to local authorities, or other external sources of funding are being reduced, these savings requirements may need to be passed on to the relevant service.

The scale of the projected savings required over the next three years, is such that the Council will need to prioritise services and whilst every effort has been made to protect essential frontline services for local people, this will inevitably lead to reductions in service areas which are considered a lower priority.

The proposals put forward in the MTSRP's provide for a balanced budget in 2013/2014 with 2014/2015 and 2015/2016 subject to government funding announcements and potential new spending pressures. These future years will also be dependent upon improvements to the global and national economy and whilst these MTSRP's proposals go some way to addressing the financial challenge in these years, it is likely that further savings will be required.

3. <u>Council Tax</u>

Council tax levels have now been frozen since 2010/2011, supported by Council Tax Support Grants from the Government. The Council Tax Support Grant for 2011/12 is now part of the baseline funding calculation for government grant, whereas the Council Tax Support Grant for 2012/2013 was a one off grant. Each of these grants was conditional on a Council Tax freeze in the respective financial years.

As part of the Provisional Local Government Finance Settlement on 19th December 2012, the Government confirmed the provision of grant funding to support to councils who freeze their Council Tax for next year (2013/2014) at the current level (i.e. a zero increase). The grant is equivalent to a 1% increase in Council Tax (approximately £740K) and has been confirmed as forming part of the future baseline funding calculation for those councils that accept and meet the conditions of the grant.

This Settlement also confirmed that Council Tax increases over 2% would trigger the legislative requirements for a local referendum on the proposed Council Tax increase.

The Council Budget provides for the acceptance of the Government Council Tax Freeze Grant and provides for a zero increase in Council Tax for 2013/2014.

4. <u>Government Grants</u>

The Council received approximately £41m in formula grant from the Government in the last financial year which, was distributed using a complex needs based formula known as the Four Block Model. This formula includes significant weightings attached to deprivation based indicators across a range of specific service blocks

This needs based formula has been replaced from 2013/2014 as part of the Local Government Resource Review. This formula is currently being updated in order to arrive at a baseline funding level for local authorities. This will be used as the starting point for the new system – beyond this point funding needs will only be considered on a periodic basis to reset funding for local authorities. The first such reset will not be until 2020

The Government originally set out its plans for tackling the significant budget deficit, in the Comprehensive Spending Review on 20th October 2010. This deficit reduction programme indicated 24% cuts to local authority spending over the Spending Review period from 2011/2012 to 2014/2015 with a significant element front loaded to the first two years.

The provisional Local Government Finance Settlement announced in December 2012 showed the Council's funding baseline for 2013/2014 reducing by a further 9.7% and then by a further 8.7% in 2014/2015. This equates to a 40% reduction in funding for the Council, well in excess of the average set out in the CSR.

The funding baseline itself has been reset to take account of the start-up funding baseline for the new Localised Business Rate element. This Business Rate baseline position is fixed and only uprated in line with inflation for at least seven years.

The Government's Autumn Statement in December 2012 reflected the ongoing market concerns surrounding the national debt position of the UK and a number of Eurozone countries. In particular the statement confirmed the need for public sector spending deficit reduction measures to continue beyond at least until 2018. The Statement also confirmed a ceiling on public sector pay of 1% per annum for the next two years.

The grant funding baseline has also been updated to remove the funding for LEA services which will now be paid directly to the LEA and Academies as the Education Services Grant (ESG) on a per pupil basis. We had anticipated this adjustment although the rate of schools transferring to Academies (particularly secondary schools) does present a funding challenge for the Council.

The New Homes Bonus Scheme which provides funding for new homes brought into use and included in the Council Tax Base, was confirmed at a total of \pounds 1.977m for 2013/2014. This is the third year for funding for New Homes Bonus and includes an additional \pounds 0.759m for 2013/2014. The New Homes Bonus scheme runs for 6 years from the year of initial allocation. This funding is un-ring-fenced and available for allocation as part of the overall revenue budget.

The Final Local Government Finance Settlement is expected at the end of January 2013 and any significant changes to the previous announcement will be highlighted.

5. <u>Business Rates</u>

From 2013/2014 the new arrangements for funding of local government will provide for up to a maximum of 49% to be retained locally.

The actual amount to be retained is dependent upon a number of factors including a start-up baseline assessment, a top-up or tariff payment subject to the level of the baseline and a levy on future growth to support a national safety net.

The setting of the baseline is key and for the Council and original budget planning reflected the Government proposal to use a historic 5 year average for this purpose. A late change in this baseline calculation by the Government to move to a 2 year average, effectively increases the Council's annual tariff payment to the Government by over £1M.

Based upon this the Council will retain 49% of the Business Rates less a tariff payment of \pounds 9.468m to the government using the initial baseline. Going forwards 49% of all future growth in business rates will be retained by the Council less a levy payment of 31p in the \pounds to provide for a national safety net.

Under this system the Council benefits from new growth but also bears 49% of the cost of reductions in business rates from for example, appeals, business closures or relocations, charitable reliefs etc. A national safety net will be provided where business rates reduce by over 7.5%, equivalent to approximately £2M for the Council.

No specific provision for growth has been made for 2013/2014 reflecting that fact that for this Council the initial years of this new system will be very challenging as some large business closures will impact on the Business Rates collected, particularly MOD sites and Somerdale.

As part of the proposed Budget, reasonable assumptions have been made for likely future Business Rates although given these new arrangements have not been tested and current economic conditions continue to be challenging, there remain a significant risk associated with future receipts.

The Council has been seeking further changes to some elements of the new system, including the baseline calculation and the treatment of historic appeals, which may provide some flexibility in the overall Business Rates Collection Fund going forwards. It is anticipated that the outcome of this will be made clear in the Final Settlement announcement.

6. Medium Term Service and Resource Plans

The Medium Term Service and Resource Plans cover the financial planning period from 2013/2014 through to 2015/2016 and have been prepared by each service area to reflect the details of the specific proposed savings to ensure the Council is in a position to consider a balanced Budget for 2013/2014 and as close to balance as possible at this stage for the entire 3 year period.

As set out in Section 2, the process was based on prioritisation of savings in order to meet the projected need to find £30M of spending reductions over the next three years.

At this stage, that there is insufficient information available to fully identify future funding pressures including new government requirements, future demand changes and emerging issues. It is therefore expected that the requirements for savings in future years may increase.

Given the scale of efficiency savings already achieved and those already planned within the MTSRP's, it is likely that any further savings will require additional changes or reductions to Council services. A rigorous process will therefore continue to be applied to the budget and medium term financial planning process to enable resources to be prioritised between service areas.

7. <u>Reserves</u>

The Budget provides for the Council's General Fund Balances to be maintained at a risk assessed minimum level of £10.5m.

There are no assumptions to change this position going forwards and the risk assessed levels will be reviewed as part of the final Budget proposal in Feb 2013.

A range of Earmarked Reserves are maintained by the Council for specific purposes. The likely commitments against each of these reserves will be regularly reviewed as part of the on-going monitoring and development of the Budget across the medium term service and resource planning period.

The Council's reserves position remains relatively strong and continues to provide some flexibility to support the Budget over the medium term, particularly to facilitate timing and implementation of recurring savings.

Any proposed use of reserves will recognise that they can only be used once, and will take account of the overarching principle of not using reserves to provide support for recurring budget pressures.

8. Pensions

The most recent actuarial review as at 31 March 2010 concluded a number of positive factors which did not require any significant variation in the Council's employers contribution level overall. These factors included:-

- The Avon Pension Fund investments have performed relatively well albeit since that review investments generally have been volatile and affected by poor stock market performance.
- The Government has switched the rate for future pensions increases from the Retail Price Index (RPI) to the historically lower measure of the Consumer Price Index (CPI).
- A national review of public sector pensions schemes is being undertaken by the Government (the Hutton Review).

The outcome of this actuarial review has been factored into the Budget plans and whilst no change was provided for in terms of the overall contribution level for the Council, the implications of a reducing workforce has required a further adjustment by the Council Budget in 2013/2014 to maintain this neutral cash position going forwards.

Work is currently underway on the next Actuarial Valuation due as at 31 March 2013. This valuation will take into account the national changes to the Local Government Pension Scheme from 1 April 2014 reflecting changes to the benefits structure of the scheme (from final salary to career average) and employee contribution rates which are expected to help reduce the cost of the scheme.

The actual implications of this review is likely to lead to changes in employer contribution rates from 2014/2015 but in recognition of the points above, a small 0.5% increase has been factored in from this point.

9. <u>Pay</u>

Discussions are currently taking place nationally between the Employee and Employer representatives regarding the potential pay award offer for 2013.

Provision has been made within the MTSRP for a small increase (1%) in line with national government expectations for a public sector pay in 2013/2014. Similar provisions have been made for 2014/15 and 2015/16.

10. Other Assumptions

Some of the other key assumptions used in the development of the medium term plans include:

- A contractual inflation target of 2% has been provided for each year throughout the period where it is deemed essential, except in the case of Adult Social Care costs where the provision for inflation has been set at 1.75%. No further inflation has been provided for general supplies and services.
- Balanced budgets are delivered for 2012/2013 there is no provision for overspending.
- Interest earnings on the Council's cash balances are based on a 0.75% return – this will be reviewed in line with the Council's Treasury Management Strategy.

BUDGET SETTING PROCESS – ADVICE OF THE MONITORING OFFICER

- 1. It is important to be clear on the process to be followed in setting the 2013/14 Budget. This paper sets out the guidance provided by the Council's Monitoring Officer.
- 2. The Cabinet has the responsibility to prepare and propose a draft Budget to Council for its approval.
- 3. The Cabinet can, in its absolute discretion, receive from any political group that so wishes, an alternative budget proposal to that published in the Cabinet agenda papers .It can only consider such proposals if it is satisfied that they have been discussed with the Council's statutory officers and relevant Directors and that an impact statement from Officers about such proposals is available.
- 4. All proposals that the Cabinet meeting is prepared to consider will therefore be cleared with the Section 151 Officer, the Monitoring Officer, and relevant Strategic / Divisional Directors beforehand.
- The Cabinet will formulate a budget proposal and Council Tax recommendation to the Council meeting on 19th February 2013. Such budget proposal may either take the form of a composite proposal or may include agreed core proposals and options for allocating parts of the budget.
- 6. The Council has available to it at the budget setting meeting two options. It can object to specific parts of the proposals and if it does so must require the Leader on behalf of Cabinet to reconsider its proposals. The Council is required to give the Cabinet the reason(s) why it considers those proposal(s) should be changed and it is then for the Cabinet to consider those proposed changes and the reasons put forward. Alternatively it is, of course, open to the Council to accept the budget in its proposed form at the meeting, in which case no further action is necessary.
- 7. Council may then determine the budget on the basis of the Cabinet's recommendations as set out in paragraph 5 above plus any insignificant changes adopted as amendments at the Council meeting. The constitution provides that the meeting itself (on advice from the Chief Executive) will decide whether any amendment to the budget proposals is of such significance as to amount to an "objection" to the budget so as to require reconsideration by the Cabinet.

- 8. However if a significant proposal is accepted on a vote at Council, from those notified at the Cabinet meeting, this stands as a formal objection within the terms of the law and will be referred to the Leader for him to secure consideration by the Cabinet and report back to the Council meeting on 28th February 2013.
- 9. When the Cabinet has considered the objections, it is required to put its proposals (which may or may not be revised) back to the Council Meeting. If the Cabinet does not agree with Council's views on a proposed change, it is required to state why and the Council can then take those reasons into account, along with its original thoughts as to why the change was desirable. At the meeting, it is open to Council to take such decision as it sees fit on any variation from the budget as originally proposed that has been the subject of consideration under the process outlined in paragraphs 6 to 8.
- 10. In setting the budget the Council is required to approve a full budget resolution including the police, fire and parish precepts and the proposed Council revenue and capital budgets for 2013/14. That budget will include within it the overall proposed Council cash limits for 2013/14 including the provision for inflation, the proposed use of balances in the 2013/14 budget (if any) and the resulting budget requirement and Council Tax for Bath and North East Somerset including any recommendations for special expenses. The Council will also approve the borrowing limits for 2013/14 and prudential indicators.
- 11. Legally, the Council must set a balanced budget for the forthcoming year and determine the level of Council Tax. If a budget is not set by the date of the reserve budget-setting meeting (28th February), this will lead to a delay in billing and a loss in council tax cash flow. It is highly likely that this will also translate into a higher level of uncollectable debt and debt collection costs and in addition this will significantly impact on council tax performance indicators. A delay until 28th February will also compromise the Council's ability to meet current billing deadlines, and there is a serious risk billing will be delayed also with negative cash flow impacts.
- 12. The final Council Tax set will encompass all parish and police and fire precepts (that is the money we collect on behalf of the parishes, fire and police and pay to them).

Vernon Hitchman Divisional Director – Legal & Democratic Services (Monitoring Officer)

Purpose and scope

- 1.1 The purpose of the statement is to provide a clear and transparent policy to the public, which demonstrates accountability and value for money for the financial year 2013-14.
- 1.2 The policy meets the Council's obligations under the Localism Act 2011 [Section 38 (1)] and the associated statutory guidance set out in the Openness and Accountability in Local Pay: Guidance under section 40 of the Localism Act (February 2012) together with the Code of Recommended Practice for Local Authorities on Data Transparency (September 2011).
- 1.3 The policy must articulate the Council's own policies towards a range of issues relating to the pay of its direct workforce, in particular its Chief Officers, as defined by the Local Government and Housing Act 1989 and lowest paid employees. Details of 'Chief Officers' employed by the Council can be found on the Council's public website.
- 1.4 The pay arrangements reflect the need to recruit, retain and motivate skilled employees to ensure high levels of performance balanced with accountability on the public purse. The policy recognises flexibility which is essential in delivering a diverse range of services and is underpinned by principles of fairness and equality.
- 1.5 The pay policy statement applies to both the lowest and the highest paid. In accordance with provisions of the Localism Act, it does not extend to schools and this statement does not, therefore, include school based employees.
- 1.6 The statement will be approved by Full Council, i.e. not delegated as an executive or committee function, in advance of the financial year to which it relates and must be reviewed at least annually. Any amendments must be approved by Full Council.
- 1.7 The Council will publish the statement on its website.

Definitions

For the purposes of this Pay Policy Statement the following definitions apply:

- 2.1 **'Pay'** in addition to base salary includes charges, fees, allowances, benefits in kind, increases in/enhancement to pension entitlements and termination payments where applicable.
- 2.2 **'Chief Officers'** refers to the following roles in the Council:

Statutory Chief Officers are:

- Chief Executive, as Head of Paid Service
- Strategic Director People & Communities, as Director of Children's services and Director of Adult Social Services
- o Divisional Director Legal & Democratic Services as Monitoring Officer
- Divisional Director Finance, as Section 151 Officer (Chief Financial Officer)

Non Statutory Chief Officers are:

- Strategic Director Place
- Strategic Director Resources

Other Directors/Heads of Service are:

- Senior managers who report director to/or are accountable to a statutory or non-statutory Chief Officer in respect of all or most of their duties.
- 2.3 **'Lowest paid employee'** refers to those employees refers to those employees in substantive full time employment currently earning £12,145.00 gross per annum. This is the lowest scale point in the council's grading structure and has been assess through the Council's job Evaluation scheme as having the least amount of complexity and responsibility.

General principles & practice

3.1 **Principles:**

Bath & North East Somerset Council values all its employees and aims to apply a consistent and fair approach to pay and benefits in line with the following principles:

- To work within financial constraints and use those limited funds in the most effective way to support the Council in the provision of quality cost effective services and its workforce needs
- To aim for consistency and fairness in the processes used to manage pay and benefits, as appropriate to service delivery and in line with its commitment to remaining within the framework of the relevant national pay and conditions agreements
- To promote an equal pay agenda by ensuring that pay and job evaluation systems, processes and systems meet legislative requirements and to actively work towards reducing any unjustified gender pay gaps
- To ensure that our pay and benefits processes and policies are transparent and accessible to all employees
- To be mindful of the market in making decisions about pay and benefits
- To be clear about how we recognise and reward performance, whether at whole organisation, service, team or individual level
- To support a flexible approach to the acceptance of changes to tasks, duties and responsibilities by employees and allow for flexibility between posts.
- To enable the Council to attract and retain its employees and in order to do so, respond to situations where market forces dictate the necessity to apply supplements to established salaries.
- Any pay structure must be affordable to introduce and maintain.
- To aim to retain a core set of benefits for all employees.

Practice:

3.2 **Basic pay is determined through**

- The job role and it's accountability in the overall context of the Council's services and responsibilities using the HAY job evaluation process which is based on objective criteria and free from discriminatory bias.
- Ensuring that all employees are dealt with on this basis with no distinction being made for senior management appointments including Chief Officers and their Deputies.

Pay Policy Statement 2013 -14

- The terms of the relevant national agreements on pay and conditions of service
- The amount available for the pay review process is also impacted by what the Councils which are party to the National Agreements can collectively afford.
- A comprehensive pay and grading structure has been adopted that positions the Council against median salary benchmarking compared to a national data base maintained by the Hay Group, is affordable and offers recruitment and retention incentive. This is kept under review.
- The outcome of reviews into the local pay and grading structures are determined within the terms of this policy and the Council's constitutional arrangements.

Note: This excludes trainees, apprentices and interns who are paid less to reflect the nature of the training and development role.

3.3 **Pay on appointment**

- Staff are normally appointed at the bottom scale point of the grade at which the post has been evaluated.
- Managers have discretion to appoint at a higher scale point within the grade band if the appointee can demonstrate that they are currently earning more than the minimum salary for the grade.

3.4 **Pay review dates**

- Grade progression (i.e. movement from a lower to a higher salary scale point (scp) within a grade) takes place on 1st April of each year until the highest scp is reached.
- Grade progression is subject to satisfactory performance (and may be withheld if performance has been unsatisfactory) and a minimum of 6 months service in the grade. Where 6 months service cannot be achieved by 1 April, progression is considered on the anniversary of six months service.
- Where an increase in pay has been negotiated through the national negotiation framework, it will be implemented with effect from 1st April of the appropriate year. If the negotiations have not been concluded by 1st April, the increase will be paid at the earliest opportunity together with back pay from 1st April.

3.5 Honoraria & other allowances

- Work outside the scope of the post can be recognised by the award of an honorarium. The conditions and framework are set out in the 'Recognition for work outside the scope of the post' policy. Assessment and payment will be based on non - discriminatory, objective criteria.
- Allowances, for example standby, may be made to employees, below senior managers in connection with their role or pattern of hours they work in accordance with National or local collective agreements.

3.6 **Re-employment of former local government employees**

The Council retains sufficient flexibility in its response to the re-employment of former local government employees to enable it to respond appropriately to the particular circumstances. It ensures that an open and fair selection process stakes place before any appointment is confirmed. 'Merit' is the sole criteria.

Pay Policy Statement 2013 -14

If the Council were to re-employ a previous local government employee who had received a redundancy or severance package on leaving, or who was in receipt of a pension covered by the Redundancy Payments (Continuity of Employment in Local Government Modification) Order 1999, known as the Modification Order) (with the same or another authority), then the Council's policy is to ensure that the rules of the Modification Order are applied.

3.7 Use of consultants, contractors and temporary 'agency' staff.

Ordinarily staff will be engaged directly by the Council as employees but on an exceptional basis, where particular circumstances deem it necessary, people may be employed through personal service companies or on an 'agency basis'. When this situation arises, the Council will give detailed prior consideration to the benefit of doing so and that the overriding need to ensure value for money is achieved. Such arrangements must be in accordance with the code of practice and require prior approval by the Strategic Director for the service area.

Equal pay

- 4.1 The Council is committed to the principle of equal pay for all posts of the same size and value and implemented the national 'single status agreement with effect from 1 April 2007. In order to put its commitment to equal pay into practice the Council:
 - reviews regularly its pay grade and rates for all current staff and starting pay for new staff in line with Equality and Human Rights Commission guidance in line with the Council's Equality policy.
 - informs employees of how these practices work and how their own pay is arrived at.
 - provides training and guidance for managers and supervisory staff involved in decisions about pay and benefits.
 - regularly monitors pay and grading data and statistics.

Ensuring consistency

- 5.1 The Council seeks to ensure consistency through the following processes:
 - All departments are provided with the same quality of internal support in the job evaluation process
 - The Human Resources Service (in partnership with senior managers, as appropriate) has an on-going responsibility to review pay levels across the Council and highlight any potential anomalies.

If there is an exceptional need to review pay outside of the normal pay review timetable, proposals will be considered and approved by the relevant Director and the Head of Human Resources.

Pensions

- 6.1 Subject to the provisions of the relevant scheme, all directly employed staff who are the subject of this policy are able to join a contributory statutory pension scheme.
- 6.2 The Council is required to set out its policies relating to discretionary payments, severance packages, employment of staff in receipt of pension and re-engagement of former staff under contracts for services. The Council has in place a policy for flexible retirement which is specifically authorised by statute whereby individual staff, with employer approval, may draw their pension and continue in employment at a lower pay grade/ working shorter hours. The Council considers all proposals on their individual merits but would not take any action beyond that authorised by existing policy without reference to the appropriate Council decision making body.

Senior pay

- 7.1 The remuneration of the Chief Executive and other senior management appointments in the Council (see Annex 1) is undertaken by external analysts using the Hay Job Evaluation process. Levels of pay have been market-related by being compared to a national data base maintained by the Hay Group of similar posts in a wide range of public and not for profit sector organisations. The pay structure for Chief Officers takes account of the clearly defined additional 'statutory 'responsibilities (see section 2 above). Five pay bands will be available place for the most senior officers as set out in the Annex 1.
- 7.2 Any increases in pay rates will be in line with those negotiated nationally by Joint Negotiating Committees (JNC's) for Chief Executives and Chief Officers respectively. The pay policy, whilst agreed in advance of the financial year to which it relates, can be can be amended during the course of the year to incorporate a pay award negotiated nationally or for other reasons.
- 7.3 Where there is a pay range for a job the Council's adopted aim is to offer an appointment to the minimum point of the appropriate salary band. In order to secure the services of the best candidate it may be necessary to offer a higher amount. In these circumstances approval by the employing Director or members of the appointing Member committee as appropriate, in consultation with the Head of Human Resources, is required
- 7.4 Where a pay band consists of a number of different salary points, any progression to the next incremental point is subject to satisfactorily meeting performance criteria agreed in advance with the Chief Executive or Strategic Director, as appropriate (in consultation with the Group Leader(s)). Any increase is paid from 1 April subject to 12 months service in that pay band and the maximum not being exceeded.
- 7.5 This is no provision for the Council to pay any bonuses, charges, fees or allowances, benefits in kind to senior employees or any other employees other than expenses necessarily incurred in the performance of their duties.

- 7.6 Senior staff are not differentiated from other members of staff in terms of remuneration on resignation or termination. The Council's general arrangements for severance and scheme for discretionary payments apply to this staff group as to all employees.
- 7.7 Other conditions of service are those determined nationally by the JNC's specifically for these appointments or, as locally determined for all other Council staff.
- 7.8 The Council's threshold level for disclosure of senior staff salaries will be at the minimum point of the senior civil service pay scale and above as at 31 March.

Relationship between senior pay and the 'lowest paid council employee'

- 8.1 The grading structure and pay line determine the salaries of the highest and lowest paid Council employees. The Council's highest paid employee is its Chief Executive (see Annex 1). The lowest salary offered for substantive, full time employment is currently £12,145.00 per annum (i.e. the lowest scale point (SCP) within the grading structure).
- 8.2 Following advice in 'Openness and Accountability in Local Pay: Guidance under section 40 of the Localism Act' and in accordance with the Code of Recommended Practice for Local Authorities on Data Transparency the Council publishes the ratio between the highest paid salary and the median salary for the whole of its workforce on its public website. It does not currently have a policy of maintaining or reaching a specific ratio of pay multiple between the Chief Executive and that of its median earner.

Publication

9.1 The Council's approach to the publication of and access to information on the remuneration of Chief Officers is to include it on its public website as part of its requirements within the Accounts and Audit (England) Regulations 2011 and in accordance with the Code of Recommended Practice for Local Authorities on Data Transparency.

Implementing a new organisational model for the Council

10.1 The Council is in the process of implementing different senior management arrangements to support a new organisational model based upon principles and a general approach agreed on 16 November 2010.

10.2 In order to recruit to the new Strategic Director roles, it has been necessary to determine the level of remuneration and conditions of employment using the general senior pay policy practice and principles set out above. Rates of pay have been established taking the advice of independent analysts and the Hay data base and are as set out below and will be effective from 1 April 2013. Work will continue during the coming year to complete the senior management structures below Strategic Director level. The remuneration for other posts in the new senior management arrangements will be determined within the terms of this policy and the Council's constitutional arrangements.

Further information

For further information on the Council's pay policy please contact the Council's Human Resource Service email <u>human resources@bathnes.gov.uk</u>. Tel: 01225 477203

Annex 1

SENIOR MANAGEMENT PAY & GRADING

BAND	MIN/MID/MAX POINTS	Composition, Terms & Conditions	
CHIEF EX	ECUTIVE & HEA	D OF PAID SERVICE	
public sector market median data provided by the Council's general starting salary policy - No variable element within the remuneration pack		 A fixed salary within the range £145,000 to £155,000 pa taking account of current public sector market median data provided by the independent analysts and the Council's general starting salary policy No variable element within the remuneration package. All other conditions in accordance with overall Council pay policy. 	
STRATEGIC DIRECTORS			
Band 2	Fixed/Spot	 People & Communities [statutory roles for Children's & Adult Services] (PC) Place (P) Resources (R) A fixed salary within the range £115,000 to £130,000 pa taking account of current public sector market median data provided by the independent analysts and the Council's general starting salary policy No variable element within the remuneration package. All other conditions in accordance with overall Council pay policy. 	

BAND	MIN/MID/MAX POINTS		
DIVISIONAL DIRECTORS/Heads of Service		leads of Service	COMPOSITION
		£91,869	Safeguarding, Social Care and Family Services (PC)
	Middle Maximum	£94,572 £97,275	 Learning and Inclusion Service (PC) Planning & Transport Development (P) Finance [S.151 Officer] (R) Property & Facilities [Chief Property Officer] (R) Legal & Democratic [Monitoring Officer]
Band 5	Minimum Middle Maximum	£81,063 £83,763 £86,469	 Non-Acute Health, Social Care & Housing (PC) Service Development (PC)) Skills and Employability (P) Environmental Services (P) Tourism, Leisure & Culture (P) [Vacant] Policy & Partnership (R) Improvement & Performance (R) Project Management (R)
Band 6	Minimum Middle Maximum	£70,254 £72,957 £75,657	 Customer Services Revenues & Benefits (R) Risk & Assurance (R) Transformation (R) Human Resources(R)

Notes:

- Details at 31 March 2013
- Band 3 is not currently in use
- There has been no annual pay award to any of this staff group since April 2008, for other staff since 2009

Bath & North East Somerset Council			
MEETING:	Council		
TITLE:	Transport Improvement Capital Programme 2013/14		
WARD:	All		
AN OPEN PUBLIC ITEM			
List of attachments to this report:			
Appendix 1 – Draft Integrated Transport Improvement Capital Programme 2013/14			

1 THE ISSUE

1.1 This report sets out the draft Integrated Transport Improvement Capital Programme for consultation. The programme of expenditure aims to develop the policies of the Joint Local Transport Plan in accordance with Government guidelines.

2 **RECOMMENDATION**

The Cabinet agrees that:

2.1 The draft Transport Improvement Capital Programme for 2013/14 as set out in Appendix 1 is approved for consultation.

2.2 The Group Manager, Transport and Planning Policy be delegated authority to alter the programme, in consultation with the Cabinet Member for Customer Services, as may prove necessary within the overall budget allocation.

3 FINANCIAL IMPLICATIONS

3.1 The 2013/14 capital Local Transport Plan Integrated Transport Block settlement is £1.225 million for transport improvement schemes.

3.2 Other funding includes S106 contributions of £29k Bristol Road Chew Stoke and a contribution of £10k from Batheastern Parish Council towards a footway improvement.

3.3 The Integrated Transport Block allocation contributes £124k to the proposed Greater Bristol Metro Project, which will be subject to separate financial approval, and supports maintenance programmes for replacing antiskid surfacing (£10k) and general highway maintenance through a contribution to de-cluttering programmes (£10k) to remove redundant street furniture.

3.4 In order to manage the budget effectively regular monthly monitoring will be carried out and exceptions reported to the Cabinet Member, with reporting to Transport Board on a quarterly basis, and programme changes approved through delegated decision making. Spending will be managed throughout the year to ensure that the overall budget is spent in a timely manner.

4 CORPORATE OBJECTIVES

• Promoting independence and positive lives for everyone

The Transport Improvement Capital Programme improves access to employment, education and health facilities for everyone and promotes low carbon, cleaner and healthier transport choices.

• Creating neighbourhoods where people are proud to live

The Programme helps to reduce the impact of vehicular traffic in local neighbourhoods and near schools through, for example, reducing speed limits, improving pedestrian crossing facilities and road safety and managing parking.

• Building a stronger economy

Along with improving access to employment by sustainable modes of transport, the Programme tackles congested pinch points on the highway network through, for example, selective junction improvements.

5 THE REPORT

5.1 The Joint Local Transport Plan 3, 2011 - 2026 sets out a number of key targets in relation to transport. The 2012 Annual Progress Report shows the clear majority of targets are on track and indicators improving.

5.2 Progress against the JLTP3 targets also contribute towards the wider objectives of the Local Sustainable Community strategy.

5.3 Appendix 1 shows indicative spending on schemes for 2013/14, subject to consultation with ward members and parishes councils.

5.4 The Greater Metro Project first phase provides half hourly train services for the Severn Beach line, local stations between Bristol Temple Meads, Bath Spa and Weston-super-Mare and the reopened Portishead line. The scheme is estimated to cost £36m to implement funded by devolved major scheme funding from the DfT and City Deal. The Council will contribute 15% towards the costs of preparing a business case to support the inclusion of additional routes and services in the new Great Western rail franchise specification due to commence in September 2015. North Somerset Council is the lead authority for the project.

6 RISK MANAGEMENT

6.1 The report author and Lead Cabinet member have fully reviewed the risk assessment related to the issue and recommendations, in compliance with the Council's decision making risk management guidance.

6.2 In the event of scheme cost variations, the programme will be re-prioritised to ensure that allocations are fully spent but not exceeded

7 EQUALITIES

7.1 An Equalities Impact assessment has not been completed because any potential adverse impacts on vulnerable groups will be identified and mitigated at the project implementation stage.

8 RATIONALE

8.1 The programme of works has been developed to achieve the objectives set out in the Joint Local Transport Plan

9 OTHER OPTIONS CONSIDERED

9.1 Projects are prioritised in accordance with the objectives of the Joint Local Transport Plan.

10 CONSULTATION

- 10.1 Cabinet members; Other B&NES Services; Section 151 Finance Officer; Chief Executive; Monitoring Officer
- 10.2 Consultation with Ward Councillors and Parish Councils will be carried out by email.

11 ISSUES TO CONSIDER IN REACHING THE DECISION

11.1 Social Inclusion; Customer Focus; Sustainability

12 ADVICE SOUGHT

12.1 The Council's Monitoring Officer (Divisional Director – Legal and Democratic Services) and Section 151 Officer (Divisional Director - Finance) have had the opportunity to input to this report and have cleared it for publication.

Contact person	Adrian_Clarke@BathNES.gov.uk Tel 01225 395223	
Sponsoring Cabinet Member	Councillor Roger Symonds	
Background papers	http://travelplus.org.uk/our-vision/joint-local-transport-plan-3/read- the-final-jltp3-strategy	
	<u>http://travelplus.org.uk/media/250770/prog%20report%201112%20</u> final%20050912%20v3.doc	
Please contact the report author if you need to access this report in an		

Please contact the report author if you need to access this report in an alternative format

This page is intentionally left blank

TRANSPORT IMPROVEMENTS	2013/14
FUNDING:	
	4005
LTP - Integrated Transport Block Grant	1225
S106 Contributions and other contributions	39
TRANSPORT IMPROVEMENTS FUNDING	1264
Less: LTP Contributions to Schemes Subject to	
Separate Approval	
Greater Bristol Metro Project	124
Separately Approved / Reported Projects	124
TOTAL FUNDING AVAILABLE	1140
Local Safety Schemes	
Anti-Skid Surfacing	10
AIP Scheme Investigation & design only	10
Tunley Fork Local Safety Scheme (Land Negotiations)	5
Local Safety Schemes Sub-total	25
Public Transport	
Keynsham Station Ramp	50
Public Transport Total	50
Managing Congestion	
Two Headed Man Junction Improvements (Design only)	10
A368/A39 Marksbury Juncton Improvements (Design only)	10
A367/Odd Down P&R Access Improvements (Design)	10
Parking schemes	45
Managing Congestion Sub-total	75
Safer Routes to Schools	
A39 Wells Rd Footway Provision Ph 1 High Littleton	50
Sydney Gardens/A36 ped phase at signals	35
Bristol Road, Chew Stoke footway provision	124
Freshford table/footway/bus stop	50
St Mary's Writhlington table	20
Safe Routes to School	35
Safer Routes to School Sub Total	314
Pedestrian Schemes	~ -
Aids to Mobility eg Dropped Kerbs & ped improvements	35
Public Rights of Way	85
Batheastern Footway	40
Paragon at Hay Hill - ped facility	15
Dunkerton Island/Footway	40
Highfield Chew Magna Footway	<u>20</u> 15
Argyle St/Grove St Ped Xing Facility New Inn - Farmborough Village Footway Provision (Ph1)	28
Tunley - Overdale Footway (Land negotiations)	205
Pedestrian Schemes Sub-total	283
Page 163	

TRANSPORT IMPROVEMENTS	2013/14
Troffia Managament Sahamaa	
Traffic Management Schemes	05
Minor Improvements	95
Toll Bridge Road Batheaston mini R/A (Design only)	5
Ralph Allen Drive/Mini R/A & Zebra	50
A367 Foseway speed reducing works Westfield (VAS)	5
Claverton Down Rd/Norwood Ave R/A & slip road entry	66
Redhill/Meadgate Camerton mini R/A (Design only)	5
Monksdale Rd mini R/a and VAS (Design only)	5
Traffic Management Schemes Sub-total	231
Miscellaneous	
JLTP Development (Inc £20k contribution to WoE)	50
JLTP Monitoring & Equipment	30
WoE Contribution	20
Decluttering Programme	14
Capital Programme Management	48
Miscellaneous Sub-total	162
TRANSPORT IMPROVEMENTS TOTAL	1140
Funding Surplus (Deficit)	0

Bath & North East Somerset Council			
MEETING:	Council		
MEETING DATE:	19 th February 2013		
TITLE:	Treasury Management Strategy Statement and Annual Investment Strategy 2013/14		
WARD:	All		
AN OPEN PUBLIC ITEM			
List of attachments to this report: Appendix 1 - Treasury Management Strategy 2013/14 Appendix 2 - Annual Investment Strategy 2013/14 Appendix 3 - Authorised Lending List			

1 THE ISSUE

- 1.1 In February 2012, the Council adopted the revised CIPFA Treasury Management in Public services Code of Practice 2011 Edition, which requires the Council to approve a Treasury Management Strategy before the start of each financial year and for this to be scrutinised by an individual / group of individuals or committee.
- 1.2 In addition, the Department for Communities and Local Government (CLG) issued revised guidance on local authority investments in March 2010 that requires the Council to approve an investment strategy before the start of each financial year.
- 1.3 This report fulfils the Council's legal obligation under the *Local Government Act* 2003 to have regard to both the CIPFA Code and the CLG Guidance.
- 1.4 This report is tabled to be scrutinised by the Corporate Audit Committee at the 5th February 2013 meeting, following which any recommendations will be reported back verbally as an update to this report.

2 RECOMMENDATION

The Council agrees that:

- 2.1 the actions proposed within the Treasury Management Strategy Statement (Appendix 1) is approved.
- 2.2 the Investment Strategy as detailed in Appendix 2 is approved.
- 2.3 the changes to the authorised lending lists detailed in Appendix 2 and highlighted in Appendix 3 is approved.

3 FINANCIAL IMPLICATIONS

3.1 Included in the report and appendices.

4 CORPORATE PRIORITIES

4.1 This report is of a corporate and technical nature and therefore does not directly contribute to individual Corporate Priorities.

5 THE REPORT

Background

- 5.1 The Local Government Act 2003 requires the Council to 'have regard to' the Prudential Code and to set Prudential Indicators for the next three years to ensure that the Council's capital investment plans are affordable, prudent and sustainable.
- 5.2 The Act therefore requires the Council to set out its treasury strategy for borrowing and to prepare an Annual Investment Strategy; this sets out the Council's policies for managing its investments and for giving priority to the security and liquidity of those investments.
- 5.3 The suggested strategy for 2013/14 in respect of the following aspects of the treasury management function is based on the Treasury Officers' views on interest rates, supplemented with leading market forecasts provided by the Council's treasury advisor.

The strategy covers:

•	Treasury limits in force which will limit the treasury risk and activities of the Council;
•	Treasury Management Indicators;
•	The current treasury position;
•	The borrowing requirement;
•	Prospects for interest rates;
•	The borrowing strategy;
•	The investment strategy.

- 5.4 It is a statutory requirement under Section 33 of the Local Government Finance Act 1992, for the Council to produce a balanced budget. In particular, Section 32 requires a local authority to calculate its budget requirement for each financial year to include the revenue costs that flow from capital financing decisions. This, therefore, means that increases in capital expenditure must be limited to a level whereby increases in charges to revenue from: -
 - 1. increases in interest charges caused by increased borrowing to finance additional capital expenditure, and
 - 2. any increases in running costs from new capital projects , and
 - 3. increases in the Minimum Revenue Provision for capital expenditure

are limited to a level which is affordable within the projected income of the Council for the foreseeable future

5.5 The revised CIPFA Treasury Management in Public services Code of Practice 2011 Edition, adopted by Council in February 2012, requires the Treasury Management Strategy and policies to be scrutinised by an individual / group of individuals or committee. This report is tabled to be scrutinised by the Corporate Audit Committee at the 5th February 2013 meeting, following which any recommendations will be reported back verbally as an update to this report.

2013/14 Treasury Management & Annual Investment Strategy

- 5.6 The Strategy Statement for 2012/13 set Prudential Indicators for 2012/13 2014/15, which included a total borrowing requirement at the end of 2012/13 of £163 million. At the end of December 2012, external borrowing was at £120 million, with no further borrowing planned in the 2012/13 financial year.
- 5.7 The proposed Treasury Management Strategy is attached as Appendix 1 and includes the Treasury Management Indicators required by the Treasury Management Code. Although the indicators provide for a maximum level of total borrowing, this should by no means be taken as a recommended level of borrowing as each year affordability needs to be taken into account together with other changes in circumstances, for example revenue pressures, levels and timing of capital receipts, changes to capital projects spend profiles, and levels of internal cash balances.
- 5.8 The budget report, which is also on the agenda, includes appropriate provision for the revenue costs of the capital programme in accordance with this treasury Management Strategy.
- 5.9 Appendix 1 also details the Council's current portfolio position as at 31st December 2012, which shows after the netting off of the £89.9 million investments, the Council's net debt position was £30.1 million.
- 5.10 The Annual Investment Strategy is attached at Appendix 2. This sets 'outer limits' for treasury management operations. While the strategy uses credit ratings in a "mechanistic" way to rule out counterparties, in operating within the policy Officers complement this with the use of other financial information when making investment decisions, for example Credit Default Swap (CDS) Prices, Individual Ratings, financial press. This has been the case in recent years, which protected the Council against losses of investment in Icelandic banks.
- 5.11 The Counterparty listing in Appendix 3 includes credit ratings from three agencies, as well as a sovereign rating for each country. Counterparties who now meet the minimum criteria as recommended in Appendix 2 as at 31st December 2012 are included in the listing in Appendix 3.
- 5.12 Interest rate forecasts from the Council's Treasury advisors are included in Appendix 1.

6 **RISK MANAGEMENT**

- 6.1 The report author and Lead Cabinet member have fully reviewed the risk assessment related to the issue and recommendations, in compliance with the Council's decision making risk management guidance.
- 6.2 The Council's lending & borrowing list has been regularly reviewed over the past year and credit ratings are monitored throughout the year. All lending/borrowing transactions are within approved limits and with approved institutions. Investment & Borrowing advice is provided by our Treasury Management Advisers Arlingclose.

7 EQUALITIES

7.1 This report provides information about the Council's Treasury Management Strategy and therefore no specific equalities impact assessment was carried out.

8 RATIONALE

8.1 This report is a statutory requirement.

9 OTHER OPTIONS CONSIDERED

9.1 None.

10 CONSULTATION

10.1 Consultation has been carried out electronically with the Cabinet Member for Community Resources, Section 151 Finance Officer, Chief Executive and Monitoring Officer.

11 ISSUES TO CONSIDER IN REACHING THE DECISION

11.1 This report deals with issues of a corporate nature.

12 ADVICE SOUGHT

12.1 The Council's Monitoring Officer (Divisional Director – Legal & Democratic Services) and Section 151 Officer (Divisional Director – Finance) will have had the opportunity to input to this report and have cleared it for publication.

Tim Richens - 01225 477468 ; Jamie Whittard - 01225 477213 <u>Tim_Richens@bathnes.gov.uk</u> <u>Jamie_Whittard@bathnes.gov.uk</u>

Background papers	None			
Please contact the r format	Please contact the report author if you need to access this report in an alternative format			

This page is intentionally left blank

APPENDIX 1

TREASURY MANAGEMENT STRATEGY – 2013/2014

Introduction

In February 2012 the Council adopted the Chartered Institute of Public Finance and Accountancy's Treasury Management in the Public Services: Code of Practice Fully Revised 2011 Edition (the CIPFA Code) which requires the Council to approve a treasury management strategy before the start of each financial year.

In addition, the Department for Communities and Local Government (CLG) issued revised guidance on local authority investments in March 2010 that requires the Council to approve an investment strategy before the start of each financial year.

This report fulfils the Council's legal obligation under the Local Government Act 2003 to have regard to both the CIPFA Code and the CLG Guidance.

Treasury Borrowing Limits for 2013/14 to 2015/16

It is a statutory duty under s.3 of the Local Government Act 2003, and supporting regulations, for the Council to determine and keep under review how much it can afford to borrow. This amount is termed the 'Affordable Borrowing Limit'.

The Council must have regard to the Prudential Code when setting the Affordable Borrowing Limit. The Code requires an authority to ensure that its total capital investment remains within sustainable limits and, in particular, that the impact upon its future council tax levels is 'acceptable'.

The Affordable Borrowing Limit must include all planned capital investment to be financed by external borrowing and any other forms of liability, such as credit arrangements. The Affordable Borrowing Limit is to be set on a rolling basis for the forthcoming year and two successive financial years.

Treasury Management Indicators for 2013/14 – 2015/16

The Council measures and manages its exposures to treasury management risks using the following indicators. The council is asked to approve the following indicators:.

Security: average credit rating

The Council has adopted a voluntary measure of its exposure to credit risk by monitoring the weighted average credit rating of its investment portfolio.

	2013/14
Minimum Portfolio average credit rating	A

Interest rate exposures

This indicator is set to control the Council's exposure to interest rate risk. The upper limits on fixed and variable rate interest rate exposures, expressed as an amount of net principal borrowed will be:

	2013/14	2014/15	2015/16
Upper limit on fixed interest rate	£201m	£196m	£193m
exposures			
Upper limit on variable interest rate	£60m	£60m	£60m
exposures			

Fixed rate investments and borrowings are those where the rate of interest is fixed for the whole financial year. Instruments that mature during the financial year are classed as variable rate.

Maturity structure of borrowing

This indicator is set to control the Council's exposure to refinancing risk. The upper and lower limits on the maturity structure of fixed rate borrowing will be:

	Upper	Lower
Under 12 months	50%	0%
12 months and within 24 months	50%	0%
24 months and within five years	75%	0%
Five years and within 10 years	100%	0%
10 years and above	100%	0%

Time periods start on the first day of each financial year. The maturity date of borrowing is the earliest date on which the lender can demand repayment.

Principal sums invested for periods longer than 364 days

The purpose of this indicator is to control the Council's exposure to the risk of incurring losses by seeking early repayment of its investments. The limits on the proportion of total principal sum invested to final maturities beyond the period end will be:

-	
£50m	£50m
	£30III

Borrowing limits

The Authorised limits for external debt include current commitments and proposals in the budget report for capital expenditure, plus additional headroom over and above the operational limit for unusual cash movements.

The Operational boundary for external debt is based on the same estimates as the authorised limit but without the additional headroom for unusual cash movements. This level also factors in the proposed approach to use internal cash-flow and future capital receipts as the preferred financing method for the capital programme.

	2013/14	2014/15	2015/16
Operational boundary – borrowing	£167m	£156m	£159m
Operational boundary – other long-term			
liabilities	<u>£2m</u>	<u>£2m</u>	<u>£2m</u>
Operational boundary – TOTAL	£169m	£158m	£161m
Authorised limit – borrowing	£201m	£196m	£193m
Authorised limit – other long-term			
liabilities	<u>£2m</u>	<u>£2m</u>	<u>£2m</u>
Authorised limit – TOTAL	£203m	£198m	£195m

Current Portfolio Position

The Council's treasury portfolio position at 31st December 2012 comprised:

	Principal	Ave. rate
	£m	%
Total Fixed rate funding	PWLB 100	4.45
Variable rate funding	Market 20	4.50*
Other long term liabilities	Nil	N/A
TOTAL DEBT	120	4.46
TOTAL INVESTMENTS**	89.9	0.67
NET DEBT	30.1	

* The market loans are 'lenders options' or LOBO's. These are fixed at a relatively low rate of interest for an initial period but then revert to a higher rate of 4.5%. When the initial period is over the loans are then classed as variable, as the lender has the option to change the interest rate at 6 monthly intervals, however at this point the borrower has the option to repay the loan without penalty.

** Total Investments includes Schools balances where schools have not opted for an external bank account and cash balances related to B&NES PCT Pooled budgets and West of England Growth Points funding.

Prospects for Interest Rates

The Council has appointed Arlingclose as its treasury advisor and part of their service is to assist the Council to formulate a view on interest rates. The following section gives their commentary on the economic context and views on the prospects for future interest rates.

Economic Context

Despite some stronger economic growth data towards the end of 2012, consumers are yet to loosen their purse strings and businesses are still reticent to make long-term investment decisions. The momentum in GDP growth developed in the Olympics-affected third quarter is therefore unlikely to be sustained while uncertainty over the economic outlook persists. Consumer Price Inflation has picked up from the low of 2.2% in September to 2.7% in December and it is expected to be affected by volatility in energy and commodity prices throughout 2013.

The Bank of England's Monetary Policy Committee is monitoring current economic conditions after voting not to extend quantitative easing in November. Policymakers appear to be hoping the Funding for Lending Scheme (FLS), which started in August, is more effective at easing restricted credit conditions. Although HSBC has opted out of the scheme, most of the UK's biggest lenders have now signed up. There has been some indication in recent data that the FLS is beginning to boost lending to the household sector, but business lending remains relatively subdued. Further asset purchases remain a distinct possibility, although above target inflation may constrain the MPC in the near future. Based on the last Inflation Report, Bank of England policymakers believe there is a good chance that the CPI rate will remain above target throughout 2013.

The US Federal Reserve has responded to the slowdown in growth and employment with large scale asset purchases of \$85bn a month until the outlook for the labour market improves substantially. The US public finance 'fiscal cliff' remains a serious risk despite the last minute deal reached before the deadline at the end of December. The political turmoil is likely to return in February when the talks on increasing the debt ceiling will create a stage for further political brinkmanship, no doubt prompting further volatility in financial markets.

The Eurozone is making slow headway, with the European Stability Mechanism now operational, announcements on the Outright Monetary Transactions programme well received, and some progress being made towards banking union. These have placated markets and curtailed some of the immediate risks to the stability of the monetary union. A sustainable solution to the Eurozone crisis is some way off though, as fiscal integration and mutualisation of Eurozone sovereign debt liabilities remain politically unpalatable.

Interest rate forecasts

The Council's treasury management adviser, Arlingclose, believes that it could be 2016 before official UK interest rates rise. The US Federal Reserve has signalled it will keep interest rates "at exceptionally low levels" until at least 2015. The UK's safe haven status and the minimal prospect of short-term rate rises should maintain gilt yields near their current lows.

Arlingclose and Markets Interest Rate Forecasts

Aningciose central interest rate forecast – December 2012					
	Bank Rate	3 month	12 month	20-year	
		LIBID	LIBID	gilt yield*	
Q1 2013	0.50	0.40	0.90	2.80	
Q2 2013	0.50	0.40	0.90	2.80	
Q3 2013	0.50	0.40	0.95	2.80	
Q4 2013	0.50	0.45	0.95	2.80	
H1 2014	0.50	0.50	1.00	2.90	
H2 2014	0.50	0.50	1.00	2.90	
H1 2015	0.50	0.55	1.10	3.00	
H2 2015	0.50	0.60	1.10	3.00	

Arlingclose	aantual	intoroct	wata	faracat	December	2012
Ariinaciose	central	meresi	rale	iorecasi –	December	ZUIZ

* The Council can currently borrow from the PWLB at 0.80% above gilt yields

	Average annual Bank Rate %			
	2013	2014	2015	2016
Highest	0.60	1.60	2.80	3.60
Average	0.50	0.65	1.30	1.80
Lowest	0.25	0.25	0.50	0.50

The Council has budgeted for interest rates to remain constant at 0.75% for 2013/14 & beyond.

Borrowing Strategy

The Council currently holds £120 million of long-term loans, and we will continue to monitor appropriate opportunities for borrowing in line with the overall Capital Financing Requirement.

The Council's capital financing requirement (CFR, or underlying need to borrow) as at 31st March 2013 is expected to be £161 million, and is forecast to rise to £201 million by March 2014 as capital expenditure is incurred.

	£m
Not borrowed in previous	41
years	
Forecast increase in CFR	40
Loans maturing in 2014/15	0
TOTAL	81

The maximum expected long-term borrowing requirement for 2013/14 is:

However, based on current expectations for interest rates, it is likely that the Council will not undertake any new borrowing, reducing the size of the Council's investment balance instead. The capital financing budget for borrowing in 2013/14 assumes no new borrowing is taken during the year.

In addition, the Council may borrow for short periods of time (normally up to two weeks) to cover unexpected cash flow shortages.

Sources of borrowing

The approved sources of long-term and short-term borrowing will be:

- Public Works Loan Board
- any institution approved for investments that meets the investment criteria (this includes other local authorities)
- any other bank or building society approved by the Financial Services Authority
- Public or Private Bond Placement
- Special purpose companies created to enable joint Local Authority bond issues.

Debt instruments

Loans will be arranged by one of the following debt instruments:

- fixed term loans at fixed or variable rates of interest.
- lender's option borrower's option (LOBO) loans.
- bonds

As an alternative to borrowing loans, the Council may also finance capital expenditure and incur long-term liabilities by means of:

- leases
- Private Finance Initiative

Planned Borrowing strategy for 2013/14

With short-term interest rates currently much lower than long-term rates, it is likely to be more cost effective for the foreseeable future not to take on new borrowing and reduce the level of investments held instead.

The Public Works Loan Board allows authorities to repay loans before maturity and either pay a premium or receive a discount according to a set formula based on current interest rates. The Council will review opportunities to take advantage of this and replace these higher fixed rate loans with new loans at lower, fixed or variable interest rates, where this will lead to overall budget savings or reduce risk.

All rescheduling will be reported in the next available Treasury Management Monitoring report following its action with all rescheduling detailed in the annual review report.

Policy on use of Financial Derivatives

Local authorities have previously made use of financial derivatives embedded into loans and investments both to reduce interest rate risk (e.g. interest rate collars and forward deals) and to reduce costs or increase income at the expense of greater risk (e.g. LOBO loans). The Localism Bill 2011 includes a general power of competence that removes the uncertain legal position over local authorities' use of standalone financial derivatives (i.e. those that are not embedded into a loan or investment). The latest CIPFA Code requires authorities to clearly detail their policy on the use of derivatives in the annual strategy.

The Council will only use standalone financial derivatives (such as swaps, forwards, futures and options) where they can be clearly demonstrated to reduce the overall level of the financial risks that the Council is exposed to. Additional risks presented, such as credit exposure to derivative counterparties, will be taken into account when determining the overall level of risk. Embedded derivatives will not be subject to this policy, although the risks they present will be managed in line with the overall treasury risk management strategy.

Derivative counterparties

Financial derivative transactions may be arranged with any organisation that meets the approved investment criteria. The current value of any amount due from a derivative counterparty will count against the counterparty credit limit and the relevant foreign country limit.

This page is intentionally left blank

APPENDIX 2

ANNUAL INVESTMENT STRATEGY

Investment Policy

The Council will have regard to the CLG's Guidance on Local Authority Investments and CIPFA's Treasury Management in Public Services Code of Practice. Both the CIPFA Code and the CLG Guidance require the Council to invest its funds prudently, and to have regard to the **security** and **liquidity** of its investment before seeking the highest rate of return, or **yield**.

Investment instruments identified for use in the financial year are listed below under the 'Specified' and 'Non-Specified' Investments categories. Counterparty limits will be as set through the Council's Treasury Management Practices – Schedules.

The strategy of this policy is to set outer limits for treasury management operations. In times of exceptional market uncertainty, Council Officers will operate in a more restrictive manner than the policy allows, as has been the case during the last three years.

Avon Pension Fund Investments

The Council's Treasury Management team also manage the Avon Pension Fund's internally held cash on behalf of the Fund. New regulations required that this cash is accounted for separately and needs to be invested separately from the Council's cash, and the split has been managed this way since 1 April 2010. The Fund's investment managers are responsible for the investment of cash held within their portfolios and this policy does not relate to their cash investments.

The cash balance held internally is a working balance to cover pension payments at any point in time and as a result the working balance will be c. $\pounds 10$ million. This working balance represents around 0.5% of the overall assets of the Fund. These investments will operate within the framework of this Annual Investment Strategy, but the maximum counterparty limit and investment term with any counterparty were set by the Avon Pension Fund Committee at its meeting on 18^{th} December 2009. These limits are in addition to the Council's limits for counterparties as set out in Appendix 3.

West of England Revolving Investment Fund (RIF)

Bath and North East Somerset Council is the Accountable Body for the West of England Revolving Investment Fund, and acts as an agent holding Government grants until they are ready to be distributed to Local Authorities for infrastructure works over the next two years.

These funds are kept separate from those of the Council, and therefore do not form part of the Council's counterparty limit restrictions. The funds are

invested primarily to protect the capital, and in order to achieve this high level of capital security, investments are made solely with UK Central Government and UK Local Authorities.

Any interest earned on these investments is reinvested into the fund.

Specified Investments

Specified investments are those expected to offer relatively high security and liquidity, and can be entered into with the minimum of formalities. The CLG Guidance defines specified investments as those:

- denominated in pounds sterling,
- due to be repaid within 12 months of arrangement,
- not defined as capital expenditure by legislation, and
- invested with one of:
 - the UK Government,
 - o a UK local authority, parish council or community council, or
 - o a body or investment scheme of "high credit quality".

The Council defines the following as being of "high credit quality" for making specified investments, subject to the monetary and time limits shown.

	Maximum Monetary limit	Time limit (or notice)
Banks, building societies & other organisations holding long-term credit ratings no lower than A- or equivalent and a Fitch Support Rating (where given) no lower than 3.	£20m each (highest limit) ¹	12 months
The Council's current bank account (NatWest) if below the criteria above.	£10m	Next day
UK building societies not meeting the above criteria that have a minimum asset size of £4bn and a long-term rating of BBB or above.	£2m each	3 months
Money market funds ² and similar pooled vehicles holding the highest possible credit ratings (AAA)	£10m each	1 week
UK Central Government (Including Debt Management Agency Deposit Facility)	no limit	12 months
UK Local Authorities ³ (irrespective of ratings)	£10m each	12 months
Organisations and pooled funds which do not meet the above criteria, subject to an external credit assessment and a specific recommendation from the Council's treasury management adviser.	£5m each	12 months

¹ The matrix for limits on each rating is provided in Appendix 3. Banks within the same group ownership are treated as one bank for limit purposes. The countries from which banks the Council can invest are detailed in the paragraph "Foreign Countries" below

² as defined in the Local Authorities (Capital Finance and Accounting) Regulations 2003

³ as defined in the Local Government Act 2003

There is no intention to restrict investments to banks and building society deposits, and investments may be made with any public or private sector organisation that meets the credit rating criteria above. This reflects a lower likelihood that central government will support failing banks following the Independent Commission on Banking report, as well as the removal of restrictions on local authority purchases of corporate bonds in April 2012.

The majority of the Council's investments will be made for relatively short periods and in highly credit rated investments, giving priority to security and liquidity ahead of yield. However, where the Council has identified a core cash balance that is not required for any current or planned cash outflow, these funds will be considered suitable for a wider range of investments, with a more moderate focus on security and liquidity and a greater focus on achieving a level of investment income that can support Council services. These may include long-term investments with registered providers of social housing, small businesses or corporate bond funds where an enhanced return is paid to cover the additional risks presented. Standard risk mitigation techniques, such as wide diversification and external credit assessments, will be employed, and no such investment will be made without a specific recommendation from the Council's treasury management adviser.

Current account bank

Following a competitive tender exercise held in 2007, the Council's current accounts are held with National Westminster Bank plc, (NatWest), which is close to the bottom of the above credit rating criteria. The Council will treat NatWest as "high credit quality" for the purpose of making investments that can be withdrawn on the next working day, subject to the bank maintaining a credit rating no lower than BBB-.

Building Societies

UK building societies without credit ratings will be considered to be of "high credit quality", but subject to a lower cash limit and shorter time limit than rated societies. The Council takes additional comfort from the building societies' regulatory framework and insolvency regime where, in the unlikely event of a building society liquidation, the Council's deposits would be paid out in preference to retail depositors.

The Government has announced plans to amend the building society insolvency regime alongside its plans for wide ranging banking reform, and investments in lower rated and unrated building societies will therefore be kept under continuous review.

However, no investments will be made with building societies that have an asset size of lower than £4 billion, or who hold a long-term credit rating lower than BBB or equivalent due to the increased likelihood of default implied by this rating.

Money market funds

Money market funds are pooled investment vehicles consisting of instruments similar to those used by the Council. They have the advantage of providing

wide diversification of investment risks, coupled with the services of a professional fund manager. Fees of between 0.10% and 0.20% per annum are deducted from the interest paid to the Council.

Funds that offer same-day liquidity and a constant net asset value will be used as an alternative to instant access call accounts, while funds whose value changes with market prices and/or have a notice period will be used for longer investment periods.

Collateralised investments

Where the Council makes an investment with an organisation that is secured on collateral in a third party (e.g. a reverse repo or a collateralised deposit), the time limit may be extended to match the limit given above for the third party. However, the investment will still count against the cash limit of the organisation receiving the funds.

Non-Specified Investments

Any investment not meeting the definition of a specified investment is classed as non-specified. The Council does not intend to make any investments which are:

- denominated in foreign currencies,
- nor any with low credit quality bodies,
- nor any that are defined as capital expenditure by legislation, such as company shares.

Non-specified investments will therefore be limited to long-term investments, i.e. those that are due to mature 12 months or longer from the date of arrangement. The maximum duration of the investment will depend upon its lowest published long-term credit rating and whether it is a UK counterparty:

Long-term	Time limit	Time limit
credit rating	(UK)	(Foreign)
AAA	10 years	5 years
AA+	10 years	3 years
AA	10 years	2 years
AA-	5 years	18 months
A+	5 years	N/A
A	3 years	N/A
A-	18 months	N/A

The time limit for long-term investments in UK Local Authorities & Local Government will be 30 years.

Long-term investments will be limited to 50% of a counterparty's limit where it meets the above credit rating criteria (except the UK Government). The combined value of short-term and long-term investments with any organisation will not exceed the limits for specified investments highlighted above.

The total limit on long-term investments, and the total limit on non-specified investments is £50m.

Information on the security of investments

Full regard will be given to available information on the credit quality of banks and building societies, including credit default swap prices, financial statements and rating agency reports. No investments will be made with an organisation if there are substantive doubts about its credit quality, even though it may meet the credit rating criteria set out above.

Risk Assessments & Credit Ratings

The Council uses long-term credit ratings from the three main rating agencies Fitch Ratings Ltd, Moody's Investors Service Inc and Standard & Poor's Financial Services LLC to assess the risk of investment default. The lowest available credit rating will be used to determine credit quality.

Long-term ratings are expressed on a scale from AAA (the highest quality) through to D (indicating default). Ratings of BBB- and above are described as investment grade, while ratings of BB+ and below are described as speculative grade. The Council's credit rating criteria are set to ensure that it is unlikely that the Council will hold speculative grade investments, despite the possibility of repeated downgrades.

Credit ratings are obtained and monitored by the Council's treasury advisers, who will notify changes in ratings as they occur. Where an entity has its credit rating downgraded so that it fails to meet the approved investment criteria then:

- no new investments will be made,
- any existing investments that can be recalled or sold at no cost will be, and
- full consideration will be given to the recall or sale of all other existing investments with the affected counterparty.

Where a credit rating agency announces that a rating is on review for possible downgrade (also known as "rating watch negative" or "credit watch negative") so that it is likely to fall below the above criteria, then no further investments will be made in that organisation until the outcome of the review is announced. This policy will not apply to negative outlooks.

If further counterparties are identified during the year that meet the minimum credit rating criteria and conform to the other criteria set out in the Treasury Management Practice Schedules, they can be added to the lending list following the agreement of the Section 151 Officer.

The Council's investments are normally senior unsecured liabilities of the borrower, and the credit rating of the investment is therefore normally identical to the credit rating of the counterparty. However, where a credit rating agency awards a different rating to a particular class of investment instruments, the Council will base its investment decisions on the instrument credit rating rather than the counterparty credit rating.

Investment instruments

Investments may be made using any of the following instruments:

- interest paying bank accounts
- fixed term loans & deposits
- call or notice deposits (where the Council can demand repayment)
- callable deposits (where the bank can make early repayment)
- collared deposits
- certificates of deposit
- treasury bills and gilts issued by the UK Government
- bonds issued by multilateral development banks
- corporate bonds
- Shares in a pooled fund meeting the definition of money market funds in The Local Authorities (Capital Financing & Accounting) (Amendment) (England) Regulations 2004 No. 534.

Investments may be made at either a fixed rate of interest, or at a variable rate linked to a market interest rate, such as LIBOR.

Foreign countries

Investments in foreign countries will be limited to those that hold a AAA or AA+ sovereign credit rating from all three major credit rating agencies, and to a maximum of £20m per country for those rated AAA and £15 million per country for those rated AA+. Banks that are domiciled in one country but are owned in another country will need to meet the rating criteria of and will count against the limit for both countries. There is no limit on investments in the UK, irrespective of the sovereign credit rating.

Overseas subsidiaries of foreign banking groups will normally be assessed according to the country of domicile of the parent organisation. However, Santander UK plc (a subsidiary of Spain's Banco Santander) and Clydesdale Bank plc (a subsidiary of National Australia Bank) will be classed as UK banks due to their substantial UK franchises and the arms-length nature of the parent-subsidiary relationships.

Sovereign credit rating criteria and foreign country limits will not apply to investments in multilateral development banks (e.g. the European Investment Bank and the World Bank) or other supranational organisations (e.g. the European Union).

Liquidity management

The Council regularly reviews and updates its cash flow forecasts to determine the maximum period for which funds may prudently be committed. Limits on long-term investments are set by reference to the Council's medium term financial plan, levels of reserves and cash flow forecast.

Planned investment strategy for 2013/14

Investments are made in three broad categories:

- Short-term cash required to meet known cash outflows in the next month, plus a contingency to cover unexpected cash flows over the same period.
- Medium-term cash required to manage the annual seasonal cash flow cycle, including amounts to cover forecast shortages, planned uses of reserves, and a longer-term contingency.
- Long-term cash not required to meet cash flows, and used primarily to generate investment income.

Short-term funds are required to meet cash flows occurring in the next month or so, and the preservation of capital and liquidity is therefore of paramount importance. Generating investment returns is of limited concern here, although it should not be ignored. Bank deposit accounts and Money Market Funds will be the main methods used to manage short-term cash.

Medium-term funds which may be required in the next one to twelve months will be managed concentrating on security, with less importance attached to liquidity but a slightly higher emphasis on yield. The majority of investments in this period will be in the form of fixed term deposits with banks and building societies. Preference will continue to be given to investments with UK banks with approved credit ratings.

Cash that is not required to meet any liquidity need can be invested for the longer term with a greater emphasis on achieving returns that will support spending on local authority services. Decisions on making longer term investments (i.e. over 1 year) will be considered during the year after taking account of the interest rate yield curve, levels of core cash and the amount of temporary internal borrowing related to funding of capital spend. A wider range of instruments, including structured deposits, certificates of deposit, gilts and corporate bonds may be used to diversify the portfolio. The use of external fund managers that have the skills and resources to manage the risks inherent in a portfolio of long-term investments may be considered.

With short-term interest rates currently much lower than long-term rates, due consideration will also be given to using surplus funds to make early repayments of long-term borrowing. In addition to potentially significant savings on the interest rate differential, this strategy can also help to manage the Council's exposure to credit risk and interest rate risk.

Review Reports

The revised CIPFA Code of Practice requires that both mid year and annual review reports on treasury activities are reported to Full Council.

Other Matters

The CLG Investment Guidance also requires the Council to note the following matters each year as part of the investment strategy:

Treasury management advisers

The Council's treasury management adviser Sterling Consultancy Services was acquired by Arlingclose Limited in October 2012. Arlingclose continues to provide advice and information on the Council's investment and borrowing activities, although responsibility for final decision making remains with the Council and its officers. The services received include:

- advice and guidance on relevant policies, strategies and reports,
- advice on investment decisions,
- notification of credit ratings and changes,
- other information on credit quality,
- advice on debt management decisions,
- accounting advice,
- reports on treasury performance,
- forecasts of interest rates, and
- training courses.

The quality of this service is monitored by officers on a regular basis, focusing on supply of relevant, accurate and timely information across the headings above.

Investment training

The needs of the Council's treasury management staff for training in investment management are assessed every year as part of the staff performance development review process, and additionally when the responsibilities of individual members of staff change.

Staff regularly attend training courses, seminars and conferences provided by Arlingclose and CIPFA. Relevant staff are also encouraged to study professional qualifications from CIPFA, the Association of Corporate Treasurers and other appropriate organisations.

Investment of money borrowed in advance of need

The Council may, from time to time, borrow in advance of spending need, where this is expected to provide the best long term value for money. Since amounts borrowed will be invested until spent, the Council is aware that it will be exposed to the risk of loss of the borrowed sums, and the risk that investment and borrowing interest rates may change in the intervening period. These risks will be managed as part of the Council's overall management of its treasury risks.

The total amount borrowed will not exceed the 2013/14 authorised borrowing limit of £201 million. The maximum periods between borrowing and expenditure is expected to be two years, although the Council does not link particular loans with particular items of expenditure.

This page is intentionally left blank

Proposed Counterparty List 2013/14

		CRITERIA							
		FITCH RATINGS			Moody's Ratings		S&P Ratings		
			S/Term	L/Term	Support	S/Term	L/Term	S/Term	L/Term
	Duration	Council Limit (£m)	F1	А	3	P-1	A2	A-1	А
UK Banks	Sovereign Ra	ting		AAA		,	Aaa		AAA
Barclays Bank	3 Years	15	F1	А	1	P-1	A2	A-1	A+
HSBC Bank plc	5 Years	20	F1+	AA-	1	P-1	Aa3	A-1+	AA-
Lloyds Banking Group Lloyds TSB Bank	3 Years	15	F1	А	1	P-1	A2	A-1	А
Bank of Scotland	3 Years	15	F1	А	1	P-1	A2	A-1	Α
RBS Group National Westminster Bank	18 Months	10	F1	А	1	P-2	A3	A-1	А
Royal Bank of Scotland	18 Months	10	F1	A	1	P-2	A3	A-1	A
Santander UK plc (domiciled in UK)	6 Months	5	F1	Α	1	P-1	A2	A-1	А
Standard Chartered Bank	5 Years	20	F1+	AA-	1	P-1	A1	A-1	AA-
UK Building Societies									
Nationwide	3 Years	15	F1	A+	1	P-1	A2	A-1	A+
Yorkshire	3 Months	2	F2	BBB+	5	P-2	Baa2	A-2	A-
Coventry Leeds	3 Months 3 Months	2 2	F1 F2	A A-	5 5	P-2 P-2	A3 A3	-	-
	5 10011115	2	12	~	5	1-2	A3	-	-
Foreign Banks									
Australia	Sovereign Ra	0		AAA			Aaa		AAA
Australia & New Zealand Banking Group Commonwealth Bank of Australia National Australia Bank	18 Months 18 Months	10 10	F1+ F1+	AA- AA-	1 1	P-1 P-1	Aa2 Aa2	A-1+ A-1+	AA- AA-
National Australia Bank	18 Months	10	F1+	AA-	1	P-1	Aa2	A-1+	AA-
Westpac Banking Corporation	18 Months	10	F1+	AA-	1	P-1	Aa2	A-1+	AA-
Austria	Sovereign Ra	ting		ΑΑΑ			Aaa		AA+
Raiffeisen Bank International AG	6 Months	5	F1	Α	1	P-1	A2	A-1	А
Canada	Sovereign Ra	ting		AAA			Aaa		AAA
Bank of Montreal	1 Year	10	F1+	AA-	1	P-1	Aa2	A-1	A+
Bank of Nova Scotia	1 Year	10	F1+	AA-	1	P-1	Aa1	A-1	A+
Canadian Imperial Bank of Commerce Royal Bank of Canada	1 Year 18 Months	10 10	F1+ F1+	AA- AA	1	P-1 P-1	Aa2 Aa3	A-1 A-1+	A+ AA-
Toronto-Dominion Bank	18 Months	10	F1+	AA-	1	P-1	Aaa	A-1+	AA-
France	Sovereign Ra	ting		AAA			Aa1		AA+
BNP Paribas Group BNP Paribas	6 Months	5	F1+	A+	1	P-1	A2	A-1	A+
Groupe Credit Mutuel									
Banque Federative du Credit Mutuel	1 Year	10	F1+	A+	1	P-1	Aa3	A-1	A+
Credit Industriel et Commercial Group BPCE	1 Year	10	F1+	A+	1	P-1	Aa3	A-1	A+
BPCE	6 Months	5	F1+	A+	1	P-1	A2	A-1	А
Group Credit Agricole									
Credit Agricole	6 Months	5	F1+	A+	1	P-1 P-1	A2	A-1	A
Credit Agricole Corp. & Investment Bank Societe Generale	6 Months 6 Months	5 5	F1+ F1+	A+ A+	1 1	P-1 P-1	A2 A2	A-1 A-1	A A
Germany	Sovereign Ra	tina		ΑΑΑ			Aaa		ΑΑΑ
Deutsche Bank	6 Months	5	F1+	A+	1	P-1	A2	A-1	A+
DZ Bank	1 Year	10	F1+	A+	1	P-1	A1	A-1+	AA-
FMS Wertmanagement	5 Years	10	F1+	AAA	1	P-1	Aaa	A-1+	AAA
KfW Bankengruppe Landesbank Berlin AG	5 Years 1 Year	10 10	F1+ F1+	AAA A+	1 1	P-1 P-1	Aaa A1	A-1+	AAA
Landesbank Hessen-Thuringen	6 Months	5	F1+	A+ A+	1	P-1	A2	A-1	A
Netherlands	Sovereign Ra	ting		AAA			Aaa		AAA
Bank Nederlandse Gemeenten	5 Years	10	F1+	AAA	1	P-1	Aaa	A-1+	AAA
ING Bank NV	6 Months	5	F1+	A+	1	P-1	A2	A-1	A+
Rabobank Group	18 Months	10	F1+	AA	1	P-1	Aa2	A-1+	AA-
Norway DNB NOR Bank	Sovereign Ra 1 Year	ting 10	F1	AAA A+	1	Р-1	Aaa A1	A-1	AAA A+
					,			∩ -1	
Singapore	Sovereign Ra	-	E4 -		4	Р-1	Aaa	A 4	
Development Bank of Singapore Oversea-Chinese Banking Corp	18 Months 18 Months	10 10	F1+ F1+	AA- AA-	1 1	P-1 P-1	Aa1 Aa1	A-1+ A-1+	AA- AA-
					•				

Proposed Counterparty List 2013/14

		CRITERIA							
			FITCH RATINGS		Moody's Ratings		S&P Ratings		
			S/Term	L/Term	Support	S/Term	L/Term	S/Term	L/Term
	Duration	Council Limit	F1	А	3	P-1	A2	A-1	А
Sweden	Sovereign R	(£m) ating	E1	AAA	3		Aaa		
Nordea Group	oovereightit	ating		~~~		,			~~~
Nordea Bank AB	18 Months	10	F1+	AA-	1	P-1	Aa3	A-1+	AA-
Nordea Bank Finland plc	18 Months	10	F1+	AA-	1	P-1	Aa3	A-1+	AA-
Skandinaviska Enskilda Banken (SEB)	1 Year	10	F1	A+	1	P-1	A1	A-1	A+
Svenska Handelsbanken	18 Months	10	F1+	AA-	1	P-1	Aa3	A-1+	AA-
Swedbank AB	6 Months	5	F1	A+	1	P-1	A2	A-1	A+
Switzerland	Sovereign R	ating		AAA			Aaa		ААА
Credit Suisse	6 Months	5	F1	Α	1	P-1	A1	A-1	A+
UBS AG	6 Months	5	F1	А	1	P-1	A2	A-1	Α
USA	Sovereign R	ating		AAA			Aaa		AA+
Bank of New York Mellon	18 Months	10	F1+	AA-	1	P-1	Aa1	A-1+	AA-
J P Morgan Chase Bank NA	1 Year	10	F1	A+	1	P-1	Aa3	A-1	A+
Wells Fargo Bank NA	18 Months	10	F1+	AA-	1	P-1	Aa3	A-1+	AA-
Supernational									
Council of Europe Development Bank	3 Years	10	F1+	AA+	-	P-1	Aaa	A-1+	AA+
European Bank for Reconstruction & Dev	10 Years	25	F1+	AAA	-	P-1	Aaa	A-1+	AAA
European Investment Bank	5 Years	10	F1+	AAA	-	P-1	Aaa	A-1+	AAA
Inter-American Development Bank	5 Years	10	F1+	AAA	-	P-1	Aaa	A-1+	AAA
IBRD (World Bank)	5 Years	10	F1+	AAA	-	P-1	Aaa	A-1+	AAA
Nordic Investment Bank	5 Years	10	-	-	-	P-1	Aaa	A-1+	AAA

Credit Rating Matrix - UK Banks, Building Societies & Other Organisations from 1st April 2013

	Maximum	Long			Long
Total Limit	Term	Term	Support	Long Term	Term
£2M*	3 Months*	BBB	5	Baa2	BBB
£10M	18 Months	A-	3	A3	A-
£15M	3 Years	А	3	A2	А
£20M	5 Years	A+	1	A1	A+
£20M	10 Years	AA	1	Aa2	AA
£25M	10 Years	AAA	1	Aaa	AAA

* This amount & tier is only applicable to UK Building Societies with an asset size of £4billion or

Credit Rating Matrix - Foreign Banks & Other Organisations from 1st April 2013

	Maximum	Long			Long
Total Limit	Term	Term	Support	Long Term	Term
£5M	6 Months	А	3	A2	А
£10M	1 Year	A+	3	A1	A+
£10M	18 Months	AA-	3	Aa3	AA-
£10M	2 Years	AA	2	Aa2	AA
£10M	3 Years	AA+	2	Aa1	AA+
£10M	5 Years	AAA	1	Aaa	AAA

Summary Guide to Credit Ratings

i.

Rating	Details
AAA	Highest credit quality – lowest expectation of default, which is unlikely to be adversely affected by foreseeable events.
АА	Very high credit quality - expectation of very low default risk, which is not likely to be significantly vulnerable to foreseeable events.
A	High credit quality - expectations of low default risk which may be more vulnerable to adverse business or economic conditions than is the case for higher ratings.
BBB	Good credit quality - expectations of default risk are currently low but adverse business or economic conditions are more likely to impair this capacity.
BB	Speculative - indicates an elevated vulnerability to default risk, particularly in the event of adverse changes in business or economic conditions over time.
В	Highly speculative - indicates that material default risk is present, but a limited margin of safety remains. Capacity for continued payment is vulnerable to deterioration in the business and economic environment.
ccc	Substantial credit risk - default is a real possibility.
сс	Very high levels of credit risk - default of some kind appears probable.
с	Exceptionally high levels of credit risk - default is imminent or inevitable.
RD	Restricted default - indicates an issuer that has experienced payment default on a bond, loan or other material financial obligation but which has not entered into bankruptcy filings, administration, receivership, liquidation or other formal winding-up procedure, and which has not otherwise ceased operating.
D	Default - indicate san issuer that has entered into bankruptcy filings, administration, receivership, liquidation or other formal winding-up procedure, or which has otherwise ceased business.

Bath & North East Somerset Council

MEETING: COUNCIL

MEETING 19th FEBRUARY 2013 DATE:

TITLE: ANNOUNCEMENT OF CHAIR(MAN) DESIGNATE FOR 2013/14

WARD: ALL

AN OPEN PUBLIC ITEM

List of attachments to this report: None

1 THE ISSUE

- 1.1 The purpose of this report is to invite the Council to name a Councillor as Chair(man) of the Council (Designate) for the next Council Year beginning in May 2013.
- 1.2 The Council, at its Annual Meeting on 16th May 2013, will still be required formally to elect a Councillor to be the Chair(man) of the Council for the forthcoming Council Year.

2 **RECOMMENDATION**

2.1 That the Council names a Councillor to be treated for forward planning purposes as Chair(man) of the Council (Designate) for the 2013/14 Council Year.

3 FINANCIAL IMPLICATIONS

3.1 There are no financial implications arising from this report.

4 CORPORATE PRIORITIES

4.1 Because of the pivotal role of the Chair(man) of the Council in the community, the nomination, at this stage, of the Councillor who will hold the office from May 2013 will contribute towards the achievement of the Council's vision of Bath and North East Somerset as a distinctive place with vibrant communities.

5 THE REPORT

- 5.1 The role of the Chair(man) of the Council has increased significantly in recent years. The number of civic engagements and requests for the presence of the Chair(man) at local events throughout the area continues to rise year on year.
- 5.2 In addition, the Chair(man) supports a number of awards and functions which require advance planning.
- 5.3 It is extremely helpful to community groups, and to the internal administration and scheduling of the Chair(man)'s functions, to have advanced confirmation of the Councillor who will hold the office from May 2013 and their preferred style of working and availability.
- 5.4 There is no requirement on the Council to name the intended Chair(man) in advance of the Council's Annual Meeting. The Council may prefer to leave the matter until then, when the election of Chair(man) for the year will by law be the first item of business on the agenda. The reasons for the recommended course of action are as set out in paragraphs 5.1 to 5.3 above.

6 RISK MANAGEMENT

6.1 No risk assessment related to this issue is required because the decision as to whether to make a nomination at this stage in the Council Year rests solely with the Council and does not affect the legal requirement for the Chair(man) to be elected at the Annual General Meeting in May.

7 EQUALITIES

7.1 The wide range of community groups who seek civic involvement in their activities through inviting the Chair(man) to their events will be helped in their diary planning by having a Councillor named as Chair(man) Designate at this stage. It will also assist the Councillor with his or her preparations for their year as Chair(man).

8 ISSUES TO CONSIDER IN REACHING THE DECISION

8.1 The Council's civic and corporate community interests.

9 ADVICE SOUGHT

9.1 The Chief Executive, the Monitoring Officer and the Section 151 Officer have had the opportunity to provide input to this report and have cleared it for publication.

Contact person	Jo Morrison, Democratic Services Manager, telephone 01225 394358
Background papers	None

Please contact the report author if you need to access this report in an alternative format

This page is intentionally left blank